A BIOGRAPHY OF NEWFOUNDLAND-BORN AUTHOR RON POLLETT (1900-55)

CENTRE FOR NEWFOUNDLAND STUDIES

TOTAL OF 10 PAGES ONLY MAY BE XEROXED

(Without Author's Permission)

AUDREY DEENA SCHULTZ





Permission has been granted to the National Library of Canada to microfilm this thesis and to lend or sell copies of the film.

The author (copyright owner) has reserved other publication rights, and neither the thesis nor extensive extracts from it may be printed or otherwise reproduced without his/her written permission.

L'autorisation a été accordée à la Bibliothèque nationale du Canada de microfilmer cette thèse et de prêter ou de vendre des exemplaires du film.

L'auteur (titulaire du droit d'auteur) se réserve les autres droits de publication; ni la thèse ni de longs extraits de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation écrite.

ISBN 0-315-31023-5

A BIOGRAPHY OF NEWFOUNDLAND-BORN AUTHOR RON POLLETT (1900-55)

BY

@Audrey Deena Schuitz, B.A. Hons.

A Thesis submitted to the School of Graduate Studies in partial fulfillment of the requirements for the degree of

Master of Arts

Department of English Language and Literature Memorial University of Newfoundland February 1985

ABSTRACT

This higgsphical study of Nowfoundhand-born author flow Pollett presents a detailed account of his life, and attempts in its final chapter to ansees his literary contribution, particularly as an <u>fining</u> author, to Newfoundhand literature. To date, no extensive study either of Pollett's life or of his works has been undertaken. This thesis attempts to redress this oversight in the area of Newfoundhand studies; and to provide, particularly through the use of information obtained from approximately 100 personal interviews and pieces of correspondence, previously undocumented and associated materials pratinging to Pollett's life and works.

Chapter One focuses on Pollett's early years, from 1909 to 1916, in New Harborr, Training Bay, a Newfoundland outport where he learned the traditional skills and values of a fishermanfamer. These skiteen years are presented as a time of preparation for the tasks and ideals which he was to pursue during his young adulthood.

Chapter Two describes Pollett's experiences from 1027 to 1924 in various professions—outport schooleacher, record-keeper, and linotype operator—as he migrated to a sumber of villages
and towas within Newfoundland, these emigrated first to Montreal, and finally to New York. Percoal and economic circumstances which prompted these changes of profession and residence are
exacted.

In Chapter Three Pollett's experiences in New York from 1925 to early 1945 are described.

This chapter provides a glimpse into the professional and demestic life of one of the thousands of
Newfoundlanders who settled in New York during the 1920s.

Chapter Four describe Pollett's life between 1945 and his death in 1955, and suggests the circumstances which led him, by 1946, to emback not a second career, as an author. Aspects of the writing he produced, based on his own and other Newfoundhanders' lives at home and abroad, are discussed in relation to his rapidly declining health and perenations finances.

Chapter Five focuses mainly upon Pollett's three major works: The Ocean at Mr. Door and
Other Newfoundland Outport Stories (1947): Peter the Grate: An Outport Character Study

(1952); and The <u>Ocean at My Door</u> (1956). This chapter addresses the following topics: the extent of Pollett's <u>ocuves</u>; the kind of writing he produced; his writing, seen in the larger literary context; and finally, his achievement as a writer.

PREFACE

This biographical study of Newfoundland-born author Ron Pollett (1900-55) was undertaken in order to provide a detailed and accurate portrait of his life and works. Pollett was born in 1900 at New Harbour, an outport fishing community in Trinity Bay, Newfoundland, and this place became the setting for much of the writing he began to publish by 1946. His writing is characterized by frequent use of sharp contrasts; at the time he began to write seriously, he resided in Brooklyn. New York, and city life as compared to traditional outport life became for him a rich source of literary inspiration. Pollett was one of the very few of the thousands of Newfoundlanders who emigrated during the 1920s to write about his experiences away from home. This, in part, is what distinguishes him as a local author. My own initial interest in Pollett was sparked by this emphasis in his writing on the émigré experience. My preliminary search for printed materials pertaining to his life and work, however, revealed a dearth of resources. A broader scope for this thesis than simply a focus on his experiences as an émigré Newfoundlander was therefore demanded. The five chapters in this thesis trace his life and development during his youth, his seven years as a transient jobholder, his professional and domestic circumstances during his thirty years as a Newfoundlander in New York, during the last ten of of which he was a writer, and, finally, his achievements as an author. As such, this study is the first of its kind to provide an extensive survey of Ron Pollett's life and writings.

Research for this thesis has proceeded by making extensive use of the resources of oral history; by gleaning and verifying hisgraphical details in Pollett's own writings; and by collecting and examining often facefung printed ourses pertinent to Pollett's life and works. Of the attempt to create an illuminating biography, Samuel Johnson once wrote, "More knowledge may be gained of a man's real character, by a short conversation with one of his servants, than from a formal and studied narrative, begun with his pedigree, and ended with his funeral." In my attempt to realize this goal, approximately 100 interviews and pieces of correspondence have been undertaken with 80 informants, unastly during 1825-3. Informants were chosen on the basis of their knowledge of the subject of this study. Interviews were based on informal questionnaires

tailored to those periods and experiences in Ros Pollett's life familiar to each informant. These questionnaires were devised from biographical and cultural data gleased from Politett's own works, and from general historical sources available in print. In so case, however, was information obtained in an interview used in this thesis (beyond personal impressions of an informant-and these are stated as such) if not corroborated and sometimes enlarged upon by at least one printed document or by a number of other informants. Due caution was exercised in devising interview procedures and questions, for though oral history has been a most necessary aid to research on this topic, it is not a technique without its shortcomings for scholarly research and documentation. A perspective on each period of Pollett's life was attempted only after all the data were collected and wrifted. This method, not common in literary study, has been unexpectedly fruitful in gaining an understanding of Pollett's life and times.

Gaps in this study can be attributed to three factors in particular. The first is failure of memory on the part of informants, the majority of whom are aged 70 or more, and who have been asked to describe events dating from the early twentieth century. The second is the scarcity of Pollett's personal papers, such as letters, journals, or drafts of sketches and stories, most of which were destroyed at the time of his death. The third is the dearth of formal studies describing in any extensive and scholarly manner the history of common life in Newfoundland during Ron Pollett's lifetime. Aspects of the rich, traditional oral forms of literature known to outport populations have indeed been written about by folklorists, but the influence of this traditional culture upon formal, written literature has not been studied in any depth. This gap is an impediment to a full understanding of Pollett as a Newfoundland author. Despite these shortcomings, it is hoped that this thesis, although each of its five chapters may be seen as preliminary and sketchy in parts, will provide the basis for a renewed interest in Pollett and his works. In the thirty years since his death his sketches have diminished in popularity, but many of his concerns, such as the quality of traditional outport life and the nature of émigré experience, remain timely, It is therefore hoped that this thesis will provide the impetus for further study of aspects of Pollett's oeuvre.

Two people—Dean Frederick A. Aldrich and Dr. Patrick O'Flaberty— are acknowledged most gratefully for making this work possible. I wish to thank Dean Aldrich for his frequent, sympathetic responses toward my study, which involved, first of all, the implementation of a number of administrative changes on my behalf. He is responsible for an extension of my demorial University of Newfoundland Graduate Fellowship (1981-84); and I wish to thank him as well for his very generous financial support given to me during the research phase of my study. Without his belo. I would not have been able to beein, pursue, or conclude my studies at this university.

Equally sincere are my thanks to my supervisor, Dr. Patrick O'Flaherty, whose own research into Newfoundland literature and the life of Ron Follett prompted this study. Most appreciated has been his invaluable, though offer-tried, patience shows toward the slow progress of this study and toward my acquiring a suitable writing style. But his prompt, helpful responses to all of my needs, such as approving alterations in my academic curriculum, and arranging for certain filanacid exceeditures and for work-tonce, also are very much asperciated.

All of my informants were most helpful to my research project: they and many others kindly extended their hospitality to me after realizing difficulties involved in my travelling to and from interviews. In particular I would like to thank Mrs. Caroline Pike, Ron Pollett's wife, and their son, Ronald J. Pollett, who could not have helped more with my project or have helped me more when I travelled to Brookkyn. New York.

I wish to extend thanks to the staff of the university's Folklore and Language Archive (MUNFLA) for providing me with tape recording equipment and casette tapes during all of my fieldtrips. I would like to thank in particular Assistant Archivist Philip Hiscock for his help in enabling my research materials to be housed, preserved, and most importantly, made accessible for use at MUNFLA.

Finally, I would like to thank my parents, Ileans and William Schultz, for their unceasing encouragement of my studies and for their concerns for me during this project. My husband Douglas Kinsella shared equally in their concerns, and I thank him for his helping me on each step of the way to bring this project to its completion.

TABLE OF CONTENTS

ABSTRACT	ii
PREFACE	iv
CHAPTER ONE: NEW HARBOUR BOYHOOD, 1900-16	1
CHAPTER TWO: LEARNING HIS TRADES: THE TRANSIENT YEARS, 1917-24	24
CHAPTER THREE: 'NEW YORK-NEWFOUNDLANDER,' 1925-45	43
CHAPTER FOUR: LOOKING HOMEWARD: THE WRITING OF THE OCEAN AT MY DOOR, 1945-55	63
CHAPTER FIVE: POLLETT'S WORKS: RECAPTURING AN OUTPORT HERITAGE	91
ABBREVIATIONS	121
NOTES	122
BIBLIOGRAPHY	159

NEW HARBOUR BOYHOOD, 1900 - 16

Ros Pollett's boyhood was an initiation into the demands and occasional pleasures of Newfoundiand outport life at the turn of the twentieth century. He was horn on June 25, 1900, at New Harbour, a fishing community located on the south side of Trinity Bay on the Avalon Penisuals of Newfoundland, where for generations his family had been resident fishermen-farmer. Activities during his early years were largely confined to hearning the traditional skills of a fisherman-farmer, a role for which his own extended family provided a model. As a young boy, Pollett worked hard in the vegetable gardens, woods, and particularly in the fishery, sharing the normal familial labours required to keep the outport system of life going. There was also time for play and companionship, when Pollett enjoyed the spirit of competition which diversions of outdoor games and storytelling afforded. Between 1908 and 1914, he was educated at the village's Church of England school where his scholarly achievements and interests led him to search for other opportunities than those available to him at New Harbour.

1

A brief sketch of the history of New Harbour prior to 1900 may serve to introduce the village of Pollett's day. The name of the community likely denoted simply a new habitation for
settlers, some distance from the long established centers of Trinity and Old Perlican in Trinity
Bay. By 1901, this new harbour had attracted a number of families such as the Georges, Hillies
and Thornes, who established themselves in the area of Cat Cove on New Harbour's southern
shore. Local waters provided good cod fishing grounds, and this was the primary inducement for
new settlers to come there, since cod was the main currency of entport trade. Local forests, rich
in timber resources of pine and sprace, provided ample building materials for losses, boots, and
shing premises, and the plentitude of timber was an inecutive for the Newbooks, a family of
master shipbuilders, to establish a firm in New Harbour by 1806. Local resources subsequently

attracted other families, among them the Higdons, Polletts, Williamses, and Woodmans, and the settlement soon extended along the northern shore of the harbour from Newbook's Point toward New Harbour Pond, where a river emptying into New Harbour Pond was an accessible source of fresh water. By 1945 the population numbered 209⁴, comprising several dozen large families in clusters of dwelling houses facing waterfront properties and the harbour, beyond which were gently rolling, thickly forested hills.

Flictuations in the cod fishery were of great concern to the settlers, for throughout the 1800s they depended on the sea more than on the land to obtain their livelihood. Failures in the cod fishery could cause widespread hardships, though some effort was made to grow vegetables and raise livestock, and the economy was also diversified through seasonal work at the spring seal fishery and occasional carpentry or manual labour for the local firm of master shipbuilders, the Newbooks. Prolonged failures occurred throughout the 1800s; and there were periodic failures in the cod fishery as well as in other parts of the rural economy, such as potato farming and the seal fishery, during each of the next several decades. Consequently, debts to local merchant firm eccumulated, so much so that merchants would refuse to risk further extensions of credit for winter provisions and fishing gear. Petitions for aid from the period indicate the frequent precarious fortunes of settlers—in one instance some families were noted to be "in a state of perfect destitution," while in another, the scarcity of food forced some families to eat their seed potatoes during winter months. Although there was a degree of interdependence among extended families and seighbours, only an ample excited at the cod fishery could provide reasonable comfort.

The education and pastimes of several generations of people raised in the relatively isolated settlement of New Harbour throughout the aineteenth century necessarily were mainly of a traditional nature. Parents and neighbours passed on knowledge of skills and duties demanded by outport life. Fathers taught sons how to fish and how to procure the wood needed for use at the fishing rooms and in the home; mothers taught daughters how to assist in fish processing and how to card, spin, and knit wool for clothing, as well as other necessary tasks of daily household management. Extensive and diligent familial ishour was the accepted norm of life; and in such rigorous daily circumstaness, the occasional holiday celebrations were marked by an almost total absence of labour. The Christmas senson was the single, prolonged festive time of the calendar year, and was observed according to the traditions the settlers, who for the most part were of English ancestry and the Protestants faith, brought with them. Christmas celebrations included "times," juaneying, and other holiday customs shared by all age groups and lasting for twelve days or more. At other times of year, Sunday house visits during which dinners were shared among families and neighbours, and out-of-doorn meetings among fishermes, provided more informal, regular social interaction among community members, all of which helped the residents to create a sufficiently hospitable community in which to maintain themselves and raise their children.

Schools were slow in developing. The first schooltescher, Charles Elford, was appointed to New Harbour in 1830, although construction of a schoolhouse was not completed until the late 1840s.* Facilities and equipment doubtless were meagre for his needs, and attendance by local children was sparse and sporadic. By 1800 seven children were in school, but sixty-nine others were not; and a similar bleak ratio was characteristic of other periods during the century. Exidence from the years 1830 to 1850 indicates that teachers', school inspectors', and occasional newspaper reporters' efforts to reverse this trend by reminding parents of New Harbour of their complicency toward formal education apparently were of little avail. W Constables appointed to New Harbour as early as the 1820s likely took no interest in the matter, since attendance at school was not deemed compulsory in Newfoundland until 1943. Demands of time and labour upon all family members continued to preclude regular attendance and payment of school fees by young students.

Resident clergymes of the Church of England were appointed to serve a developing parish in New Harbour after the mid-1800s, and the degree of literacy achieved by the people was attained mainly through the teaching of scriptural texts. Thribur C. Waghorne served at New Harbour from 1877 to 1893, and dering his years of service the parishoners helped to construct a new church, the Church of St. Augustine. Building began in 1879, and as edifice was raised to repines the rough structure at Cai Cove which had served as a church for the sporadic visits of internat clergymen since 1815. Apart from his formal duties, Waghorne shared with the community his varied avocations of studying nature, music, and literature. He sent samples of local flora to the British Museum; he and his wife taught piano music and singing; he lest books from his large library; and he wrote magazine articles about the folklife of New Harbour, two of which were published in the <u>Journal of American Folktore</u>. Information for his publications was obtained from local residents, but the full influence of Waghorne's character and interests upon the community at large cannot be documented. The position of clergyman, like that of merchant or constable, was such that frequent informal contact with ordinary fishing families may not have been regular; and the small group of officials and merchants that formed the village slite may have seemed precupied with tasks and concerns that were relatively incompatible with the essential labourn of daily outport life.

As a result of both groups' labours, nevertheless, by 1901 New Harbour was an established and productive community. If a 1901 eight merchants and traders did business there, the most promisent being Fred Woodman, a New Harbour resident. They served a thirving community whose population had more than doubled since 1845, now reaching a total of 551. Unlike smaller local communities, a schoolteacher and a clergyman now resided at New Harbour, and a school and a church were maintained for the community. Some developments elsewhere in Newfoundhad also became important to the people of New Harbour as the new century dawned. More opportunities for economic growth had become available on the recent completion of a transantional railway and on the formation of the Anglo-Newfoundland Development Company, which by 1905 was developing a paper mill industry at Grand Falls in central Newfoundland. Many New Harbour men went to work permanently in Grand Falls, and eventually took their families with them. Others obtained seasonal work at the company's lumbercamps and as well at the lumber operations set up by the Reid-Newfoundland railway company. These activities affected New Harbour in the first decades of the twentieth century, although its way of life, as it had been upon its inception, was still very much centered upon the fishery.

It was in these conditions of established tradition and gradual change that Ross Pollett grew to young manhood. Christened Rossald Pollett on July 55,1000, he was the firstborrs on and third child of James and Honorah (Reed) Pollett of New Harbour. The christening took place as the Cherch of St. Augustiae, for his family was Anglican. In 1947 Pollett recalled being sold of an incident which occurred during his baptismal service; his "waits topped the peals of the church organ," and his godfather said encouragingly to his father, "You've got a strong feller there, Jim. Nothing better than a boy who can holler. He'll be manning the for'rd our in no time." This was an appropriate remark, perhaps, about the life which the two men knew, one centered as it had been for generations upon the inshore, family-based cod fishery.

Pollett's father Jim was a fisherman-farmer who lived his entire life in New Harbour. Born in 1864, he was the youngest of seventeen children born to Sarah (Newhook) and her third husband John Pollett, who also was a fisherman-farmer. Jim's grandparents had been among New Harbour's earliest settlers: the Newhooks had settled there by 1806, and the Pollett family arrived soon after, although they had lived there seasonally for some twenty years previously, 16 The home in which Jim grew to manhood was one of the oldest in the village, and had provided a dwelling for the Pollett family for at least three generations. Ron Pollett would later imagine that in this home "the lean ghosts of many years of frugal living lay sandwiched among the layers of wallpaper....[and] buried in the stale odors [were] the tender words of family prayer [and] the iron curses of hard-shelled pioneers."17 Eventually Jim inherited this house near New Harbour Pond. together with his father's lands, which consisted of acreage surrounding the house and two garden plots some distance away; one on New Harbour Point across the harbour, the other near Hopeall, a village slightly north of New Harbour. Both of Jim's parents had been illiterate, and Jim himself had learned to read print but could not write. As a young man, however, he had learned the skills of a fisherman-farmer and had prospects of establishing a family of his own in the New Harbour community. Well remembered in New Harbour as having been a "real 'fishkiller' " and a "wonderful hard worker," he was equally well known for his wit and humour, attributes which perhaps made him an attractive suitor. 18

In 1889 Jim married Honorah Reed, aged eighteen, who in New Harbour church records is listed as a servant from Speed Eagle. Transity Bay. ²⁰ Her father, Jacob Reed, also was a fisherman-farmer. He or his ancestors probably had migrated from the populose Old Perlican region to Spread Eagle, which was a tiny fishing village south of New Harbour near Chapel Arm. No reads connected Spread Eagle and New Harbour, and whatever regular courting between Jim and Norsh took place prior to their marriage was made possible only by Jim's journeying by boat, or by foot for five or more miles through the woods, to see her. A tall, Jim, and dark-haired woman, Norsh was, according to those who remember her, a diligent and hard-working house-keeper. ²⁰ Her first home upon her marriage was in her husbands's ascentral bouse, where the couple resided with Sally Pollett, for as the youngest soa Jim had assumed the traditional responsibility of caring for his mother after his father's death. Norsh and Jim began to rear a family that grew to seven children by 1014. ²⁰ Their first two daughters, Lory, born June 11,1894, and Sara Jane, born July 17, 1997, were, like their firstborn son Ronald, delivered and raised in the Pollett ancestral home.

Some years prior to 1900 Jim Pollett kad legsa building a hone of his own next the Pollett family homestead. The building estatled felling and hashing logs from the nearby forests to a recently opened local sawmill, manually hewing other wooden parts for the structure, and setting money axief for those materials whick had to be perchased, seck as nails and roofing materials. Roo Pollett commented on this venture many years later: "Building the house was a spare time job; and with so many mouths to feed, my father had little time to spare." He added, "A typical NewYoundland fisherman-farmer, [my father] was not the kind to "static" moore buying anything he could make himself." The new house, completed in about 1904, was a two-storied gabled home with a veranda. Built facing the harbour, it was eventually surrounded by dogherry, apple, and fit trees, a vegetable garden, and a mendow. In it Pollett's four younger brothers and sistens were born: James Heary (on May 15, 1904); Clara (on July 4, 1907); Mary Florence (on May 14, 1910); and Jonathan Leonard (on August 17, 1914). Pollett's paternal grandmother Saily Pollett

also resided with the family in her son's new home.

Roa Polleti's family, like other families in New Harbour, was self-eupporting. Hard work and fregality enabled it to be well provided for, by the standards of that time and piace. His mother Norah was expected to help her bushand in curing fish and tending to the family's three gardens; she also had to perform all of her domestic tasks, which included making clothing for the ten members of the Pollett household. In later years, Pollett recalled her creation of "Josephized" jackets and other garments from the various materials available or affordable. In father Jim fished from his punt in the spring and summer, tended to his three gardens, and in the fall of the year settled his accounts with merchant Fred Woodman before going to work in the woods of central Newfoundland for the A.N.D. Company. The latter opportunity brought cash wages, whereas other modes of seasonal employment had facilitated only trade of goods between New Harbour fishermen and local merchants. Wages in the lumbercamps, however, were low, and it remained accessary for the entire Pollett family to contribute to the family's maintenance.²⁴

Jim Pollett was well-known for his strict and demanding upbringing of his children. According to the accounts of W. Bramwell Pollett, Bertha Pollett, Raiph Higdon, and other New Harbora residents, Jim's expectations from his family were equal to the demands which he made upon himself—he did not habitually drink or smoke, and though not a churchgoer was a disciplinarian who appears to have been steadfast in his beliefs in the virtues of good conduct and hard labour. Pollett's older sisters thus likely would have helped with the family work, such as tending to the family sheep and horse, and hauling water. Although Ros Pollett's own role in the family abours was during his sarliest years sonewhat peripheral and limited, in later years he was expected to help procure the family's wood supply, help with the garden work—planting vegetables, weeding, gathering, and harrowing— and assist with chores in the cod fishery, tasks which he learned from, and which were supervised by, his father.

The Pollett children's labours were determined according to their sex and age; and Pollett's younger brother Jim appears to have shared labour and also other activities with him. They shared in the tasks of filling the wood stoves with billets; and one pastime which Pollett later recalled was a game of his own devising called "Who Can Keep their Hands on the Hot Funnet the Longest," which he played with Jim in the bedroom they shared. He described this as a game "which was easy to learn but hard to play." ⁶⁸ Other competitive games he devised within the home, which, as he later stated, he sometimes used to tormeat his younger brother, merely provided more opportunities for manly sport and challenge. But he did not have a domineering disposition; indeed Pollett is best remembered by family and friends as a good-natured boy who "was always lots of fun." ²⁸ As long as he performed his own chores and did not interfere with his parents' labours, Pollett and his brother seem to have enjoyed a certain amount of freedom to entertain themselves as they pleased.

Despite the strictaess of his father, Poliett appears to have been satisfied with the life he knew, and in his carliest years he was relatively unaware of any sacrifices which his parents made for the family. Norah Poliett, a generous and intelligent woman according to those who knew her, apparently was heavily burdened by the many tasks demanded of her in her home, tasks which precluded her participation in social activities elsewhere. "There was always a big pot of something cooking on the store," Mrs. Jesse Fogwill, former neighbour and a frequent visitor, recalic: and she also remembers the rarity of Norah's visiting neighbours or leaving her hosse for other social reasons." Nor did she got outurn's. She welmod visitors to the Pollett home, however, and they often called in the evenings for card games and chats, or were invited as guests for Sunday afternoon dinners. Mrs. W.J. Lundrigan, for example, whose mother was Norah's sister, recalls as a child many very enjoyable weekend visits to the Pollett home, for the family was amiable company. She remembers a pleasant atmosphere in which all family members helped with their daily share of the work."

Oaly Sally Pollett, Ros Pollett's grandmosher, born in 1820 and blind by the early 1900s, was incapable of helping with the family labours. It was her pleasure, however, to rock young Pollett in a cradle kept in the kitchen and sing hymns to him; and in later years he listened to her tell "many wonderful yarns about ploneer days on the Avaion Peninsula." Her role in his upbringing Pollett appears to have thought significant from her be learned of his own family: and community's cultural beritage. He seems to have taken some pride in the labours and subsequent accompliahments of his New Harbour ancestors. It was this milies which he explored during his childhood years, years when his father taught him those skills which would equip him for a life in the fishers, and when he learned as well to enjoy the delithits of his outsets home.

111

Tagglt at an early age by his father at the family well to handle a bamboo trouting pole, Pollett later claimed that he, like other New Harbour boys, could wield one "almost from the time we could handle a teaspoon."

Man an animal of other ways also: learning from their fathers how to perform tasks which would contribute to the family's maintenance; exploring their communities; and participating in group pastimes mainly with friends their own age. Domestic tasks and responsibilities varied according to the esascanal nature in which they were performed, as did other activities. In retrospect, Pollett recalled his eagerness for the variety offered by the yearly cycle; not the slightest chance of monotony could be experienced, he thought, given the array of activities of each sharply chanting season.

**The father is a state of the property of the control of the start of the s

During his earliest years, Pollett's assigned tasks involved simply gathering wood from the wood pile in the yard and carrying it into the house, making wood shavings for the wood stoves during winter and summer, and from May to October performing tasks like picking berries to serve with lunch at the Pollett garden site on New Harbour Point.⁵² The simplest of tasks will were performed seriously: he took pains to please his father by producing neatly stacked piles of wood. He used his time at the garden site to explore the natural surroundings of the area, and as a result discovered an icy water spring which the family could use for refreshment, and gathered juniper herries which could be taken home and used for medical purposes. Though too young to help with the labour of digging potatoes and hastling hay, some of his time during the daily gaden tasks was speat fushioning trouting poles from tree branches. With these he undertook to

provide trout and tomcods for the family breakfast.

Taught and encouraged by his father to learn and improve skills in the woods, gardens, and fishing grounds, Pollett apparently was eager to assume a traditional role in his family and also among his peers. Fishing, in particular, occupied a good deal of his time, both as a chore and as a form of recreation. As children, Pollett later explained, he and his companions "had eyes only for the fish hooks and our minds were wholly on cod." He developed a preference for a certain type of worm to use as bait and laboured to obtain it. As he recalled, "Every spring I sifted tons of loam in our cabbage garden and upturned insumerable rocks seeking the special kind [of worm] I fancied. But the extra toil was well rewarded: the bait was so good I caught more and bigger trout than any other boys." His success is confirmed to some extent by Jesse Fogwill, who as a child was a daily visitor at the Pollett home: she recalls Pollett "always coming through the kitches door with a string of fish." Si

Pollett's predilection for fishing substantially determined much of his childhood activity.
From his window and the veranda of his home, he could see the fishing properties stretched from
one end of the harbour to the other. He observed the men as their work at the splitting tables
and on their cod flakes, and hours were spent venturing about this area. Pollett explored the
landwash, the fishing properties, and particularly the cuddy of his father's boat. In time Pollett
gradually widened his territory in search of local choice fresh water fishing grounds and salt water
ones near the shore. He explored nearby ponds and rivers in New Harbour Barrens, an area east
of his home which stretches eleven miles toward Conception Bay. His contemporary, W.
Bramwell Pollett, says that the Barrens area was a familiar spot to young New Harbour Doys;
they fished at Sutton and Rocky Ponds in warm weather, and in winter, when they went there to
gather wood for home use, they carried along a hook and piece of ealt meat to try their luck
fishing through the ice. Along with other boys his own ape, Pollett perfected his own fishing
whills on the Barrens; and at New Harbour Bridge, a 100 foot long wooden structure near Pollett's
home, the children fished for realtwater tors, etc., and vishnon peet.'

In general, Pollett's childhood activities in the home, at chores, and at play were associated

with male companionship and macculine values, such as mastery of fishery-based skills and competitive sport. In warmer weather, New Harbour Bridge became a central meeting place for Pollett and boys his own age such as Llewellyn Thorne, Ern Woodman, Selby Newbook, and about half a dozen others.⁵⁷ On and near the bridge they devised competitive feats that each had to master to prove his manliness. Walking across the anarow handrail, on either side of which was the harbour and New Harbour Pond, and swimming the half mile across the pond were two of these rices. Pollett's peers recall his attempts to swim further than any other boy, but strength, endurance, and precision were goals shared by all of the boys of his group. Skating on New Harbour Pond and rabbit-busting in New Harbour Barrens were activities which occupied them duing winter months, and they practiced skating swiftly and on thin ice, and competed with each other by trying to mare larger rabbits. Close observation and knowledge of the landscape were essential to these pastimes; mastery was attained by experience and practice. Parusing pastimes devised to heighten physical challenges, Pollett's competitive and ambitious nature was developed.

With other boys his own age, Pollett enjoyed a freedom to develop skills and qualities, both physical and imaginative. The company of older men also was part of Pollett's early experience, for adults often met on New Harbour Bridge and told ghost stories and tall tales to young boys and to each other. In the absence of radio, these oral forms of entertainment provided more localized and intimate diversions. Pollett later said he listened to local stories about haunted culverts, headless riders, and ghost ships as though they were gospel.²⁰ The ghost tales, in particular, made his hair stand on end. Nonethriess he was an eager participant when he and his peers often retold ghost tales which they had learned from older men, dramatizing each of them with "groans, whisten and cries coming over the water." These tales then became a source of creative amusement which marked and perhaps dissipated their fearfulness.

Despite the relief which may have been provided by such amusing retellings, these ghostly stories perhaps served to confine Pollett's territory to daylight adventures and to areas close to home, for beyond "in any direction there seemed no end to the spectral hurdles." to the years of his childhood, he never ventured further than New Harbour Barrens. An avid participant and listener, he appears to have been well satisfied with his pattimes and with the companionship of his peers. He learned an appreciation for the physical environs and people of New Harbour, and developed skills and qualities which would enable him to achieve a place among them. Influence upon him and his group from the world outside New Harbour were few. When Pollett described his "first amazing sight"—the transmational train that passed at some distance beyond the nearby hills are route to the capital city of St. John's—he suggested that its influence was limited. Only the novelty of the sound of its steam engines was attractive, and the boys imitated the chuffing sound while running races, rather than imitate the wheeting tound of a horse as they had always done."

One regular exception to the ordinary pastimes and fairly rigid good conduct of Pollett and his peers was displayed annually on November 5, Bonfire Night. a The significance of this ritual for Pollett and other participants was no longer the same as it had been for the English ancestors: to commemorate the hanging of a Catholic compirator, Guido Fawkes. Their concern only was to create a bigger and longer-lasting fire than in previous years. For weeks before this night, therefore, a communal undertaking of local boys was the gathering, sometimes pilfering, of empty wooden casks and other lumber to be set affame in a clearing near the bridge. As they matured, Pollett and his peers took greater risks to obtain materials for more elaborate annual fires. Two of Pollett's contemporaries, Jim Thorne and Charles Hillier, recall that on at least one occasion benhouses and pighouses, including the animals within them, were taken and thrown into the fire. ⁴³

Pollett emphatically stated that Bonfire Night was "the one night [all] young outporters lived for," He recalled that most of the population gathered annually near New Harbour Bridge to view the spectacular bonfire. The night's providing of entertainment for the community was of value to even the youngest boys: they became producers of community recreation as adults were of celebrations at other times of year. For the most part, Pollett's activities during his childhoot were limited to those shared with the very young and the very old. Calendar holidays such as Christmas brought boys into contact with the wider community and its rituals and traditions, usually as observers of parades and janneying. At Christmas, "times" exclusively for children were held by the two local lodges, and other annual planned activities also were provided for them. Pollett appears to have looked forward to these occasions as much as any other child; but in his writings, those activities devised by and for himself and his peers appear to him in retrospect as "a treasure of memories." Developing an expertise in faiting, exploring New Harbour Barrens, participating in Bonfire Night autics, and his other activities with his peers appear to have made New Harbour in his eyes a place of wooder and a world to matter.

1.

By 1907 Foliett had began his years of formal education at the village's Church of England school. The schoolhouse, located across the harbour sear the Church of St. Augustine, was open to children of all ages. Annual payment of school fees and purchase of texts were required from each child, as were daily contributions of wood junks for the school's stove. These expenses, as well as time spent for classes and study, prevented nearly half of New Harbour's school age population from attending regularly. Contemporaries of Pollett's, Hananh Brown and Charles Hillier, note from their own experience that work demanded from children in the home often took precedence over their education. The Pollett was not affected by this practice, however, for although his family was no more prosperous than most in New Harbour, his mother, herself illiterate, apparently insisted upon the need for educating the children.

Pollett's first teacher, Colin Jones, taught at New Harbour from 1906 to 1909. It like most male teachers in that day, his tasks and responsibilities included the teaching of students at all levels simultaneously, tending to administrative concerns, and setting an upright example for the children and for the community. Jones appears to have been a demanding and attentive teacher of the roughly severaly pupils in his care. In ansual School Inspector's Reports, his devotion to his tasks was consistently praised by Superintendent William Blackall, who noted that Jones was responsible for raising attendance levels and improving the quality of teaching. Blackall stated

that "Mr. Jones, an energetic teacher, gives his whole time and attention to his work," ⁶⁰ Owing to Jones's influence, maps and books were purchased for the school, and Newfoundland history, Bookkeeping, and Algebra were tranght for the first time along with the basic subjects of reading, writing, and arithmetic. The teaching of these subjects may have encouraged curiosity above places distant from New Harbour and the acquiring of skills not ordinarily employed in daily life. How these concerns affected Pollett while a young student is not known, but Jones himself appears to have been a decided influence upon Pollett's developing character. Jones seems to have provided a new adult model for him, for acquaintances recall how Pollett deliberately learned to refine his "barmant's accent." Jone scenario the habit of wearing a necktic in cubil."

Jones possibly motivated the community to build a new two-room school, which was in operation a year after he had departed from New Harbour. It replaced the dilapidated poorly lit. and inadequately furnished small school in which Pollett had begun his education. It onesed in January, 1910; and in April was converted into a high school to serve the entire New Harbour school district from Cavendish to Spread Eagle, Trinity Bay, Schooldays perhaps were made much more pleasant in this comfortable setting. In 1910 the new school was praised by Blackall after his annual visit as "splendidly spacious and bright" and built on "one of the prettiest school sites in the country."51 This location provided a setting compatible with what Blackall believed to be a necessary adjunct to fundamental classroom courses, namely local nature studies. Blackall offered assistance to pupils and teachers, a number of student prizes, and detailed instruction for the undertaking of these studies; and he gave assurance that "That which is most needed is a pair of wide-awake eyes." Although Pollett's adult writings suggest a wide and intimate knowledge of the New Harbour landscape and its wildlife, formal teaching of this subject may not have taken place during his school years. Blackall's ardent national campaign for the undertaking of local nature studies probably met with as little success in New Harbour as eisewhere; though for a time, Arbor Day, a school holiday in spring, was devoted to the planting of trees and caring for the schoolgrounds.

Generally, studies during Pollett's school years were devoted to the contents of the Royal

Reader, which were a set of seven school testbooks. The initial Reader, the "Primer," "was the book from which children learned the alphabet and simplest words. Readern one to three comprised mainly lessons in grammar and English literature. These subjects were learned by rote and through oral and written exercises, as were spelling, counting, and writing lessons. The higher the number of the Reader, the more difficult were the lessons. In literature, for example, the simplest stories such as "The Spider and the Fly" and "The Dog and the Shadow" were intended for beginning readers. These simple fables, in their brevity and didacticism, were suitable reading materials, for they were meant to build moral character in the young children by instilling virtues such as modesty and discretion.

The higher numbered Royal Readers, four to six, contained lengthy and more difficult literature and grammar lessons. British and world history also were included, as were other subjects such as cultural geography and elementary natural science, appearing under the headings common to all Readers, "Useful Knowledge" and "Great Inventions," and which probably were instead to serve to expand the students' horizons beyond the local and ordinary emphases of earlier Readers. ⁴⁸ Whereas lessons pertaining to parts of the house and attributes of the seasons were contained in beginning Readers, by Reader six lessons related to the history and development of papermaking and printing were among those which students were required to learn. These subjects were taught in the second classroom of the two-room school, separated by a folding screen and reserved for higher graders; but practices of rote-learning and daily oral and written exercises were common in both classrooms.

Each perface to the <u>Readers</u>, from the <u>Primer to Reader</u> six, contained explanations such as these: "The whole book, like other books in the <u>Series</u>, has been constructed with a view to induce children to take a real interest in what they read, and to make them delight to exercise their power of reading," and "The lessons aim not only at teaching the <u>art of reading</u>, but at training the pupils to a <u>love of reading</u>." ⁴⁰ Pollett in his early school years did develop a love for reading, and eventually acquired a small book collection. One book, <u>South and his Priends</u>, an adventure novel about a young <u>English</u> boy's experimence in the Swiss Alps, was given to Pollett as

a prize for his distinguished achievement in 1912 in the subject of Geography. Other books in his collection included one called Business and Law, and a volume of treasure stories given to him for selling a large number of subscriptions to a magazine, The People, 80 These books, like the subjects of his school studies, may have broadened his knowledge of the world outside New Harbour and created an awareness of differences in life elsewhere. The adventure novels of Norman Duncan and the Horatio Alger stories, which were available in his school's small lending library, also may have played some role in shaping Pollett's ideals about a richer variety of experiences than he could know in New Harbour. And the contents of the Royal Readers, each volume of which students had to master before proceeding to the next number, also appear to have influenced Pollett's sentiments and imagination. Pollett later recalled his morning walk to school carrying his book satchel filled (figuratively) with companions, the characters in the poems of Wordsworth, Shelley, and Tennyson. As he crossed New Harbour Bridge, he remembered thinking about Lucy, of "Lucy Gray" by Wordsworth, who was lost in a snowstorm as she made her way across a wooden bridge in her own town. He wondered if her fate would have been different if she lived in New Harbour and had crossed New Harbour Bridge. 87 In this instance, Pollett in his imagination succeeded in incorporating into his own place the figures and setting of quite another world. Indeed in his daily conversation, as his cousin Marion Gosse recalls. Pollett would dwell on the ordinary aspects of daily life that other local people would not notice.58

The ability to learn by observation and to exercise his imagination may have quickened his grasp of more formal subjects in which he was trained. During the years 1912 to 1914 in particular, he excelled in Geography, English, Algebra, and History.³⁸ He passed all his other subjects. Owing partly to the discipline and rigorous demands from two of his teachers. Samed Brett, who taught at New Harbour from 1911 to 1913, and Walter Scammell, who taught from 1914 to 1916, Poliett appears to have been an exceptional student for his day. These two men guided Poliett through his studies for his three Council of Higher Education (C.H.E.) Examinations—Primary in 1912, Preliminary in 1913, and Intermediate in 1914—in all of which he excelled.⁵⁰ A second influence which may account for Poliett's achievement was a competitive atmosphere in the classes.

room. A number of Pollett's contemporaries, Llewellyn Thorne, who was his closest friend, Herbert Cranford, and Selby Newhook, also distinguished themselves in their studies. They may have provided some incentive for Follett to achieve recognition in his yearly course of studies.

In the spring and summer of his school years, however, Pollett's love for formal learning, like that of most boys, appeared to decline; he became impatient for the end of Jane to come in order to spend his time fishing for trout off New Harbour Bridge and in the Barrens, and for cod in his father's punt. The time when he could "switch from sharpening pencils to polishing fishhooks" was eagerly awaited. As his acquaintances reall, each spring morning Pollett fished on the bridge in the hours before classes began, and left his trouting pole on the bridge in order to save time upon resuming his fishing at the end of the school days. Spring and early summer school days, which interfered with regular outings with his father, were viewed almost as dismally as Sundays, when, according to local custom, no work or fishing could be undertaken. On Sundays from the bridge, as Pollett later recalled, one could only observe and measure the enormous learts of Shi in New Harbour Pond, which were sure to be sone by Mondays.

Yet at some point during his later school years, Pollett discovered that his love for fishing was confined to an entertaining pastims; never again did he foster a desire to become a fisherman like his father. "He didn't like the work," one informant says simply, "I he loved to fish, but balked at the labour involved in the cleaning and curing processes which took long hours of exertion to accomplish. Nevertheless, "Jim Pollett made Ron work" on his boas and flakes," As well, Pollett's labours in the woods and in the gardens had increased by the time he had reached about age twelve. He was now expected to cut the wood for his family and haul it home with the use of his work dog Downey and his catamaran, and to contribute to the garden work as a participating adult. There is evidence that he felt burdened and frustrated by this drudgery: Pollett said that whereas he once had looked forward eagerly to his days at the family garden across the harbour, by this time, when "back-breaking," extensive manual labour was demanded from him, he would have preferred to run the other way (perhaps back home, toward the bridge, "Apparently his concerns and interests were moving gradually away from the constraints imposed by family life in

New Harbour.

Poilett differed from both of his parents also in that from an early age he did occasionally so to church; and one annual ritual to which he appears to have looked forward was presenting the minister James Johnston White with his first trout catch of the season. From White he received a nat on the head and a dime, which he denosited in the collection plate on the next Sunday 67 He appears to have regularly attended Sunday school classes, too, which were directed by the local merchant's wife. Mrs. Fred Woodman. In 1912 Pollett was given a book prize by Mrs. Woodman for his distinguished achievement in the high school subject of Geography, although no prizes usually were given at the Primary level or for this subject. Perhaps Pollett's achievement in Sunday school classes may account for this exception, for the giving of "reward books" in recognition of students' merit and diligence was a standard practice at Sunday schools in England 68 Some forty years later. Pollett would recall the rarity of receiving presents in any form as a how even at Christmas he wrote "Santa Claus travels ever so light" to the outports " Indeed this incident of his receiving a "reward book" was imaginatively recreated in his short story "The 'Dicky-Bird' Dish" in 1949.70 Its rather didactic theme is splendid reward gained through the good works of a young boy who possesses promising scholastic abilities. The notion that achievements pursued in the church-related school could foster possibly the only tangible rewards that an outport boy could expect, was one that Pollett as an adolescent was perhaps gradually made aware of.

For by 1914, if not before, the spheres of domestic responsibilities and academic achievement appear to have become very separate in Pollett's mind. Actious manual labour associated with his tasks may have interfered with his academic work. As he recalled, "the stoves were always hungry, the gardens forever glutted with weeds," To Owing to his recognized achievements in his studies, Pollett had probably become aware of the limitations imposed by New Harbour life on his developing aspirations and future accomplishments. His achievements in part may have led him to develop interests in acquiring skills of more diverse natures than the traditional one employed in New Harbour daily life. In the next two years, he was to search for possible outlets v

Between 1914 and 1916, Pollett prepared himself at New Harbour to become a school-teacher. His choice of profession was a fairly predictable one. As Patrick O'Fisherty has suggested, schoolteaching was one of the few "secape routes" available to an outporter who wanted no part of the traitional occupation of fishing. To Pollett, however, was not eightle to obtain a teaching certificate until he reached the age of sixteen. After his completion of the Intermediate grade in June, 1914, and a summer of fishing with his father, he apparently arranged with his schoolteacher, Walter Scammell, to receive private instruction in more advanced academic courses. Pollett had intentions to write examinations for which he would receive a Junior Associate certificate, in courses which the New Harbour high school was not certified bigh schools stewhere would have led to expenses which his family either could not afford or was not willing to meet.

Instead, Scammell instructed Pollett during the regular school day and in return Pollett
ansisted him by teaching children in much earlier grades. Clareace Cranford, who had been one
of these pupils, recalls a rather disdainful attitude felt toward Pollett, an ordinary fisherman's son
acting, as schooleacher, and who spoke with a "twang". "We used to throw wads of paper at
him," he remembers, when Pollett's back was turned as he wrote on the blackboard." Although
schoolteaching was a respectable position, all of New Harbour's teachers had come from outside
the community, and perhaps never were challenged in quite this way. By assuming this position,
Pollett had overstepped the social barriers between traditional Sherme and local notables—merchant, clergruman, teacher, and a handful of others—which historically had developed in the village; nevertheless, doing so enabled him to prepare himself for his new profession.

These two years of student teaching apparently were considered to be adequate training for a teacher in the outports. Training as such, in fact, was not required until 1920; and Pollett had perhaps sought out this opportunity independently to gain classroom experience which might eventually help him in his chosen career. An able student and avid reader, Pollett probably had a close relationship with Scammell, who as a teacher was himself consistently given excellent ratings by Superinteedent Blackall for his teaching in all subjects and for maintaining discipline and order in the classroom.²⁸ Apart from his interest in Pollett, Scammell also may have encouraged other slightly older boys (who also were Pollett's closest friends) to enter the teaching profession: by 1916 Liweslips Thorns and Herbert Cranford had become outport teachers, and Selby Newbook also trained (more formally, at Bishop Feild College in St. John's) for the profession.²⁷ This seems remarkable in that it was unusual at New Hardour for a student to get as far as the Intermediate grade, and still more so to attempt further academic achievement.

During Pollett's two years of student teaching, a number of his peers volunteered for service overseas in the Great War. Newfoundland, as a British colony, entered the war on August 4. 1914 but men from the outports were more slowly recruited than residents of the capital of St. John's The Cadet, a military newspaper of the day, cites some reasons for this discrepancy; outport families were greatly dependent on their sons, and no provisions could be made for the families' support while their sons were overseas: furthermore, recruiting agents rarely went beyond the principal towns, and no recruiting campaigns or clubs like those in St. John's were established in the outports.78 Frederick Rowe notes, however, that an intensive recruiting program was undertaken island-wide in 1915.79 Friends of Pollett's such as Herbert Cranford, and some other New Harhour young men who like Pollett were not of legal age to enlist, may have joined the forces as a result of this program. Certainly when they joined in or before 1915 they were made aware of events which had occurred in the war from daily newspapers available at New Harbour; news also was posted daily at the village's telegraph office. Pollett tried to enlist perhaps in 1915, but he was rejected for service owing to an abnormal condition of the feet. He had been born with a convenital deformity common in the Pollett lineage: two knob-like growths near his instens which. although they in no way impeded his mobility, caused his feet to turn inwards. 80 His ambition to become a soldier was thus thwarted, and he returned to school and resumed his tasks.

In 1916 Pollett became the school's Supervising Officer during the examination period held

for the district at New Harbour. This was a responsible position, and its detine establed attending on examiner, collecting the examinations, and arranging for their dispatch to the Registrar at St. John's.¹⁸ His teacher, Scammell, had probably proposal his name to the school board, its members comprising the missiter T.W. Upward, merchant Fred Woodman, and a sumber of other prominent community residents. Their approval suggests that Pollett was seen as a competent and respected young man by his elders. As Pollett was to tarn sixteen years of age that Juse, Scammell possibly also had recommended Pollett to the Board of Examiners at St. John's, from whom he obtained his teaching certification by January of 1917. However, in part because Scammell was to join the armed forces in the fall of 1916, Pollett's two years of teacher-training and private instruction ended.

A sumber of events during the late spring and summer of 1916 also affected Pollett and his ambitions. In May his mother died; she was a victim of a mealer epidemic which had spread throughout Trinity Bay and the island. A notion which perhaps added to Pollett's grief was his knowledge that he had brought the meades into the house, having caught them at school. This experience probably accounts for his failure to write his Justice Associate examination in June, he had intended. In addition, his grandmother died in July of that year at the age of ninety-six. Although the had been bedridden for many years, he apparently had continued to think of her as an important member of his household. Both deaths, furthermore, appear to have affected his father Jim tremendously, for he was the only wage earner, and now was solely responsible for his family of seven, which included a young son Len, aged two. Jim Pollett's rigorous demands upon his family, given the circumstances, could only have increased at this time. More than ever before, perhaps, he impressed upon Pollett the need to start exzing his own living.

In the summer and fall of 1916 Pollett assisted his father in all of his chores, and obtained seasonal work building snowferees for the local railway. A branch line from the canantional railway, from Whitbourne to Heart's Content, had been built and was in operation by July of 1915. ⁵⁴ Pollett joined a number of his peers who also had obtained similar jobs. This paid labour enabled him both to satisfy his father's wishes and to establish himself among other boys of his own age group. In his adult writings, while he makes no mention of the death of his mother, he recalls with delight the disposition of his friends, these "eagle-eyed bedinners of the depot station" who appeared worldly-wise and "ripe for anything that smacked of storybook adventure. "⁶⁶ They, too, were waiting for opportunities to earn their keep; they probably had in mind permanent jobs in Grand Falls, where their fathers and relatives had found employment for some time.

Though somewhat apart in his ambitious from most of his peers, Pollett enjoyed the selfmade entertainments of the group during that summer and fall. Summer evening sessions at New
Harbour Bridge included storytelling, dancing, and sing-alongs. As avid a participant in these
activities as he had been in sports and pustimes with his friends years before. Pollett was know
for his ability to make up rhyming stories about recent local events and to perform them for the
gathered group. Jim Thorne recalls one of Pollett's poems, "Beling," which described humorously
the actions and maneuvers of some local boys who set their ecl-lines on New Harbour Bridge, as
boys always did each summer evening." Storytelling such as this was intempersed with jipstepjulg on the wooden planking of the bridge, and singing, for which local young men, particularly
Aubrey John Woodman, provided accompaniment on the violin or concertian. In cooler weather
these nightly meetings were relegated to family houses. Bertha Pollett recalls many imprompts
concerts at Ron Pollett's home, noting Pollett's lower of performing for the assembled group stories
and jokes of his own devising, and of singing locally composed songs, one of which was "Nicholas
Tohin..." Nightly card games and weekly dances at local stores and the schoolhouse were other
diversions for the group.

This before Pollett left New Harbour in January of 1917, he had experienced several seasons of labour as a participating adult at home and at the nearby railway, and from years of close observation and listening had learned to contribute to local entertainment imaginative literary diversions, of which "Eeling" is one example. Remembered by his contemporaries, Ralph Higdon and Reg Woodman, as a quiet and good natured young man, a hard worker and a scholar, and by Bertha Pollett and Jesse Fogwill us a friendly, well-managered young man who never wore and

had a pleasant disposition, his personal qualities were ones that members of his community found attractive. Profilett's ambitions nature, however, is the quality for which he is best remembered by all those who knew him. That he never fostered a desire to settle at New Harbour during his young manhood is one notion about which his peers are certain. Instead, he was the first of six of his family to abandon the traditional outport way of gaining a livelihood, namely the cod fishery, and to seek our opportunities and challenges elsewhere.

LEARNING HIS TRADES: THE TRANSIENT YEARS, 1917-24

Roa Polleti's life between the years 1917 and 1924 was characterized by job-husting and moving from place to place. In January, 1917, he began a brief career as an outpoot scholarscher at English Harbour, Trinity Bay; two years later he held a similar position at Green's Harbour, Trinity Bay. The challenge of developing other skills and exploring more distant places may partly explain Pollett's search for responsible positions elsewhere in the summer of 1910. He obtained one such post as a record-keeper in the offices of the A.N.D. Company at Grand Falls, and he lived and worked in this town for more than three years. Subsequently, in 1923-4, Pollett served an apprenticeship in linotype craftsmanship at Montreal. His mastery of this trade, which in the 1926 was an integral part of the printing industry, facilitated his engignation to New York. In Glen Cove, Long Island, Pollett began to practice his profession of linotypist, a trade which enabled him to realize his ambitions and, finally, to begin to establish himself firmly in the world beyond New Harbour.

1

Ros Pollett arrived at English Harbour in Jassuary of 1917 to take the position of schoolteacher at the local All Suists' School maintained by the Church of English English Harbour, located near the town of Trinity on the north shore of Trinity Bay, was a fishing community of about four hundred people. The history of English Harbour dates from .1675, but the village had not prospered substantially. Because all trade was linked with the centre of Trinity, no local merchant firms operated there, and until the twentieth century there was rarely an opportunity to obtain fall-lime employment, apart from the fisheries, except for the position of schooleaching. At the time Pollett embarked on his journey to English Harbour via the newly instituted Heart's Content branch line of the transactional railway from New Harbour, the majority of young adults from English Harbour had left there on the slightly older Bonavista branch line. 'The steam trains had effectively become a symbol of opportunity in distant lands for many Newfoundlanders of Pollett's generation. Pollett later recalled how as a boy of about twelve, he had wasted with faccination the smoke of the Bonavista branch line train, which he could see at a distance from his family's garden on New Harbour Point,' and for him, some four years later, perhaps the area across Trinity Bay encompassed the scope of his ambition to explore the larger world.

The place represented a striking contrast to the physical area with which Pollett had been familiar. The name of English Harbour is a deceiving one, as Ian Penney points out, because there is no harbour in the community:4 the place is exposed to the open sea, the appearance cliffy and rugged. Pollett boarded with the family of Lucy and Frank Penney, whose house was located among the cluster of homes in the village, which were connected by numerous narrow footpaths near the gravel main road along the shoreline. The family's daily routines were familiar to Pollett for, like the people of New Harbour, they were dependent upon the fisheries, but his new role in the community set him somewhat apart. Schoolteachers were accorded high regard there because their occupation was relatively inaccessible to the ordinary young person whose tasks were confined to traditional nursuits, and because teaching demanded a close association with church interests. Apart from daily teaching. Pollett was expected to assist the local minister. Arthur Pittman, by acting as lay reader each Sunday at services held in the local church. He also served the church vestry by recording the minutes of meetings. These records written in a fine hand still exist in the local church, and the copious detail of the meetings' proceedings recorded by Pollett suggests that he served the vestrymen well. A remarkably extensive and sophisticated vocabulary for a young man can also be noted. And there were still other duties expected of Poilett in English Harbour. Each week he taught the parish children Sunday school. Like all Newfoundland teachers, he also was theoretically responsible for presiding at the burial and baptism of members of the church. Pollett was only sixteen and his exalted new role in an unfamiliar community presented great challenges.

With two years of student teaching behind him, he was anxious to assume fully the duties and responsibilities of the profession. Pollett's daily tasks at the school occupied most of his time. although his classes were comparatively small; he had twenty-eight students during his first year of teaching, and thirty-serve during his second. He was the only teacher for the one-room school, however, and therefore he was expected to teach and to supervise each of half a dozen grade levels insultaneously. His first experiences were no doubt difficult since he began midway through the school year, and needed to ensure that his students had mastered the first halves of their yearly <u>Readers</u> before proceeding to teach them the concluding halves. To help them to accomplish this, Pollett made his time available to all students both during the seven-hour school day and after hours. "He was always there if we needed him," one former student, Jo Aan Penney, has said, recalling Pollett's constant attentiveness to them." Facilities must have caused great discomfort to Pollett no less than to his students. "In the students sat for long hours on wooden benches, each of which accommodated six children. They did their lessons in the dilapidated school building which was inadequately heated by a single woodstove, and was poorly lit by several oil lumps. By day and night this paucity of physical amenities may have affected students interest in the clasurout, and thus Pollett's tasks were made fairly formidable.

From the beginning, however, Polletts seems to have been determined to do his job flawlessly, and an indication of his abilities may be found in the annual school reports. In 1917 Pollett's teaching of union rand seems studies, his maintaining of claseroom discipline and orter, his keeping of the school's register, and his performance of other duties all were given excellent ratings by Superintendent Blackall. Such ratings, on a scale of one to five, or excellent to poor, were electroniced both by annual progress reports which teachers submitted to Blackall, and by Blackall's personal wint to each Church of England school in Newfoundhand. Both forms of report enabled assessments to be made of the teacher, the students, and the schoolhouse. The visit, however, probably was the more valuable source of information, helping Blackall to form his own judgment of the teacher's abilities and weaknesses. During these visits, Blackall scrutinized the performance of students; it would seem that the students abilities to answer Blackall's questions and in other ways to demonstrate their progress were critical in shaping his opinion of the teacher. As the report of 1917 indicates, Pollett and his students performed extremely well during

the visit, although the rating for the school building was very poor. ¹³ Like his own former teacher Colin Jones, Pollett may have helped to secure a loan of \$100 to be used for the building's improvement.

The meagre salary of \$259 per annum which Pollett earned for the variety of tasks which he performed in the school and church did not dissuade him from conscientious work. In 1918 Blackall in fact noted the difficulty for the school administration of keeping qualified neonle in the profession giving as the reasons teachers' noor salary and noor prospects 14 Lack of incentives in addition, may have affected their disposition toward the work and perhaps toward the children; but if Pollett responded in a negative way his students were not aware of it "They didn't make teachers nicer than Mr. Pollett." Debbie Butler has said: she was one student to whom Pollett taught the alphabet and other early lessons. 15 He was relatively patient and even tempered yet exacting. Gertrude Bugden recalls that Pollett was angry if students did not have their homework done, although he never became excessively upset or resorted to physical punishment. 26 As well, in her case he understood that poor evesight presented difficulties in studying by lamplight during the evenings and that her family could not afford to buy her evenings and he was not disposed to respond harshly to boys, in particular, whose work at home precluded regular school attendance. Thus it appears Pollett understood well the constraints on education imposed by local economic conditions. His performance seems to have paralleled the achievement of men like Jones, Brett, and Scammell, who had been his own teachers. Pollett had learned from them the skills of the profession; and he was able to impart a love for learning similar to that which they had implanted in him. At least three girls whom he taught later became teachers themselves and one of them. Jo Ann Penney, attributes her ambition to Pollett, whom she afterwards took as her model 17

Evidence suggests that in his first year in the profession Pollett proved himself an able teacher and provided a positive example of character and behaviour to the students his care. If Pollett had left home, not merely to test his abilities as a teacher in a new place but possibly to discover adventure, then he was no doubt disappointed by the ordinary, traditional lifestyle which now restricted his activities. Much of his time was spent with the children outside the regular school hours each week he taught them Sunday school, and he supervised occasional school picalica and achool concerts. Familiar pastimes such as dancing and card-playing, which he had always enjoyed, were disapproved of by local parish officials, and as teacher he was expected to adapt to the derout character of the community where most social activities were organized and supervised by the ciergyman. ¹⁸ He therefore apparently spent his free hours at the limited pastimes that were acceptable at this village; reading at the Peasey home, chatting with the family
and visitors, and in warm weather taking long solitary walks in the surrounding woods and hills. A photograph of Pollett from the period depicts him as a conditiont and distinguished gentleman,
holding an opea book and sitting on the grass in the countryside of Champeys West, chin arched sideways and gaze focused on the nearby hills. ¹⁸ He appears to fit well the stereotype of dillideat
schoolmaster. His students recall him as a handsome, dark-haired young man who was always
well-dressed, wearing 'his Sunday clothes everyday.' He was a very quiet man, one former stadent recalls; and the says further, "There's not much to be said about him—only on the good
side. ¹⁸⁰ Pollett remained as sole teacher in English Harbour until 1918.

ï

When, during the summer of 1918, Pollett learned that a teacher was needed at Green's Harbour, Trinity Bay, he resigned his position at English Harbour, It was at his home in New Harbour, apparently, that Pollett learned of and obtained the new position, for St. Matthew's Church of England school at Green's Harbour was part of the New Harbour school district, and the church was served by the minister who resided at New Harbour. Pollett had returned to New Harbour in the summers of 1917 and 1918 to spend time with his family and friends, and to resume his favourite pastimers: thing at familiar spots, dancing, card-playing, and stoytelling with companions on New Harbour Bridge. Teaching at Green's Harbour would make these homely pleasures accessible throughout the year, for one could use the daily train between the two communities or else walk the distance of about five miles. A slight increase in his annual

salary to \$326 was an incentive for him to make the move, the but Green's Harbour's proximity to

New Harbour was no doubt also a factor.

His ties to New Harborn beloped to make establishing himself at Green's Harborn in September, 1918, a relatively simple process. Pollest arranged to receive room and board from Sarah and George Harsum, a resident family related to his mother. Other residents show that him beforehand, for local young men frequently came to New Harbour to take part in Staurday night entertainment on the bridge, and some women from New Harbour had married into local familia.¹⁸ In these circumstances Pollett was not forced to cope with as many challenges as he and excountered at least initially in English Harbour. The physical geography also was familiar. Green's Harbour was a community spread out along four miles of a horseshors-shaped cove; flakes and stages bordered the shoreline, above which were the house, and the land beyond rose to a rolling hill topography. In terms of physical appearance, the fishing village, comprising a population of about 660 persons. It resembled the beauties of New Harbour, and prhapa after his twoywas absence from home, young Polletts to longer sought new vistas.

Pollett's new working conditions at first must have appeared ideal, for although his salay, was increased, he was expected to teach only about half the number of pupils he had taught in English Harbors, he average attended was now seventeen and most of these pupils had been through only the first few Boral Readers. It appears, however, that their low level of achievement was due to slow progress previously. One student, Bertha Burt, can recall her own and other students' great difficulty in mastering Reader, three during Poliett's year three. She has said, "I say rapide three Reads algoring to put the said of the grade three Reads algoring the said and the said was the said of the said of the said of the said of the said to the said to the said of the said to the said to the said of the said to the said the said to the said the

There is no evidence to suggest that Pollett, now aged eighteen, made his time available to

students after regular school hours as he had at English Harbour. Nor is there an indication that his help was sought. For the most part, Pollett simply performed his required tasks of teaching daily all grades and all subjects in the one-come school, teaching the children's weekly Sunday school classes, and conducting the Sunday service at St. Matthew's Church. He left he school promptly after each school day, and such to the local telegraph office where news from the outside world was posted daily. In this way he kept in touch with developments in the Great War and other current issues. He had made a point to announce to his pupils that he had been rejected for was service owing to, he said, "flatfootedness," and perhaps his thwarted ambition to become a soldier now resurfaced as a keen regret. He also was made aware that many parts of the Newfoondland economy were experiencing a "boom" as a result of wartime prosperity, and he may have now begun to feet that his position and responsibilities as a teacher demanded more sacrifices than other modes of employment. In

Other circumstances during Pollett's year at Green's Harbour may have focused his attention on the limitations of the profession. Although his new position enabled him to resolve the
dilemma with which he had been faced while an unpaid student teacher at New Harbour- to be
able to eajor the delights of his outport home and yet also to secure a satisfying, salaried position
away from the fishery—Pollett probably was aware that his father Jim disapproved of his son's
choice of profession. The choice brought about a soticeable strain in their relationship. On one
occasion, Jim and other members of Pollett's family were annoyed at a joking remark Pollett
made at the diamer table to the effect that as he was so longer a fisherman, he could not distinguish a haddock from a herring. To Jim's mind, not only was his eldest son neglecting the raw
opportunity of prosperous years in the fishery, but he seemed at times to disown his heritage.
Jim's attitude, together with a recent addition to the Pollett household—namely, a housemaid,
Sophia Piercey, from nearby Wilaterton—may have made weekend and holiday visits home
difficult for Pollett. The high standards of moral conduct which were demanded of Pollett in the
teaching profession possibly had caused him to volunteer some criticism about his widowed
father's and housemaid's living situation. Evidently the situation had become a subject of village

gossip.²² It is possible that for this reason young Pollett confined his activities to out-of-doors pastimes, a circumstance which probably quickened his desire to break home ties, and to re-evaluate his plans for the future.

It is perhaps revealing that Pollett did not take the opportunities available to him to advance himself in the teaching profession; instead he continued to hold a Grade Three certificate. This level of certification required simply that a candidate hold a Preliminary diploma (which Pollett received in 1913), and he "qualified, in a reasonable measure, to assume the responsibility of taking charge of a small elementary school." Advancement to the Grade Two level, which offered a rather substantial increase in salary, required candidates to hold an Intermediate diploma (which Pollett received in 1914), and have a half year of teaching experience. Pollett obviously met these two requirements to obtain Grade Two certification, although additional requirements may have been in the form of prescribed courses in advanced teacher training at St. John's and subsequent formal examinations. **Pollett's meagre salary may not have been sufficient for him to undergo the latter requirements. Fairly poor prospects for financial and intellectual advancement in the profession thus may have caused him to lose latters in his demanding work.

As a result of these circumstances, perhaps, Pollett and his pupils did not fare well, overall, in the annual school inspection by Superintendess Blackall. One student recalls the difficulty students had in answering the questions which Blackall posed to the class. In Blackall's personal assessment, Pollett thus was given a number foor rating, "fair only," for his students' progress in justice and senior studies; number two rating, "good," for his maintaining of discipline and order, teaching of divinity, and supervision of students' written work; and only in one area number one, "excellent," rating this for the keeping of the school registers. In Blackall responded to the major weakness he observed, the students' general lack of progress in their studies, by sending to the Green's Harbour school a new set of school text called the Princess Reader. These texts were similar to the standard Royal Readers except that they contained a greater number of illustrations, which Blackall perhaps felt would make schoolwork more interesting." As such, Pollett probably found no difficulty adapting to the new set of texts. Students recently that he was pasted

with them, and he did not have a strict classroom manner; furthermore Pollett was very efficient and he "knew how to organize things well." ³⁸

In addition, he tried to make the school year more entertaining by organizing sports and games, and several students recall a school concert which he arranged in the fall of 1918 to aid the war effort. Pollett taught his students the "Battle Hymn of the Republic," which they sang at the concert. One student recalls his wide smile and gleeful expression when he was conducting the chorus: "we really sang out the words 'Giory, glory, hallelujahi" so Sch respites probably helped to make Pollett's year at Green's Harbour occasionally fairly pleasant. If he had intended to stay at the village, the rostine tasks at the school and church which he would have continued to stay at the village, the rostine tasks at the school and church which he would have continued to stay at the village, Pollett chose to abandon the teaching profession after his one-year stay in the Green's Harbour community. By the end of the school year, an alternative for which he may simply have been validing became available. He was able to secure a position in the offices of the A.N.D. Company at Grand Falls. This was an opportunity which demanded a different set of skills, and once again, a move to a new community.

iii

Pollett arrived at Grand Falls in central Newfoundland early in the summer of 1919, this journey being the furthest from New Harbour that he had yet undertaken. Grand Falls was quite unlike the fishing villages which Pollett had so far experienced: its population was more than 3700 persons, and from its inceptions the A.N.D. Company paper mill was the community's migrossorre of employment. Alfred and Harold Harnoworth, British newspaper magnates, ewend and managed the entire town as a result of an agreement with the Newfoundland government made in 1905 to develop the woodlands of central Newfoundland. By 1000 the paper mill at Grand Falls was the largest producer of awayprint paper in North America. Although the Harmsworths' establishment of an inland town was anomalous in the pattern of settlement in Newfoundland, both in terms of its location away from the sea and its means of year round employment, it

readily attracted outporters seeking to improve their lot. In 1910, Lord Northcliffe (formerly Alfred Harmworth) axid publicly: "We have two objectives in view is Grand Falls, one is the making of the best paper in the world, which we are doing, and the other is to create a happy community." The latter objective was accomplished in ways unique to Newfoundhand like. Buildings at Grand Falls were equipped with conveniences such as electricity and running water, and among Northcliffe's projects was the establishment of a town club equipped with a library and a howling alley. If The facility was open to all employees of the company, and it was only one of a range of the town's leisure structures. At late as 1937, Joseph R. Smallwood was unable to compare the town with any other place in Newfoundhand, naying of it, "Like a little world in itself, is Grand Falls."

In one sense, Pollett's move there was part of an established pattern among some dozen New Harbour men who arrived at Grand Falls at its inception, many of whom eventually obtained permanent positions usually as carpenters, millwrights, and general labourers. It was a common practice, according to oral report, for these men to inform their fellow villagers of positions about to become available. 45 Pollett was aided in just this way in 1919: a New Harbour friend. Ches Woodman, employed as a paper inspector, told him of an opening in the company offices. 45 Woodman, some ten years Pollett's senior, may have also helped him to obtain lodgings at the Exploits Hotel, which catered to unmarried men employed in high ranking positions at the paper mill. The Exploits Hotel was owned by the A.N.D. Company and operated by a plump. cheerful woman named Maud Burry. There Pollett was able to enjoy all the modern conveniences of the town, and yet to some degree maintain the lifestyle to which he was accustomed. Charles Edwards, who boarded at the Exploits Hotel, calls it "a typical Newfoundland outport hotel;" at a cost of less than \$1,00 per day, men were given pleasant sleeping quarters and were served large meals of Newfoundland-style foods. Parts of the premises were given over to activities such as card-playing. Such rooms were equipped with woodstoves and comfortable furniture; and a number of former residents recall a boisterous, smoky atmosphere in these rooms, as men played cards and smoked cigarettes together on winter evenings. The hotel was located at Church and Riverview Roads, which was a convenient location for the boarders since the area was only a short walking distance from the mill.

Pollett arrived during the town's second major period of growth, for at the time two additional paper machines were about to be installed at the mill, and a second section of the town was being developed. Be the dobtained a post as record-keeper in the company's Orders and Shipping department, an differ located sext to the Finishing Room, where rolls of paper ere readied to be shipped to England. Pollett's duties establed recording the numbers, weights, and destinations of these rolls, among other record-keeping tasks. This was a responsible position; the additional machines which were in operation during his years there had brought about a substantial increase in paper manufacture. It is 1920, for example, the mill produced 63,000 tons of paper, the recording of which was the task of Pollett and his three co-workers, Dave Green, a Mr. Forbes, and a

Pollett's background in education was no doubt a valuable asset in his zew job. As far back as his school years, he had shown exceptional mathematical abilities, and his new job. As far back has he had shown exceptional mathematical abilities, and his new job. The chance to use them. He must also have been pleased to have a substantial increase is income. The exact salary Pollett earned is not known, although it probably ranged from fifty creats to one dollar per hour for an eight-hour work day. He sarnings, using a most conservative estimate of \$1000 per year, would have totalled three times the wages he had earned as a schoolteacher. The majority of workers at Grand Falts, in contrast, joined the company as manual labourers and earned from ten to twelve cents per hour, until they proved themselves capable of performing more responsible duties. Dut such opportunities for advancement were made possible quite readily to all workers in a number of ways: a night school had been instituted by Lord Northcliffe, and the expansion of the mill operations creased various openings in all facets of production and management. This emphasis on personal achievement and industrial development was one Pollett probably found attractive. "If he was a quiet man," one of his friends, W.T. Howell, said of Pollett, "the also was an ambitious one." **

Writing, evidently, became one of Pollett's pastimes, for although his work was exacting, his

new job afferded ample leisure hours. He wrote out his work and shared it with friends, who in turn passed them on to other friends. Eventually, it is said, he became known in the community are "as brilliant fellow with a pen." Although there was no local newspaper in Grand Falls, some recidents think Pollett may have also written regularly for a newspaper in St. John's and thus became even better known as a writer. However, the only known example of his writing from this period is a poem that he wrote about life in the Exploits Hotel. In it, each boarder at the time was described by same and characteristics, as were other members of the household, such as "the little waitense who could shuffle plates." No copies, apparently, were printed, but handwritten copies were given to peers, who can recall its humorous qualities.

Those who knew Pollett in the Grand Falls years recall him as an earnest and gregarious man. He was well dressed, turning out in tailored suits; and to some people, Pollett appeared as a "dapper young feilow." Me he attended church regularly at Boly Trinity Anglican Church, and socially he "ran with the better crowd—the churchogen.. Within this particular group of friends was a young woman named Emms Hackett, the church organist and choir director, with whom Pollett experienced his first romance." It would seem that their interests were compatible, for both were artistic to some extest—he in writing and Emma in music. Both maintained a high and reputable profile in community affairs. Pollett, for his part, early in 1920 became a charter member of the Loyal Oranga Young Briton Association, a club which likely evolved in response to the institution of a local Knights of Columbus order a few months earlier. His activities in this new venture required some industry; noting Pollett's contribution to this organization, his friend W.T. Howell emphasized that Pollett "was a good citizen."

An unprecedented workers' strike at the A.V.D. Company, which lasted from May 9 to August 8, 1921, nevertheless affected Pollett no less than the entire company work force of about eight hundred men. Owing to a general decline in the paper industry, the company had felt forced to slash all salaries by one third or more. The strike was called after it was evident that no negotiations were expected to take place; it was a protest against the difficult terms which men were forced to accept. Although their wages were comparable to other North American paper.

makers, the cost of living in Newfoundland was about twenty per cent higher than in Canada and the United States. Pollett, like hundreds of other men, returned to his home during the summer of 1921.

In New Harbour, he found living conditions in his family home much altered. Just prior to Pollett's return, on May 5, his father had re-married, taking his housemaid, Sophia Piercey, for his wife, ⁵⁸ Whether Pollett was now willing to develop a close relationship with his new step-mother is not known; such an undertaking was made extremely difficult because Sophia was deri, although his father Jim managed to communicate by shouting requests in a booming voice. ⁵⁴ As a result, no longer did his home have the quiet and amishle atmosphere Pollett had enjoyed during his youth. Furthermore, although Sophia may not have been particularly aged, her new husband and later the earlier New Harbour community took to calling her "The Old Woman," a nickmane suggesting some mockery. Family life as Pollett had known it during his youth perhaps changed a great deal owing to this circumstance. Nosetheless Sophia is remembered as a "wonderful worker," always "one for the home"— making dothing, cooking, and in other ways assisting the familit. ⁵⁸

Despite her efforts, lowever, which were equalled by the labour of Pollett's father, economic conditions of life were such that Pollett, at home and without an income, may not have felt quite comfortable. After a period of prosperity in Newfoundland during the Great War, the country's economy estered a long period of decline. Throughout most of the 1920s, a national depression brought great deprivation to most outport people. If Pollett had begun at this time searching for a job elsewhere, he would have been assured that economic conditions in other parts of North America were no more prosperous, and that prospects for him were limited. When the strike at Grand Falls ended in early August, Pollett probably welcomed the opportunity to return there, where, for the most part, life had not been affected by economic conditions suffered by the rest of the country. Pollett was rehired by the company and his salary was maintained at the level he had received previous to the general strike. Jobs at Grand Falls were in tremendous demand; and the government and millowners were forced to publish warnings directed to thousands of hopeful

outporters seeking employment there that all jobs were taken. es

But for the returning workers, life in Grand Falls appears to have been basically unaltered by the strike. Pollett returned to work and to his living quarters at the Exploits Holei, aithough after his return he does not appear to have participated so actively in community affairs. (In fact a few of his peers do not recall that Pollett returned to the town after the strike) Other leisure activities may now have consumed his time. At the town club Pollett could read at the library, and learn to how!. In the town hall he could see movies and vandeville shows, both of which may have whetted his appetite for the outside world.

He also resumed a serious courtship of Emma Hackett, although Pollett's firm decision to leave Grand Falls in 1923 abruptly ended their close relationship. Emma rejected his proposal of marriage, ostensibly because she was not willing to leave Grand Falls. She continued to correspond with him until his death, but at the time she, like his friends, knew only that he footered a keen desire to "get ahead." From her point of view, prospects for their future together may have seemed vague and not promising. Life in Grand Falls, by contrast, was secure. Whatever migrings Pollett felt about her refusal are not known. If he had found the amenities of social life in the town much to his liking, his desire to develop his intellectual capacities and sets out life in the town much to his liking, his desire to develop his intellectual capacities and sets out life in the town much to his liking, his desire to develop his intellectual capacities and sets out challenges elsewhere mattered more. In each of his working days at Grand Falls he had dealt with paper used for newsprint, and his interest in the printed word may have been sufficient reason to ruptore aspects of the printing trade. Opportunities for doing so were not at that time arrailable in Grand Falls. These ambitions naturally led him away from Grand Falls and the intand of Newfoundiand.

lv

Late in 1923 Pollett decided to emigrate to Montreal, where he secured an appressive position in the linotype trade. Exactly why Pollett initially chose Montreal can be answered only by a series of guesses. While in Grand Falls, he may have been made aware through a trade journal, Palls and Palagar, published in Quebec, that in 1922 its publishers were planning to institute training courses for workers in the paper and publishing industries.⁷² Pollett may have inquired about the most efficient place to learn the printing trade; if he did so, Montreal rather than, say, St. John's, would plausibly have been recommended to him. St. John's in 1923 was not prospering, in Montreal, by contrast, a number of apprenticeship opportunities were available, for training in the printing trades had been neglected during the yearn of the Great War.⁷³ As business began to revive in most of North America by about 1922 (the country of Newfoundiand being among the exceptions to this trend), a shortage of skilled labour in the printing industry was seen to exist.

Pollett may have once apain followed the pattern of securing a position and a place to live prior to leaving for a new destination. He could have achieved the first by writing to one of the three local Montreal branches of the International Typographical Usion (IT.U), all of which were now concerned with the problem of a dearth of local apprentices. Pollett's minister at New Harbour, John Beauchamp, may have offered some assistance or information about living quarters in Montreal. Almost nothing is known, however, about where exactly Pollett lived or at which newspaper he worked in the city. His family and friends cannot recall his ever speaking about his exceptiones in Montreal; and in his own writings appear only brief references to his memories of "smodging my hands with printers" ink" and of a routinely chilly wait for a streetcar at the corner of Craig and St. Catherine Streets. The apparently used his savings to buy a train ticket to Montreal; once there, sometime in November of 1920, he must have immediately begun his apprenticeship in linotype craftsmanship, for the duration of which he earned about \$25 per week. To

The terms of Pollett's linotypist training according to international and local union agreements stipulated that he obtain technical and practical experience on the linotype machine for at least a three-month period. The linotype machine was a relatively new invention which allowed, through a complex hot metal process, an entire line of type to be cast in a single operation. To Until the last decade of the nineteenth century all type was assembled individually by hand; the linotype machine, which included a mechanical keyboard, enabled type to be designed, set, and rapidly cast both by manual and mechanical composition. Among the prorequisites for Pollett's training were, foremost, a good knowledge of the English language, including the ability to spell words and as understanding of grammatical construction, and secondly, a wide voxabulary and an ability to read manuscripts intelligently. Thus qualified, he proceeded to learn the keyboard of ninety symbols and then learn the various typesetting processes of the machine, and procedures for casting the lines in a hot metal substance. Another necessary skill which Pollett needed to develop in order to meet the requirements of high rates of speed and accuracy was the ability to read quickly the final compositions, issued backwards and spuide down, and thus to odit his work. Knowledge that linotype operations were not limited to newspaper work but were used readily in book printing, journals, advertising, and all branches of the graphic arts was an incentive for Pollett to master the trade. By the time of the Great War, linotype machines had become essential to all composing rooms.

On February 5, 1924, Pollett was entered as a member of the LTU, through the Montreal Typographical Union, No.176.77 Pollett apparently fulfilled conditions for acceptance into the union and thus was declared a journeyman printer. He was to find that opportunities for employment, however, were severely limited in Montreal. As an apprentice, Pollett probably would not have been intimately involved with a number of local printers' strikes, one of which had begue in Montreal as early as May of 1921, and all of which were to continue until August of 1924.77 These strikes, which involved initially nearly seventy-dive per cent of Montreal's unionized printers, were caused by an international movement spearheaded by the unions to shorten the work week from forty-eight or more hours to forty-four. At a conference held by the Quebee and Outsrie printers' unions in 1924, the chaotic condition of the printing industry was discussed: not only was it recognized that large numbers of skilled workers were compelled to migrate to the United States because of the forty-four hour week issue; it was further noted that the very high unemployment which now existed was in part attributable to the common practice of Canadian firms having their printing done abroad. 77 To enther of these problems was a solution put forward.

These conditions seriously affected Pollett's opportunities for advancement, and his acceptance into the local typographical union, its members comprising nearly thirteen hundred other skilled mea, did not necessarily assure him even a position in a local firm. Printers, however, had traditionally been as ilinerant craftmen group in North America as elsewhere. In the nineteenth century there had been a good deal of movement between print shops in Americas and Canadian cities, particularly after the National Typographical Union in the United States (later incorporated into the LT.U.) authorized recognition of Canadian union cards in 1844. This movement by 1924 was fairly regular, and was facilitated by the process of simply transferring to a different local branch of the LT.U. Pollett took advantage of this opportunity immediately upon his acceptance into the LT.U.; he became a member of local union No.6, located in New York City. The left for New York in Pohranz of 1924.

v

It would seem that Pollett's desire to advance himself in his trade was the primary reason for emigrasing to New York. He was assured of an increase in salary and reasonable daily working hours; what is more, he was offered a responsible position as the head of a composing room for a Gles Cowe, Long Island, New York, publishing firm. Exactly where Pollett was to work in Gles Cowe is not known—perhaps at a local newspaper called the Glen Cove Echo, which was a Republican party organ established some fifty years prior to Pollett's arrival. Other firms then operating in the city, the Sanford White Company, and another newspaper, the Glen Cove Record-Advance, were smaller operations. The position which Pollett obtained at any one of them, however, was for him a great achievement, for he advanced from the rank of apprentice beyond the rank of journersman to "loss-linotrypist." ¹⁰⁸

Whether Pollett chose to live in Glen Cove during the time he held this position there is not known. Glen Cove, established as a town as early as 1068, was expanding rapidly during Pollett's time there in 1924; from a town of 10,000 persons at the turn of the century (when it was called Masquito Cove) it had as recently as 1917 been incorporated as a city. In some ways the developing city resembled Grand Falls; and it was surrounded by lakes and overlooked Hampsteed Bay, which gave it as attractive appearance. But after his experience of living in the much

larger city of Montreal, Pollett may have come to fasey a more compositias setting. Brooklyn was easily accessible via the Long Island Railroad, and by November of 1924, if not before, Pollett was made aware that in Brooklyn a community of <u>émiref</u> Newfoundlanders had become established. In the Park Slope area of South Brooklyn, hundreds (very possibly, thousands) of immigrants had settled so that, in time, the neighborhood became known locally as "Little Newfoundland" or "Newfoundland Alley." ⁸⁸

The Brooklyn Daily Eagle was a newspaper which frequently printed articles of interest to this community.⁸⁶ On November 25, 1924, the Eagle featured an interview with Pollett which described how he had become the first New York recipient of a legacy bequeathed by Lord Northeliffe to all his employees (or sometimes former employees) in Grand Falls and elsewhere. 87 It may have been at this time that Pollett first became aware of this Newfoundland community's existence in Brooklyn, for in this article he suggested some feelings of estrangement from his island home and the people there. A photograph of the twenty-four-year old Pollett which accompanied the interview depicts his surprised expression; he explained, in fact, that he was amazed that the executors of Northcliffe's estate had been able to locate him, for, it was stated, he had spent the year since his employment at Grand Falls "roaming throughout the northeastern section of Canada and the United States." Furthermore, he said that there was no one in Newfoundland with whom he corresponded. Throughout his publishing career Northcliffe publicly had praised the community-minded spirit of the Eagle; 89 and this fact, given the occasion of the interview, probably urged the staff reporter to collect more details about Pollett than one would likely expect to see about a young immigrant in a major city paper. One detail which no doubt was thought to appeal to the paper's Newfoundland immigrant readership appeared at the close of the interview. Pollett spoke for the first time of a book he intended to write "dealing with the home life of his people in Newfoundland." "There has never been one written that I know of." he said. "and I think I'm qualified to write it."90

Pollett's most recent experiences perhaps gave him every reason for confidence and ambition. The challenge he had set for himself in Grand Falls only one year previously, that of learning a trade appropriate to his literary interest, mastering it, and thus prospering financially, had been fulfilled. It would appear therefore that the image Pollett hoped to convey of himself in his interview was that of a capable, independent young man, who had consciously severed his ties to home and had done well enough in his profession so as not to be in any immediate need of the forthcoming money from the Northcliffe estate. "I'll put it away with the rest of my savings," Pollett said. "Since I really haven't earned this money and as it comes under the head of 'investments,' I guess I may as well save it." His progress in his career and keen ambition for further advancement may have helped him to land a job as printer at the Brooklyn Daily Eagle, the major paper of the city, subsequent to his interview. The paper's owners and publishers, the Hester family, resided in Glen Cove, but whether Pollett made their acquaintance, perhaps as a result of his feature interview, is not known. Pollett was probably aware, however, of the well-known Horatio Alger-style career story of William Van Arden Hester, father of the family, who had begun his own career as a printer's apprentice. 22 Hester's accomplishments may have inspired Pollett to obtain a position at the Eagle, a newspaper held in high esteem nationally. By 1925 Pollett had chosen to reside in Brooklyn. In effect, his seven-years' experience of travel from New Harbour, first across the bay, then across the country, midway across Canada, and finally to New York, was a sound beginning for the very different future he intended for himself when, as he later wrote, "I scrubbed the squidsquirt from my neck and quit my outport home to seek a fortune."38

'NEW YORK-NEWFOUNDLANDER.' 1925-45

Ros Pollett's life between 1925 to 1945 was characterized successively by achievement, mifortuse, and the rediscovery of his homehand. Soon after his narrival in Brooklyn in 1925, he decided to settle in the city, and early in 1927 he married and began to enjoy the comforts of relatively afflicent family life. Crises suffered during the Depression, however, thwarted his ambitions. Financial stability eluded his grasp, and he floundered in attempts to establish himself satisfactorily in New York. Possibly because of this grim experience, Pollett sought to renew intimate ties with his homeland, particularly between 1934 and 1939. He succeeded in reestablishing many close ties to Newfoundland, is addition, by the early 1946s, he regained both financial security and prospects for achieving his career ambitions in New York. Thereafter, Pollett continued to seek out various means of bridging the two very different worlds, one earthy and traditional, the other affluest and novel, both of which he had experienced by 1945.

-

Ros Pollett had established himself in "Little Newfoundland," a section of Park Slope, Brooklyn, by the summer of 1925. In a brownstone bouse on Ninth Street, he rested pleasant rooms from the Bartlett family of Brigas, and here he resided area Fifth Avenue, the major thoroughfare of Brooklyn. At the crossroud of Fifth Avenue and Ninth Street, groups of Newfoundlanders congregated frequently; and the site was one which Pollett later referred to as "the night-time village square." There he joined other recent arrivals to the neighborhood, some of whom Pollett later recalled had come directly from the island and still had "salt-spray in their neck wrinkles." but soon "stimmed off their woof peaker caps and donned quiffs and straws." Discarding the accounterments of their homeland was but one indication of their shared desire to embrace the new American culture. Earlier settlers in the area provided an important social new work for recent arrivals, providing formal references for new jobs, and acting 20 witnesses at immigration proceedings, christenings, and marriages; as well, they told of local stores that carried Newfoundhand food products and aewspapers, of churches where local Newfoundhanders were members, and of halls where Newfoundhand-style square dances were held. In the company of one of these men, Abe Snow, Pollett came to know Prospect Park and Prospect Hall, but he eventually breastened his territory to Central Park, and the diverse cultural amenities of vaudeville, theatre, and movies in Manhattan. Pollett once characterized Manhattan as "the jigging ground for the whole shore;" it seemed to Snow that Pollett observed everything there was to see, and he remembers Pollett's delight and curiosity toward the novelites as much as toward the unpretentious commosphaceness in the city. In effect, Pollett was now able to enjoy the best of two worlds: the opportunities and pleasures of one of the largest cities in the world, and community life among local Newfoundhanders. He was able to see the sprawling and various cities of Brooklyn and nearby Manhattan as a series of neighborhoods, and recognize his connection with one, "Little Newfoundhand" in South Brooklyn.

Life in this neighborhood suited Pollett's ambitious character. He eventually befriended dozens of young men from Newfoundland who, like himself, were employed in highly stilled and well-paid trades, and be later recalled aligning himself with this expectant group of "young Newfers then making our mark on the face of New York." Indeed he sought out opportunities to increase his income and broaden his range of experience, and now established habits that were to characterize most of his working life. He routinely chose to work from mid-afternoon sattliminight at his regular employment at the <u>Brooktyn Daily Eagle</u>, for this shift was more highly paid than daytime work. He also "moonlighted," frequently securing work at the <u>New York Times</u> and elsewhere when skilled linotypists were needed temporarily. In addition, he acquired the habit of reading a minimum of five New York newspapers per day, and he may have studied the papers thoroughly, for a wide knowledge of design techniques and composition could prove useful to his career. Printers could, for instance, receive premiums for correcting errors in style and language they detected in their working copy. A broad knowledge of the techniques of journalistic writing would later manifest itself in his publications. By 1926 Pollett was earning approximation of the proper stransparency of the proper stransparency of the proper stransparency of the stransparency of the verticing would later manifest itself in his publications. By 1926 Pollett was earning approximations of the proper stransparency of th

mately \$50 per week, a substantial salary for an unmarried young man, and his devotion to his work was such that, according to his friend Abe Snow, Pollett worked harder than any Newfound-lander he knew.¹⁰

In June 1926 Caroline Mariah Gilbert a recent arrival from Newfoundland was introduced to Pollett through his friend Jack Brooking, an iron worker from Bonavista Bay; and the two men pursued a courtship of the young girl, 11 Caroline, born and educated in the small and isolated outport of Havstack, Long Island, Placentia Bay, was at the time a slim, red-haired, nineteen-yearold girl, and had come to New York only two weeks before to assume a pre-arranged private nursing position. Her background differed from Pollett's in that she was the youngest of nine children in a Methodist family, and her father, Stephen Gilbert, had been a local merchant who also owned a lobster canning factory in Haystack. Gilbert's prosperity enabled him to finance Caroline's two years of training in St. John's, which qualified her as a practical nurse; and in May, 1926, she assumed her first position, this as a private nurse for a wealthy German woman in a luxurious Greenwich Village home, "Ron was a gentleman," a polite, handsome, and fashionably dressed man, Caroline recalls when describing her first impression of Pollett. She soon showed a preference for Pollett above her other suitor, and encouraged the quiet enthusiasm he showed in speaking of their prospects together. Giving up her career was part of the plan, for although Pollett appreciated her effort to advance herself beyond the traditional Newfoundland woman's domestic roles, working wives were considered disgraceful among the Brooklyn-Newfoundland community, in that they seemed to suggest inadequacies in their husband's ability to provide.12 Pollett, to his credit, refused to allow such an inference to be entertained by his peers. A second of his demands concerned Caroline's religious beliefs, and by late February, 1927. Pollett was pressing for a June wedding and for her conversion into his own church.

The wedding took place on March 4, 1927, at St. John's Episcopal Church on Waverly Place in Greenwich Village, ¹³ Two Newfoundisanders living in Brooklyn, Bill Whelan, an iron worker from Conception Harbour, and Emma Halfyard, a housemaid from Haystack, presided as witnesses. The haste in planning the wedding may account for Policit's failure to invite his family ily and friends from New Harbour to the ceremony or to the reception at the stylish Hotel St. George is downtown Brooklya. But his wife recalls that Pollett never expressed a desire for the attendance of members of his outport family, and the presence of only his newly-made friends may suggest his desire to begin life afresh in his new role of husband and provider. He was optimistic about his future, and apparently wanted to leave the memory of outport toil, "hasling and mauling the cod trap," if far behind him.

The couple soon moved to a four-room apartment on 82nd Street in the Bay Ridge section of Brooklyn. Moving from the Park Slope neighborhood was now becoming, among Pollett's Newfoundland-born friends, an indication of financial achievement. 15 That part of the metropolis no longer fitted the stereotype of "a city of homes and churches" which had long since characterized Brooklyn; increasingly by the late 1920s, the spacious brownstones had been cut into inadequate rooms lacking kitchen facilities and privacy, and were rented to transient immigrant families who often were strangers to their neighbours 16 In Bay Ridge, by contrast, Pollett could provide his wife with spacious and pleasant living quarters, and begin acquiring more of the comforts of middle-class life. Elegant furniture was purchased, and in their leisure time husband and wife spent hours reading together, sitting in two plush Queen Anne-style armchairs, referring to one another as "Queen Anne" and "King George." Pollett often read aloud on these occasions from Wuthering Heights, the plays of Shakespeare, and other classics, and did so in a well modulated voice, crisply enunciating each word. He held the belief that Newfoundlanders who moved to New York ought to learn to refine their "baymen's accents;" he had taken pains to attempt this many years before when he aspired to become a schoolteacher, and, he now confided to his wife. he felt annoved when his Newfoundland-born acquaintances persisted in retaining their distinctive dialects. In addition, he wished to develop a familiarity with ordinary American fare unknown to the outport; and often asked Caroline to prepare meals he had seen featured on restaurant menus or had watched other diners consume. It was not long before they had their first experiences with such novelties as spaghetti and teabags. Developing a liking for many new tastes and pastimes was to commingle with the novelty of their first year of married life and establishing a household

of their own.

Pollett's first child, Ronald James, was born on December 21, 1928. The addition to the family caused Pollett to become a more patient and good-natured man in the home. Caroline had grown stout, and he now referred to his wife as "Mommy," 18 He shared his pride of fathering a son with other Newfoundlanders, among them Abe Snow, Jack Pike, and Bill French, whom he met with their children when wheeling his son to nearby Fort Hamilton Park in warmer weather. This was an idyllic spot, well-tended, and overlooking New York Harbor. Such excursions may have reminded Pollett of the rare occasions in his earliest boyhood when his mother walked him about New Harbour, visiting with relatives and neighbours. Other reminders of home perhaps resulted from a new, closer relationship that he now fostered with local Newfoundlanders who like himself had prospered, moved to Bay Ridge, and had begun to raise families of their own. Cardparties took place in his apartment each Saturday night; and recitations of poems such as "The Hesperus" and "Lucy Gray," remembered from the Royal Readers, were featured on these occasions.10 Such parties, where the wives sat embroidering and crocheting items for their families. and men played spirited card games and looked forward to "big feeds" at midnight, could well have taken place in any house in Newfoundland. Pollett was probably well pleased with the differences in his life since his days as a teacher. He had prospered; and his pleasant home and own smail family were evidence that he had adapted well to his new circumstances.

The value placed on progress and personal achievement in America was one that Pollett embraced during these first years in the city; and by 1920, his income well exceeded the yearly average of \$2500 them thought to be sufficient for a family of four. He was happy in his worth-his wife emphasires that "Ron empoyed mind work"—and the heetic pace of his daily routine excited him. The Even travelling to work appears to been a source of wonder. Pollett later observed that only on the New York subway could one have the opportunity to stand exect to "a beautiful girl on one side and the erragey hip boses of a repellent gardic-furned fish pediar on the other." Theoperity enabled him to partake of a variety of local attractions, which ranged from an evening spent at a boistorous Newfoundiand-style square dance at Prospect Hall, to the exit free evening

attending a local vandowille theatre where Max West, Pollett's favourite performer, was featured.²² The couple's shared enthusians for city life, and their feeling that "there was everything to see, there was everything to do," bespoke their intentions to stay in the city.²⁴ In Pollett's later writings, when he recalled these years, he would revise his somewhat unarticulated ambitions and plans fostered in the outport, speaking of his many travels as a "circuitous" route,²³ and indeed implying that he had always intended to reside eventually in Brooklyn.

.

Owing to an emphasis on his work and career, perhaps, Pollett had managed to accumulate an impressive amount of savings. Exactly how he invested his money is not known, but when the banks crashed in October, 1929, he lost a considerable sum. As his son recalls from conversations he later overheard, "Maybe my father's bank paid back ten cents on every dollare even till the forties they were paying it back. That source my father a little." If This los prompted Pollett to move his famility to a less expensive apartment on 89th Street and to cut most luxuries out of his budget. Ensuing economic conditions during the Depression demanded even more self-discipline: by the early 1930s, printing firms cut back production, and Pollett could no longer rely on his weekly surnings. He became one of the millions of workers forced to accept "shared employment." an expedient tactic characteristic of the Depression era which at once levelled differences in terms of experience and qualifications among men in particular occupational groups, and thus gave to each at least a fraction of work, rather than none at all. Owing to this practice, at times Pollett managed to secure work only from one to three days per week. This situation may have fostered in him his idea of the worker in New York being "just another capilin in a puntiond."

Ideals of progress and prosperity which Pollett had long christed were frustrated as he increasingly suffered insecurity and poor prospects. He still lived in the Bay Ridge area close to other Newfoundhaders, although now he perhaps contemplated some differences between his own and other immigrant groups. Whereas other local groups usually had established mutual aid and bearvolest societies, Newfoundhaders never had organized formuly as a group for such purpose.

"No; the outporter in New York," Pollett later wrote, "had to rew his own dort, both financially and socially, as far as organized help was concerned." He preserved: He often walked about Manhattan trying to locate temporary printing work, although at a time when many printing companies were forced to go out of business, it could be expected that he usually failed in these attempts. He hearned the layout of the city well, he saw the billboards advertising the latest Broadway shows and newest products, and he saw the expensive restaurants. All of these sights helped to foster his rather belated awareness that only prosperity enabled one to enjoy the novelty of New York life, for indeed a striking disparity had always existed in this, the largest city in America—a key cultural centre for the effile, but its attractions remaining virtually unknown to many other residents who knew only noverty.

Early in the 1920s, poor financial circumstances forced Pollett to move his family again, now to an inexpensive three-room apartment on Fort Hamilton Parkway, in a building where five American families shared with his own a drastic decline in fortune. It One neighbour, Paul Plumber, who had until recently worked on a glamorous yacht oward by millionaire J.P. Morgan, now was not of work, and barely made ends meet by depending on his wife's kills as a professional hairdesser. The prestige of providing for his own wife so that she need not work outside the home still mattered to Pollett, but the responsibility of tending to their young son also precluded the possibility of Caroline's searching for work. Conditions were relieved somewhat by his family's joining in the practice of sharing dinners in nightly rotation among the group of families in the building. Each of the six took turns providing meals for the rest and thus saved sums on electricity and fael costs. This scrimping was one way in which Pollett was forced to reduce his expenses. If he looked forward to a certain variety in the conversation and fare at the evening meals, then the telium of occupying hisself through long days when no employment could be secured no doubt did not relieve his feelings of worry and frustration.

Sometime in 1933, Pollett contracted a painful gall bladder disorder and thus found it difficult to walk about Manhattan in search of work. An operation was deemed necessary, and the expense of hospital care alarmed him. He had no medical insurance, could count on earning no income during his convalescence, and as a result put off the operation. It may be that at this time he acquired a passion for gambling, and racetrack betting remained a habit of his for about the next fifteen years.33 Like most men who bet on horse-races, and indeed there were then hundreds of thousands of such men residing in Brooklyn,34 the mentality of getting rich quickly was one that Pollett fostered. The excitement and tension of choosing a winner from the listings in the paper, while he would chain-smoke Camel cigarettes- another nearly lifelong habit- and wait anxiously for the results of the races, occupied him daily, "He always lost," his wife recalls, 35 As a result, perhaps for the first time since his ambitious move to New York. Pollett began to entertain doubts about his prospects. But he was not a man to give up easily, and he thought things through carefully. As his friend Abe Snow recalls of Pollett, "he would never let a subject drop, he was always probing, until he got to the bottom of it."28 Despite the levelling experience of sickness and poor finances, he came to the conclusion that he could survive well enough in New York on his own and that financial conditions could be relieved by sending his family to New Harbour. Life in New Harbour would be more comfortable for his wife and child than life in New York, he thought; in addition, he now conceived, it was easier to "make do" with less in an outport.37

Pollett accompanied his family to his father's home in June, 1933, the worst year of the North American Depression. His wife and son were to live there for fourteen months. Pollett's own stay was brief, as he intended to undergo his gall bladder operation immediately upon his return to New York. He must have been appalled by the impoverished conditions he saw briefly at New Harbour; in 1933, Commissioner William Warren Mackenzie Amulree described Newfoundlanders' lives as "desperate." Dellett concentrated instead on the positive benefits of his plan, one of which was the opportunity to have his five-year-old son learn some of the joys of Pollett's own ostiport boyhood. His son could spend his days happily fishing for tomcods or romping through the woods, rather than stay at home, see his sick father, and witness Pollett's inability to obtain work. He may now have begun to see New Harbour as a pastoral idyll in comparison to the big city. Raymond Hillier recalls seeing Pollett on this visit: "I still see him stand-

ing on New Harbour Bridge leaning over the rail gazing out over the harbour while gossiping with old friends." Perhaps he was tempted to stay and enjoy the simpler pleasures he imagined.

Sixteen years had passed since Pollett abandoned the traditional mode of outport existence, yet upon his return to New York he must have been struck by a keen awareness that he actually had little to show for his many years of ambitious labour. He put his furniture in storage, moved to a small, rat-infested room on 86th Street, underwent his operation, and continued a fruitless search for permanent work. Frugality was a fact of life. He managed to send adequate sums to his wife, and during their separation wrote her long lettern as often as three times a week. "I hope you are having a good time," he frequently wrote, much to his wife's chagrin, for he seemed to think of his family's stay as a holiday retreat. In magined pleasures loomed large, although he knew from her replies that widespread poverty and poor prospects in Newfoundland in the early 1000s necessitated both the relinquishing of that country's powers as a self-governing Dominion, and the providing of a government dole of six cents per day to its people. If Thousands of immigrants now forced to return to Newfoundland, as Pollett must have guessed, were viewed cheigh as burdens on their outport families. For Pollett, a man who had always taken much pride in his accomplishments and merit, there now seemed no choice but simply to stay and bide his prospects in America.

III

In the summer of 1934, Pollett obtained a permanent position at AD Press in the Soho district of Manhattan, and was in a sense able to begin life afresh after years of suffering insecurity and poor prospects. AD Press, printers mainly of legal work, was then one of the most reputable firms in the industry, and according to some tradesmen, a job of "law printer" was the "pinnacle" of status linotypists could hope to achieve. It was here that Pollett was to work for about the aext fourteen years, and where he eventually became known to his fellow workers as a master of his craft. Law printers were in fact required to have mastered diverse skills and techniques in order to fulfall clients' stringent demands for extremely precise, fast, and clean productions. In addition, owing to their expertise, they could command a higher salary than most other printers.

Recognition of his ability and the promise of financial stability no doubt helped to restore

Pollett's confidence in himself and his plans.

He now set about reestablishing a harmonious home life. By September, 1934, he had rented and refurbished a pleasant four-room apartment on Gelpson Avenue, a tree-lined street near Fifth Avenue hardering the Bay Ridge/Fort Hamilton sections of Brooklyn 44 However, he soon received his first awakening to difficulties to be faced; both wife and son returned from New Harbour extremely thin and sickly, and Caroline was diagnosed by a Brooklyn doctor as having contracted tuberculosis. She recalls that Pollett, too, had lost a considerable amount of weight; "there were the three of us, thin as rakes- you could count the ribs,"45 Although he was soon heartened to discover that Caroline's disease had been misdiagnosed, he hastened to relieve her distressed condition, caused by exhaustion and noor nutrition, by hiring a maid to help with the housework, and he himself occasionally prepared meals for the family. Pollett must have been gratified when, to add to his pleasure over his financial recovery and his wife's physical improvement, his son was promoted to the first grade of grammar school after only one day at kindergarten classes. Caroline had taught the boy to read and write during their stay at New Harbour. Pollett now began to take a keen interest in his son's education, and bought him, on weekly installment plans, a number of texts, including an encyclopedia called The Wonderland of Knowledge and a two-volume dictionary. He appears to have wanted to encourage as fully as his means allowed his son's love for reading. Furthermore, he now found time to share with his son pastimes similar to those he experienced as a boy-deep-sea fishing at Breezy Point, Long Island. ice-skating, swimming, and sleigh-riding. At Prospect Park he often rented a boat for some hours to row about the scenic man-made lakes with young Ronald. "Here, I could get the taste and smell of home." Pollett later wrote of this immense park. 65 In effect, these activities with his son instilled in the boy a love for the out-of-doors, and enabled Pollett to relive the pleasures of his own childhood.

His insecure and lonely existence during the Depression years obviously had some effect in

altering the desire he once fostered of jettisoning parts of his heritage. More than at any time during his adulthood, Pollett apparently now wholed to cultivate ties to his homeland. His herif trip to New Harbor in 1933 had enabled him to renew ties with his family and friends, and he since had surtured these connections through regular correspondence. He thas was kept informed about the conditions of life in New Harbour, and his own regained prosperity provided him with the means to become, to some degree, a benefactor. He now learned how crucial the money he had sent in 1933-4 had been to the Pollett family in New Harbour. After 1934 he continued the practice of sending mouthly sums to his father, a routine which was applicated by the community. He also began to collect used clothing to send to needy members of his family and others in the outport. Bertha Pollett, his sister-in-law, recalls how necessary these items proved to her six sons during the mid-1930s. Pollett's sense of companion prompted these gestures, and the soundness of his own finances enabled him to establish a new relationship to the island.

By 1000 Pollett found another outlet through which to assist Newtoundlanders. He learned, apparently from his correspondence, of a new business venture—a printing operation, part of which would be devoted to a town newspaper—started in Grand Falls by two young brothers, Walter and Michael Blackmore. Pollett managed to obtain a copy of the paper, the Grand Falls Advertiser, and was so pleased with the fledgling attempt that be immediately began a correspondence with one of the brothers, Walter Blackmore. Blackmore, being known only slightly to Polett when as a youth be was employed as an errand boy for the A.N.D. Company, results that Pollett offered generous advice on all aspects of the new business. Pollett sent trade magazines dealing with the graphic arts in order to expand the Blackmore's knowledge of design. He searched New York for a secondhand linotype machine, for he assured the Blackmore is that they needed one to build a sophisticated operation, and to the brothers' amazement he also intended to pay for the machine. In addition, he sent critiques of articles which had been published in the Advertiser, that, as Walter Blackmore says, matched the "wooderfully apt phrasing" of which all of Pollett's letters, even the simplest greeting card, were composed. Blackmore only, matched the "wooderfully apt phrasing" of which all of Pollett's letters, even the simplest greeting card, were composed. Blackmore only, matched the "wooderfully apt phrasing" of which all of Pollett's letters, even the simplest greeting card, were composed. Blackmore only, matched the "wooderfully apt phrasing" of which all of Pollett's letters, even the simplest greeting card, were composed. Blackmore only, matched the "wooderfully apt phrasing" of which all of Pollett's letters, even the simplest greeting card, were composed. Blackmore only and the process of the process

that Pollett possessed a "Buge rocabulary," not flawed by affectations, and that he "was a great hand at painting a picture with words," ⁶⁸ Pollett was to remain keenly interested in the <u>Advertiser</u> until his death, and became, as Blackmore acknowledges, one of perhaps half a dozen benefactors who helped make that paper a success. ⁶⁸ Indeed, he says, Pollett "was a man of no halfway measures. When he set his mind to doing a thing, he'd do it and do it the best way. ⁶⁸ Pollett apparently saw in the Blackmores' ambitious undertaking a ready opportunity to make a contribution to his homeland.

Pollett's involvement with the Grand Falls venture enabled him to renew his literary interest in Newfoundland, and perhaps reminded him of his ambition, spoken of in 1924, to produce a book "dealing with the home life of his people in Newfoundland." Whether he saw his contributions to the <u>Advertiser</u>, particularly his editorial assistance, as a means to practise his writing skills in preparation for his own creative output is not known. Developing an intimate consection with Newfoundland, however, apparently became an increasing concern of Pollett's in the next few years. Blackmore helped him to maintain contact with current event by stending him a copy of the <u>Advertisine</u> each week." Pollett's regular correspondence to Newfoundlanders in the mid-1900s, according to his son, was voluminous; and he recalls, "there were always letters around the house from Newfoundland," so many that young Ronald began an extensive Newfoundland stamp collection." In 1937-8 Pollett also began the practice of listening to Newfoundland programs on the short-wave hand of his radio. The "Doyle News Bulletin" and JR. Smallwood's "The Barrelman" were programs, among others, which helped him to keep abreast of events in Newfoundland and to remind him of colorful bits of the island's history.

Activity relating to Newfoundinad apparently took on special importance in Pollett's everyday life. His son recalls that the only time his father became angry with him was when he once "fixed" Pollett's ratio by tightening the screws and so interfered with the tuning mechanism.⁴⁰ This incident stands out in his son's memory, for Pollett was ordinarily a good-natured, eazygoing parent. Pollett's interest in Newfoundiand soon manifested itself in a desire to spend a full summer in New Harbour. Indeed as early as the spring of 1935, his wife recalls, "Every year." come April or May, he'd say, 'I guess we'll be going to New Harbour come summer, "**⁶⁹ He ignored her own desire to vacation at a tropical resort, giving as his reason, "You can't shake hands with a tree," and and was instead insistent on travelling to New Harbour. Caroline's unsettling experience there in 1933-4 perhaps was overlooked, for Pollett now could provide the inducement of travelling to Newfoundland first-class, and had the financial means to ensure a leisurely holiday. His estabasism convinced her, and the family left for their journey in late June, 1939.

This extended tip was for Follett his first chance to appreciate fully the pleasures he had known as a boy in the outport. Once his family was settled in the house of his boyhood, he eagerly set out to rediscover familiar ponds, and with his son often joined his father Jim, fishing for cod in his pust. "A brand-new codfish always looks good to an ex-outporter," Follett later wrote, adding that catching one was "like meeting a long-lost friend." In preparation for his vaxaios, Pollett had purchased at Mary's department store in New York an elegant spill bamboo fishing rod with brass fittings and multicoloured silk line, along with special cod hooks. "I The rod so impressed his father that." I'm asid "a fish around here would be proud to be hooked on an outfil like that." Pollett also busied himself socializing with the people of New Harbour-house visiting, picnicking, and chatting with the men at the wharves. He frequently appeared in fushionable dress, coming into homes bearing gifts for relatives," and with ample lessure time to enjoy angling and socializing, Pollett was well aware of his improved status among fellow villagers. The holiday afforded the opportunity, he later wrote, "to play peacock where you grew up crow."

When his wife grew retelless and bored during a week of rain, reminding him of the pleasures they were missing in the city, he planned a family camping trip when the weather cleared to New Harbour Barrens, a site he had favoured during his youth. The weather nevertheless turned unseasonably cold, and on the third night camping Caroline recalls, "Even the horse tried to get into the teat. Oh, it was freezing! But Ron wouldn't admit it." **60 As a result, perhaps, he and Caroline now took excursions to St. John's, Buchans, and Grand Falls, and in these ways Follett tellfilled his promise made to Caroline hat the trip would not be similar to the financially difficult

and wearisome year she had spent at New Harbour in 1933-4. Walter and Michael Blackmore recall a brief visit Pollett paid to their printing offices while he was in Grand Falls. Rather than introduce himself, he sonehalantly sat down at the wacast linotype machine and began "banging away at the keys," proceeding to "do things with the machine no one else could make it do, things we'd never seen." No doubt the Blackmores' sentiments to the effect that Pollett was a "genist" and "could have 'made it' in Newfoundland" were communicated to the visitor and later reinforced through the correspondence he continued to undertake with Walter Blackmore.

His holiday experiences proved graifying, and it would seem the only sethack Pollett acknowledged was simply that his trip did not hast long enough. The onset of World War Two on September 1,1939, forced him to take a burried leave with his family despite his plan to stay until mid-September. They had some difficulty reentering the United States, as they had so documents to prove they were residents of Brooklyn. A paid utility bill which his wife carried in her pure was in effect the only such evidence they had. "That's what saved us," Caroline recalls. "It is unforeseen problem, as well as the growing restrictive measures on immigration to the United States and the uncertainty of the war years, may have alerted Pollett to the necessity of obtaining American citizenship. He had not followed a trend during the 1920s among Newfoundhand immigrants of giving up their citizenship. "His reasons for not doing so are not known, although his most recent experiences on the bishad may have made this choice more difficult. Indeed in the next few years, his memories of that trip were to influence much of his experiences in the city.

iv

The early years of the Second World War would appear to have brought greater prosperity and stability in Pollett's life than he eajoyed at any other time. Printing companies Sourished and salaries increased, circumstances facilitating Pollett's move in September, 1950, to a four-room apartment in a prestigious six-story brick apartment abuse on 92nd Street in Fort Hamilton, Brooklya, 70 The building was replete with such adoraments as marble stairways and canopied entrance doors serviced by a doorman. His new apartment cost him the weighty sum of \$52 pc.

month, and it was here that Pollett lived for the rest of his life. His daughter, Ethel Mae, was born during this period, on August 4.1941. News that his wife was pregnant with their second child after a twelve-year interval, probably accounts for his decision to seek the security of American citizenship. Pollett and his wife had been fond admirers of President Franklin D. Roosevelt, and realized that as citizens they could benefit from the federal programs which Roosevelt had instituted.⁷¹ For Pollett, the most important of these measures was the Social Security Act passed in August, 1935, which stipulated that all Americans would be provided for in their own lifetimes. Pollett became an American citizen on May 6, 1941.⁷²

According to local custom, two Newfoundlanders were his witnesses at the citizenship hearing- Alice French and Bill Whelan, both of whom lived closeby in Bay Ridge. When subsequently Pollett declared himself to be a Republican, the national party associated generally with larger business interests and men of wealth be alienated himself somewhat from other local Newfoundlanders, who for the most part were Democrats. In the Bay Ridge/Fort Hamilton locality, his friend Abe Snow has said. "The Democrats lived on Fifth Avenue: the Republicans lived on Shore Road"73 - this being the accented, broad geographical distinction between the politics and financial means of men from Bay Ridge on the one hand, small businessmen and tradesmen, and men from Fort Hamilton on the other hand, citizens of substantial wealth. Pollett's career and family now, clearly, figured larger in his plans than did local, ethnic affiliations, and indeed these concerns had long surpassed his initial bond with the local community. Contact often was infrequent. Whereas many Newfoundlanders, including Pollett's wife, belonged to local churches and socialized frequently in their church organizations. Pollett had stopped attending church soon after his marriage, though his motives for doing so are not clear.74 Abe Snow has said that this change in Pollett was not unusual, for male Newfoundlanders "lost their religion in New York Religion didn't play much of a part in their lives." In fact, except for his union affiliations which were required for his work. Pollett was never to belong to any any organized group- a sharp contrast to others' involvement in local groups such as the Knights of Columbus and the Newfoundland War Veterans' Association.

Pollett may have wished to cultivate an image of himself as an independent and diligent worker and provider, one aloof from community concerns, and this attitude persisted despite a variety of Newfoundland-related activity that now occurred. During the war years the general Newfoundland community in New York began to foster a somewhat higher public profile than in the past as a result of a resurgence of the Newfoundland War Veterans' Association, and the instituting of various New York-Newfoundland clubs and a branch of the Newfoundland Tourist Bureau in Manhattan. 78 Ventures within these organizations included the publishing of a community newspaper. The Newfoundland Weekly, edited by Major R.H. Tait, sponsoring dances and card parties for fundraising, and presenting lectures about Newfoundland's history and development at which slideshows and movies of the island's scenic, tourist attractions also were featured. Given his own keen interest in Newfoundland, Pollett's infrequent participation in these ventures seems inconsistent; indeed he responded critically to the new activities he saw, and began favouring the traditional world he remembered. He later wrote, "You only have to peek in at some of the square dances...to see the change- the floor either deserted or flanning with the second generation in the tame 'harn' interpretation of the one-time knock-down-and-drag-out dust raisers of their parents. And at the card parties the old verve that smashed the table whisting the Five has largely been superseded by the subdued bustle of women playing Bridge."77 Attending Lee Wulf's films about sport fishing in Newfoundland aroused in Pollett only this fleeting fancy: "as a trouter I might have become internationally famous like Lee Wulf...if I had stayed in Newfoundland."78

Notatigic sestiments he fostered would increasingly affect his enjoyment of urban pleasures.

Current films such as <u>Gone</u> with <u>the Wind</u> and <u>The Grapes of Wrath</u> were taken in three or four

times; the avorels on which the films were based he read as many times, rather than induleg in the

impressive variety of entertainment available in New York. The time when he felt that "there

was everywhere to go; there was everything to see" clearly had passed. He joined in eagerly auticipated family extraions to Coney Island, although at forty years of age amusement parks and

eccentric side-shows had little appeal for him. He had a marked distasts for the place. In his

view, Coney Island was nothing more than a beach crowded with obese women in bathing suits

whose broods of children threw garbage into the Athastic waters. A photograph of Pollett from the period in which he is fully dressed in tailored suit, hat, and matching accessories while sitting the crowded beach, sternly staring into the camera, appears to suggest this disapproval of the highly commercialized feniand. The only break in the tedium was provided by reasons with visitors from Newfoundiand, some of whom were sent to Brooklyn on military manaevers, and others who were able to plan vacation trips as a result of wartine prosperity. As his wife recalls, "Whenever he heard someone from Newfoundiand was coming to New York, held write and say. Don't go to a hotel, they're expensive and lonely. Stay here, "E Eventually their apartment became known among their neighbours as "Pollett's Hotel." The visitors joined Pollett in animated recollections of their homeland, and he reliabed the pastine, relegating to his wife the task of hosting his guests on daily sojourns into Manhattan. His nice Blaache Gilbert remembers that he did show some eathusiasm for the 1939-40 New York World's Fair, to which he eccorted here, a although, as he recalled, what impressed him most about the event was the Maine's fishing chabit. "Here was a dream!" he write, "I could stand on a rustic bridge and look into a real stream, and what could be in that stream but trout!"

This experience apparantly encouraged Pollett to seek out what to his mind were authentic Newfoundland pastimes. In the summer of 1941, for instance, he made his first attempt to fish for trout in nearby Westchester County. On the day of Pollett's first outing, however, no impressive fishing gear, special bait, or personal expertise brought him, or any one of perhaps one hundred other New Yorkers who lined the bank of a favored stream, a successful catch, ⁵⁶⁶ As his son comments about the typical horder of trouters, "The poor trout don't stand a chance!" Pollett and his wife also began the practice of sharing the restal of a summer home at Wareham, Massachusetts, on Cape Cod, along with Caroline's sister Lil and her hubband Marcel Fournier, who lived in Boston, "During these two week-long visits, Pollett's activities were similar to those he could have enjoyed at New Harbour; he spent his days fishing, clamming, and herrypicking. Although these vacations were restful and scenic, his wife recalls that Pollett would never agree with her that Wareham was in any sense "better" than New Harbou; Just as Worchester with her that Wareham was in any sense "better" than New Harbou; Just as Worchester

County lacked the natural panorams he was used to—"the babbling brook with over-hanging crags and money banks... and all the other happy surroundings that are as much a part of real trouting as the root and reel itself!" — so Wareham lacked the friendity camaradree hoved at New Harbour. Pollett's inability to travel to Newfoundland during the war years owing to a lack of accessible, speedy transportation was cause perhaps for a growing impariance with a well-heeled lifestyle that yet could not afforth im the simplest of pleasures to be had on his native island.

A small business vesture Pollett undertook in 1940-1 with his brother Jim, who lived in Buchana, a mining town in central Newfoundland, perhaps reflects his desire to create still further ties to home. **Pollett exported stylinh clothing from New York to Buchans, these items ranging from suede coats for women, fashionable rainwear for men, and chainty outfits for children—all of which were not rendily available for purchase to the workers in Buchans. Pollett's motive in establishing this business is not known; for while immediate profit was not his aim, his effort were beyond the realm of mere charitable gestures. Jim sold the clothing for the price which Pollett paid for it in New York, and Pollett simply used the money from purchases to buy more of the items in demand, shopping for them in his spare hours, and himself shouldering the costs of mailing. Eventually, however, he was persuaded by his wife to dispose of the business when it became too time consuming. Jim too was made uneasily aware of the mining company's

disapproval of outside ventures undertaken in the company town. Pollett abandoned the business by 1942. If his own renewed prosperity led him to search for broader horizons, he nevertheless underestimated the difficulties of creating for himself new opportunities in Newfoundland.

•

The ambition which had most attracted Pollett to New York initially—advancement in his profession—was one that he realized fully by the mid-1940s. He was promoted to the position of foreman at AD Press, and moreover had gained a reputation among his fellow workers as a master at his craft. Pollett, as his son once observed, could disassemble and repair with little effort the complex listoyce machine whenever his work appeared slightly less meticulous than he wisted. We His family perhaps had no reason to suspect he had developed this capability, for at home, in contrast, Pollett would never help with home repair even so far as driving a nail. "Ron would use a hammer and take his thumb foll" his wife recalls." It was in the workplace that Pollett gave most of his energies as he approached middle age. As well, he joined the company's weekly bowling group, and he visited his co-workers at their seaside summer homes on Long Island. Despite his achievements, it was to simpler pleasures with the familiar group of workers to which he now traved.

"A Brief for Appelhats" is the only known example of Pollett's writing during his initial twenty years in New York, and this was a comic prose sketch he wrote for and about his coworkers. It described a series of absurd charges in an imagined legal case and named each of the company's employees as the litigants and defendants. In his spare hours, Pollett printed "A Brief for Appelhats" on bise legal paper, folding it like an authentic brief, and gave a copy to each of the men. No copy is known to exist, although its attention to detail and its humour apparently were qualities that his fellow workers long remarked on. For one of them, a Mr. McNulty, the most memorable quality about "A Brief for Appelhatst" was that it was composed by so quiet a man. Pollett was well liked and respected but never demonstrative of the affection he felt toward his fellow workers. In the workplace, amidst the noisy machines and the busy composing room atmosphere where independent and rapid productions of precise work were required, Pollett pursued his tasks and interests apart from the other men.

Grand Ealls Advertiser, maintaining his correspondence, and listening to radio programs. But he did so for the most part alone, for his wife was busy tending both to Ethel Mae and to church affairs. So busy was she that she told him she did not desire the company of his frequent guests from Newfoundland. Pollett and his son's paths now seldom crossed. Young Ronald left for school before Pollett arose for the day. New courses training boys in wartime mechanical skills required him to remain at school until after Pollett left home for work on his regular evening shift. Occasional festive times dever the family together, and as can be surmised from a very

While at home he pursued his active interest in Newfoundland- keeping attuned with the

large collection of ornate greeting cards from Pollett to members of his family— for birthdays, Mother's Day, and other such special days— he was never one to forget such an occasion; but in daily life, Pollett's activities and interests and those of his family were on disparate paths.

In the spring of 1945 Pollett learned that his father had suffered a serious heart attack, and the news prompted a hasty journey to New Harbour. 94 Despite the wartime urgencies which made all travel difficult, he pressed the matter with authorities and obtained speedy passage on a modern seaplane from New York to Botwood, near Grand Falls, arriving at New Harbour with a plan to bring his eighty-year-old father back to New York. Perhaps he envisioned days spent chatting and joking with his father, providing Jim with the comforts of his home and valuing this addition to his family. He learned, however, that Jim had no desire to see the city. Pollett now purchased an expensive cabinet radio for his father to help make his convalescence more comfortable. Jim thoroughly enjoyed the gift, but he always kept the volume hardly audible. Pollett being unable to convince his frugal father that monthly electricity charges would not increase if he raised the volume. Pollett stayed in New Harbour only long enough to ensure that Jim was able to manage his chores independently. The speed with which Jim recuperated enabled Pollett to make a brief visit to Grand Falls and see his friends the Blackmores before departing for home. but again he felt it was too short a trip. It would seem, given the exigencies of his career and family responsibilities, that only in New York could be nurture his ties to his homeland. Subsequently, both his affinity for his homeland and his ideals of a simpler, far richer experience of daily life on the island were to intensify. For the time being, however, the image of a dedicated worker and provider was one that Pollett clung to. His conception of himself as an achiever and as a man of substance indeed was one Pollett had embraced for many years, and it remained a measuring glass to indicate the place he had carved out for himself in America and the world he had left behind.

LOOKING HOMEWARD: THE WRITING OF THE OCEAN AT MY DOOR,

Roa Pollett's life between 1945 and 1955 was characterized by disease and a longing for a return to his homehand. In mid-1945 he became seriously ill with heart disease, and there followed a ten-year struggle against chronic pain, financial pressures, and confinement to hospitals and home. Pollett took up the avocation of writing in 1946, his pieces focusing for the most part on aspects of his outport inheritance. He rapidly won popular acchaim among the readership of the <u>Adhantic Guardina</u>, a new mouthly magazine which was devoted to the interests of Newfoundhaders at home and abroud; and his early literary success led him to plan a second carer as an author. A visit to Newfoundhand in the summer of 1947 provided new insights and materials for his imagination and provoked a desire to retire there eventually. With this goal before him, he kept himself informed of changes occurring on the island as a result of Confederation with Canada in March, 1949. But on observing these changes during a summer visit to New Harbour in 1951, Pollett realized that he could no longer dream of returning to the pleasures of his boyhood, and that the familiar outport way of life he had loved was vanishing. As his health worstend considerably in the early 1950s, he aevertheless tried, both in his writings and his personal plans, to recepture the traditional way of life he had abandoned some forty years before.

.

This ten-year period of physical struggle, defeating Pollett's ambitions, was unmatched by any other setback he had so far experienced. In mid-1945 Pollett began to suffer from symptoms of beart disease: shortness of breath, stabbing chest pain, and dizziness; and as a result,

henceforth even the most ordinary routines would lead him to perceive the pace and demands of New York life as formidable obstacles. Climbing the fifty-eight steps at the 96th Street Subway station upon his return from work, for example, became an effort he would compare to scaling the cliffs at Powth Cove.² Although he was only forty-five years old, Pollett began to associate his chronic illness with aging, and later wrote, "While you're young and can bounce around like a rubber ball, everything is fine and dandy; it's afterwards, when you want to lie around like a dead fish, that the stairways and work and everything she get to be tough... for the New Yorker who must still believe he's as [fit] as ever." It will be noted that he referred to himself now as a New Yorker, although clearly the tag denoted none of the ideals of sophistication and achievement to which he had appired some twenty years before.

Pollett was never made fully aware of the terminal nature of his disease, called "acute mijocardial infarction." This condition was to be the cause of numerous major heart attacks, the first of which he suffered by the autumn of 1945. At this time, his chronic pain was accompanied by the strain of worrying over loss of salary and accumulating medical hills during his seven weeks of confinement at Brooklyn Hospital. And this experience, too, was exacerbated by a much longer period of enforced convalescence at his home, where an oxygen tank became a fixture of his bedroom, and uneventful days passed tediously. For an active man used to a rigorous daily work routine, succumbing to long periods of confinement would always be a source of frustration. In these circumstances. Pollett had to content himself with reading a great deal and enjoying as well as he could the pleasures of family life. As his wife recalls, their hedroom and parlour were "simply filled with library books and magazines and newspapers;"4 and he and Caroline now resumed the practice they had begun as newlyweds of reading together occasionally. Pollett often reading aloud. When his daughter Ethel Mae brought home a stray black dog, Blackie, he gave her permission to keep it, and the pet became a welcome addition to the household, for the dog would sit with Pollett as he listened to his radio, or delved into his small collection of books-- including the works of Shakespeare and a tattered copy of E. Everett-Green's Souib and his Friends, which as a boy he received as a prize. Caroline recalls that he now would often speak fondly of Downey, his own work dog to whom he had been devoted in his youth. He developed an anecdotal tendency in much of his conversation. His wife and daughter remember his daily practice of fixing lunch for his daughter, while he told her of incidents and customs from his own boyhood. Pollett was a doting father, his niece, Blanche Gilbert, recalls from a visit she paid to the family at the time.

She also remembers an aquarium Pollett bought for his daughter, and his daily practice of giving ber small sums of money for a new fish. Perhaps he began to contrast some of these daily, limited routines with the freedom and pleasure he knew in his youth. For although he now had as much time as he had had as a boy to enjoy the out-of-doors, he was unable to walk the ten city blocks to rest among the trees by New York Harbor, and even the simplest of his new domestic activities left him exhausted.

Early in 1946, however, when one of his friends gave him a present of Thomas Wolfe's autobiographical novel, Look Homeward Angel, Pollett at once found temporary release, it would seem, from the boredom of confinement.7 In this novel Wolfe, born in the same year as Pollett, wrote on such subjects as a boy's enthusiasm for learning despite an education at a backwoods school, and the excitement youths had felt as they imagined the glamorous life of a soldier fighting in the Great War. These descriptions of boyhood were comparable to aspects of Pollett's own life experiences- the ordinary, nevertheless, was rendered in Wolfe's prose lyrically and poignantly. This first introduction to Wolfe led Pollett immediately to acquire all of Wolfe's works. In these, too, he possibly recognized artful renderings of events that intimately reflected his own experiences and concerns. In Of Time and the River, Wolfe provided a highly detailed evocation of a cycle beginning with man's need to break home ties, to his ambitious experience in foreign cities, and finally, to his desire to return home.8 The enthusiasm with which Pollett read, often re-read, and spoke of these books suggests the impact of Wolfe's works upon his experiences and sentiments. Since boyhood Pollett had returned again and again to the works of Shakespeare for leisure reading but in Wolfe's version of "the seven stages of man" in You Can't Go Home Again. he possibly discovered a review of the passages from youth to aging that now spoke directly to him. Wolfe writes on the aging city man: "If he is lucky, he saves money;"but despite having it, "his weary eyes look upon the scenery of strange lands for which in youth his heart was panting, Then the slow death, prolonged by costly doctors," His first experiences of reading Wolfe possibly encouraged Pollett to develop an idea that he, too, had a story worth telling.

When Pollett was able to return to work in 1946, he had the opportunity to typeset a

manuscript called The Road to Wimbledon to be published by Charles Scribner and Sons that year, and this book too apparently provoked vivid memories of his own early experience. The book was an autobiography by Alice Marble, a former Wimbledon tennis champion, which began with a description of her childhood in the Sierra Mountains of Nevada- an area which though distant from Newfoundland, was in fact similar in some ways to the outport Pollett had known as a boy. 10 Marble's portrait of traditional life in an isolated setting was one Pollett easily could identify with her father had been a "high-climber" in the lumber industry, his a fisherman-farmer and woodsman; and her family, like Pollett's own, participated in a daily, rigorous work routine, She marvelled, nevertheless, at the comforts and rewards obtained in ordinary, rural family life. this despite her worldly experience. Pollett evidently read the rest of her story with great interest, although Marble, it can be assumed, was not known to him beforehand, for he was not a fan of professional tennis or of any other sport, even baseball. 11 But indeed she wrote very little about the actual game, instead depicting her defeats and achievements, both professional and personal, as universal human events. She also evoked an intimacy that engaged the reader in her story by perpering her book with passages such as: "I know that everyone is endowed with qualities of the champion, and can succeed in spite of handicaps in the most important game of all the game of life."12 According to his wife and son. Pollett wrote Marble a long letter conveying his appreciation of her work. She replied by sending him one of the first printed copies of her book, inscribed on the flyleaf, "For Ronald Pollett, 'My First Fan,' With best wishes and sincere thanks, Alice Marble 1946," 15 Thus as Pollett's latent ambitions to become a chronicler of his own life and times were now ripening, perhaps as a result of illness and physical confinement, he was spurred on by the example of Wolfe Marble and possibly other writers, who had made literature out of domestic experience.

11

Sometime in the spring of 1946, Pollett became aware of the creation of a new national magazine, the <u>Atlantic Guardian</u>, and saw in this journal, subtitled "The Magazine of Newfound-

land," the opportunity to express his own literary impulses, 14 The Atlantic Guardian was published in Montreal by three expatriate Newfoundlanders: Ewart Young, Brian Cahill, and Arthur R. Scammell. Its first issue had appeared in January, 1945; and in it a characteristic format was established. Each month it featured autobiographical sketches, fiction, and poetry confined to an island setting, graphic articles detailing historic events and future developments in Newfoundland. pictorial stories of island towns, and other items that would help achieve the editors' goal, "To Make Newfoundland Better Known to Newfoundlanders at Home and Abroad." In addition, each month advertisements appeared welcoming the submissions of native-born novices. Nevertheless, by June, 1945, owing to a deluge of unsolicited manuscripts of dubious literary quality, Brian Cahill had been forced to publish these guidelines to potential contributors; not only should submissions be brief and concise, but, he added, "Don't write about 'Sunset on Fortune Bay," or 'Boyhood Memories of a Land I Love.' Believe us...it's been done."15 Pollett's early literary attempts would conform to these directives but eventually his approach to Newfoundland topics also would establish his role among the magazine's contributors as a spokesman for the émigré Newfoundlander. Brian Cahill recalls that by the late spring of 1946, "We received a very nice, diffident letter from Ron from his home in [Brooklyn]. He said he was a native Newfoundlander who had come across the Atlantic Guardian and liked it and its role as a two-way window for Newfoundland. He enclosed some quite good bits of writing about life in the outports and we were delighted to publish them."18

Pollett's first two pieces of writing printed in the magazine were "The Outport Millionairs," in July, 1946, and "A Born Trouter," in August, 1946. In each, he contrasted the life of a harried, frustrated exile in New York with the simpler, more satisfying life of the outport Newfound-lander. The folly of chasing a dream in some far-off land when all the comforts of a full life were to be had on the island was now a prominent idea in Pollett's thought. His approach to the Newfoundland experience was also one perfectly in keeping with the goals of the magazine's managing editor and publisher, Ewart Young. Young was then part of a growing movement advocating Confederation with Canada. 17 The Atlantic Guardian was in part a vehicle for per-

stading Newfoundhanders to adopt a new political philosophy—namely, one recogniting that an istand heritage was both using and valuable, and that Newfoundhand could contribute to, and develop as an integral unit within, the larger country of Canada. The sentiments of an aging called like Pollett were seen as enhancing these views, and as a result, both of his first two vignettes were listed as "special features." As co-editor Cahall remembers of Pollett's early work, "the geatle, humorous—if perhaps idealistic—accounts of life in the outport submitted by Ron Pollett were just what Atlantic Guardian was looking for at the time. The fact that they were written by a Newfoundhanders living abroad was a pless in that we were...trying to put across the idea that Newfoundhanders living abroad was a pless in that we were...trying to put across the idea that Newfoundhanders living abroad was a pless in that we were...trying to put across the idea that Newfoundhanders living abroad was a pless in that we were...trying to put across the idea that Newfoundhanders living abroad was a pless in that we were...trying to put across the idea that Newfoundhanders living abroad was a pless in that we over...trying to put across the idea that Newfoundhanders living abroad was a pless in that we over...trying to put across the idea that Newfoundhanders." Thereafter, almost all of Pollett's submissions were given prominent attention, either by their heading an issue's table of contents, or by the use of "cassers" on an issue's cover, a typical example being, "lanide: A new story by Ron Pollett". These techniques alerted the magazine's audience to the new author's work, and, in the editors' view, his timely concerns.

Popular acclaim from the magazine's wide and notably large readership was won almost immediately. Pollett's uscomplicated, anecdotal style and beniga view of his outport heritage obviously streek a familiar chord. The glowing reception to his work in Newfoundland was expressed by one fan who, on the basis of Pollett's first two publications, commented that he would rather read Pollett than Shakespeare. When a German-speaking tourist to the island asked for and received permission to translate and publish "The Outport Millionaire" in a Swiss magazine, Pollett's dream of creating a new future for himself as a writer was enhanced. Almost overnight, in fact, he could begin describing himself as a professional author, since he earned from \$25 to \$45 per item. Yet this most ordinary and expected financial transaction between author and publisher nevertheless was rather remarkable. Many former contributions have noted that the fledgling magazine "was run on a shoestring" budget until poor financial management forced it to merge with the Altantie Advocate in December, 1957; and Brian Cahill has said, "Some 'contributors' were just that — unpaid!" Arthur R. Scammell, whose work had appeared regularly in the

magazine since its inception, has corroborated this remark, noting that he received no payment whatsoever either for his submissions or for his elidiorial assistance; and Michael F. Harrington, a professional journalist and poet, has said that, though he was paid a relatively meager sum (perhaps \$15) for each of his own submissions, on at least one occasion, owing to the magazine's poor financial straits, payment was long deferred and in fact never materialized. Pollett, however, was probably unaware of these circumstances. He was flattered by his early success, welcomed this new source of revenue, and was naxious to earn more.

Writing became more than a hobby for Pollett: it began to shape the nature of his daily routine.24 He now wrote for a minimum of four hours per day before going to work and during weekends. Hearing the incessant tapping of typewriter keys from behind his closed bedroom door, is perhaps the most vivid of his family's memories of Pollett at this time. Like many of his previous ventures and interests in Newfoundland, he obviously devoted himself to this new one with great enthusiasm. He was not restricted by deadlines, however, and was aware of only one demand by the magazine's editor-namely, to produce more works "with a salt-water tang to them." 25 Achieving this quality was in fact quite formidable. Working alone in his bedroom, composing on a manual typewriter set upon his wife's dressing table, he occasionally neered out the window not at the open sea but at a brick apartment house. The Majestic, across the street from his own, But despite this inauspicious setting. Pollett apparently achieved the "salt-water" quality his editor required in his next three sketches: "Let's Look at the Squid," published in September, 1946; "The Cat with the Yaller Face," in November, 1946; and "Where Christmas is Christmas," in December, 1946. In these he drew heavily upon boyhood memories of outport life. His confidence grew as each piece he submitted was accepted without excisions or other editorial interference. Furthermore, he was aware that scores of fan letters were received regularly by the magazine, and so knew that his talent was quite marketable. Young had found a writer who could sell.

Pollett's new commitment to writing brought forth altogether positive results. The Ocean

at Mr Door and Other Newfoundland Outport Stories, was his first published collection, and
appeared in mid-1947. In this work, published by Ewart Young's fledgling Montreal enterprise,

Guardian Associates Ltd., were reprinted the five pieces Pollett had published in the Atlantic Guardian during 1946, along with illustrations; a sixth item, the autobiographical essay entitled "The Ocean at My Door;" and a brief biographical sketch of the author. In sum, this small volume represented the literary achievements of the author to date, and foretold the direction of his literary efforts in the future. In his forward to "The Ocean at My Door" written in April, 1947. Pollett explained his intention- to describe the traditional Newfoundland outport "as seen through the eyes of a boy,"28 He further indicated that he saw this particular essay as the first chapter of a much larger work that would provide a detailed "record of the comings and goings. the ins and outs, and ups and downs of family life in the outport village;" adding that "The subject material for a readable book [using this setting] is limitless."27 Pollett thus was preparing to fulfill an intention dating back at least to 1924, though the deliberate use of a boy's perspective represented a change from his original conception. Co-editor Brian Cahill's unreserved enthusiasm for those of his writings, particularly his detailed story "The Cat with the Yaller Face," narrated by an outport boy, no doubt reinforced Pollett's decision to use this same narrative point of view exclusively. In a published review of The Ocean at My Door and Other Newfoundland Outport Stories, Cahill wrote that Pollett "manages with humor and unobtrusive sentiment, to convey the unique quality which 'outport' life, at its best, has to offer in early youth and adolescence."28 He added, "If he gives one side of the story only, however, he gives that side superbly. The work of the professional and amateur folklorist is too often pedantic and almost inevitably dull. Mr. Pollett's work suffers from neither of these handicaps."29 His first published collection reflected his editor's enthusiastic promotion of local literature; 20 and through his active correspondence with Young, Pollett surely was encouraged to embark on his second project and was assured that he had a ready market both in the Atlantic Guardian and in its parent firm, Guardian Associates Limited. All signs pointed to a bright future.

In the spring of 1947, however, Pollett suffered a second heart attack, and herein began a pattern that was to characterize most of his life as an author in middle-age. Bursts of writing were punctuated by long periods of silence. Enforced convalescence frustrated his new plans for a literary career. Nevertheless, his illness had caused his Newfoundland-born friends to draw closer to him and in nurturing a renewed relationship with the local community. Pollett also developed ideas for writing. Card parties resumed in his apartment, and on these and other occasions he would question his friends closely about their own experiences in the outports. Abe Snow recalls Pollett's keen interest in such matters as how Christmas New Year's, and other holidays were celebrated at his boyhood home in Clarke's Beach, Conception Bay, 31 This eager interest in the island obviously exceeded the casual and occasional interest displayed by his family and friends The topic of Newfoundland was, according to Snow, "a real mania with him," 22 Indeed, Pollett once described this avid attachment of his to Newfoundland as "summer madness," a seasonal "deep-down yearning for the homeland in summer."33 Whereas in winter "the happy pictures of the homeland" would fade, and city conveniences could be thoroughly enjoyed, for then one could lounge "with warm feet in soft chairs-no firewood to chon no water to fetch from the well and no cows mooing for hav"- in spring. Pollett said, "the roots we left imbedded in the home soil begin to breathe again."54 "Summer madness" was simply, "the going home complaint."55 His urge to return home no doubt was now spurred by his desire to refresh his memory with materials for his intended book. But he was wary of travelling alone in his weakened condition, and could not persuade his wife to accompany him to the island, particularly since he wanted to enjoy an extended, leisurely stay.

Perforce he merely read. Still essentially a novice, he perhaps was now influenced somewhat by yet another literary model along with Wolfe and Marbie- Edmund Wilson, a noted journalist and critic, whose work of Scion, Memoirs of Herate Country, recently had appeared. Memoirs, seen by certain critics as a rather unwieldy novel. in fact a collection of six distinct stories, their continuity deriving from Wilson's use in each of the same, quite inscrutable main character who is the narrator, and of a central location, Herate Country, a suburban summer resort near. New York City. During subsequent months and indeed later years, Pollett would adopt some of Wilson's techniques. He also would come to share somewhat in his own writings the concern of the narrator of Memoirs; the concern to chronicle the often shallow, materialistic values of many

characters be meets both in the country and in the city. Pollett's mimediate reaction to Wilson's work, however, was muted, as his son recalls; Memoirs was after all being suppressed as obsecue soon after its publication, though in fact only a few of its passages could be cited in justification of the banning." But perhaps merely reading of a summer resort wheeted still further Pollett desire to return home, there to observe a whole range of traditional activities he remembered in a village far simpler than Wilson's. Thus when Pollett was able to persuade his son to accompany him on a summer trip to New Harbour during July and August, 1947, he embarked on the journey with great esthusiasm.

m

The many comforts of improved travel he and his son now enjoyed may have caused Pollett to envision the island's eventually becoming a "tourists' mecca," 38 an idea perhaps encouraging him to look ahead to many more journeys even before the one in 1947 had begun. The pair travelled on Royal Dutch Airlines (KLM) from New York to Gander, then obtained first-class railway passage on "The Newfie Bullet." A mere glimpse from the train window would have assured Pollett that the beauties of the island remained unspoiled. Once in his native village, he fully expected to find the traditional, simple style of life he remembered to have remained unaltered, and materials in his first article written during his stay, "The Manly Sport of Handlining," provide every indication that he did find this to be so. In it he describes his first night and morning in the village; hardly had be taken off his hat before his father began speaking of the next morning's fishing outing. 39 Pollett relished the opportunity of fishing with his father, and found little changed in Jim's practices, such as preparing the night before by forecasting the weather and storing necessities in the punt, and then, despite some seventy years of repetitious experience, still eagerly looking forward to fishing at familiar grounds, and routinely cooking a dinner of fresh cod aboard the boat. Pollett's re-experiencing of these timeless pleasures obviously spurred his enthusiasm to see everything with a fresh eye; and indeed, in "The Manly Sport of Handlining." he attempts to catalog every maneuver on the boat, even how his father gutted and cleaned a fish. Though he adds somewhat sheepishly, "Leave it to me to note such triflest." he wanted to recapture the ordinary and mundane accurately and vividly. He apparently also spent much time questioning his father about events and characters he now only vaguely recalled, such as one incident concerning two men who had let their new boat burn at sea as they wondered where they could find water for dousing the fiames. So remarkably foolish had the men appeared to the people of New Harbour that a song entitled "Nicholas Tobin" had been composed to commemorate the event, and it was one Pollett as a youth had known well. 40 Other local figures such as Jake Javeis, the man who could drink tea from the bib of a boiling tea kettle, were also discussed with his father, 40 and again suggest Pollett's wish to reacquistal himself with the lore of his boybood.

Pollett's son and many villagers corroborate this impression of Pollett as a folklorist. His son says of Pollett's activity during this trip, "My father was like a sponge, just soaking up the [outport] atmosphere for his stories. He was always going off to visit someone, he really liked talking to the old men."45 Clarence Cranford and Ralph Higdon recall Pollett's keen enjoyment of meeting with local fishermen at New Harbour Bridge, noting that he "was always one for a yarn," and that, like his father Jim, he "could make a joke about anything at all,"44 But perhaps his son's most vivid memory is of his father's relaxed posture as he leaned over New Harbour Bridge in the evenings, where he often went after supper to fish for trout and eels, and a benien expression on his face as he looked about, one suggesting that he was musing, "Gee, it's good to be here."45 Taking part in these simple pastimes of chatting and fishing represented a welcome relief from city living and caused Pollett to think of his holiday diversions as brief but potent "tonics."46 His dwelling on these differences in the qualities of rural and urban life is most apparent in several vignettes which formed part of a subsequent article entitled "Vignettes of the Village." In one of these. Pollett reflects on the traditional virtues of endurance and frugality as he tells of a villager who once was unable to chop his own woodsupply, being so crippled by arthritis that he "couldn't pick up a \$1000 banknote from the floor." By the time Pollett observed him stacking firewood in the summer of 1947 the man's health had been restored, and the author explains why. The villager had switched to coal, but being afraid it might lose some "burning power" if left out in the snow, ingeniously contrived ways to haul it manually into his shed; and, in doing so, "curred" himself of his malady. It his was one experience that convinced Pollett there was a great deal to be observed in the quality of traditional village life worthy of being recorded in literature.

Widespread changes occurring in the village reinforced this conviction. For the traditional world to which he clung in his memory and sought out among the elderly villagers had not after all remained altogether static. A salaried mode of life available to many Newfoundlanders for the first time in the war years, had given many local people opportunities to start businesses, to travel, and to obtain previously unheard of luxuries. Tastes for American culture had developed. At Cranford's Hotel, where owing to his heart condition he arranged to take a special diet of plain fare, and at small local stores, jukeboxes had been installed, their magazines comprised almost exclusively of imported country music records. It was to these establishments Pollett would go to escape from the din and fumes of the equally recent occurrence of heavy local traffic, only to be met, he said, by "a tangle of raucous tintinbulations that jolted my pet fly sipping soda drops off the wood table."51 The tone and sentiments of an alienated local analyst would come to dominate many of Pollett's subsequent publications based on his experiences in 1947; and in one of these he wrote, "Hearing all that western wow was like meeting the Lone Ranger chewing hard tack in the jigging hole."62 Other changes, particularly among the younger generation, seem to have been just as incongruous and even more disquieting. Not without some disdain did Pollett explain that the rust on the clothes of local "young bloods" resulted from their new preoccupation with cars and trucks rather than from labouring with cod-trap anchors, and the callouses on their hands were from monkey wrenches rather than from splitting knives. 55 Being physically unable to participate actively in daily chores or even in the variety of pleasureable pastimes he remembered, he observed new activities among the young very critically, and consistently ignored the drudgery of performing traditional daily tasks from which he himself once had longed to escape. The marked changes Pollett perceived in what he termed "the new village"54 heightened his sense of nostalgia. Indeed in many of his subsequent writings, remembered tasks and skills would be presented as invaluable traditions and as virtues: but he was not reticent in voicing his criticism to the youths of New Harbour at the time of his visit. Of Pollett's conversations and attitudes that summer, one of his nephews. Ted Thorne, recalls, "He lived in the past," ⁶⁵

But the issues of change and progress were in no way limited to Pollett's small village. In the summer of 1947, propaganda broadcast over the airwaves and in pamphlet literature reached a national audience in preparation for the country's first referendum on Confederation on June 3. 1948.56 Popular views on the topic were voiced in every home and store. Some men, like Pollett's father Jim, were adamantly opposed to Confederation, fearing its impact on the traditional Newfoundland lifestyle; and Pollett's son recalls the speed with which the literature promoting this choice was relegated to the woodstove and outhouse his grandfather owned. Tollett himself could not be altogether critical, however, for it was becoming widely known that among the benefits accruing to Newfoundlanders from Confederation was Canada's social security program. The chance for his eighty-three-year old father to be relieved of financial worries and uncertainties in his last years was one which Pollett welcomed. He looked ahead, thinking how his own recurring illness might bring to an end his longstanding practice of sending monthly sums to his father. In addition, during this summer he no doubt learned of the will his father had drawn up in June, 1947, in which Jim bequeathed to him his house, lands, and other possessions, 58 Pollett saw in this the possibility of retiring to New Harbour. Much of his time would be spent hereafter envisioning the opportunity of beginning life afresh at some later date in the familiar village where, in part, "the wonderful world of the Newfoundland outport" he recalled from his youth still endured.

iv

As a result of this trip in 1947, Pollett's literary concerns and to some extent his personal plans became quite closely aligned with consemporary political issues in Newfoundiland. Evidence suggests that immediately upon his return to Brooklya he wrote a number of sensitive but critical articles, evoking his sentiments as an ex-outporter returning to his native village only to find tikeboors drowning the birdsons, and well-used hiraways conting the gardens with dutc.²⁰ In "The Village Goes to Town," "On Going Home Again," and "Vignettes of the Village," Pollett was now developing a different dimension in his writing from the simple urban/rural contrast that had characterized his early work. Instead he now nondered the discrepancies between the traditional outport lifestyle and the commercialized contemporary village life. Pollett's negative assessments of the Newfoundland village in cultural transition, however, obviously were not wellreceived by his editor. *O Young, an ardent supporter of the Confederation movement, no doubt perceived Pollett's new work as too blatantly anti-Confederate, or at least anti-development. The Atlantic Guardian found it difficult to maintain a non-partisan approach as the Confederation era dawned, and attitudes voiced in published editorials and articles were aimed at promoting Confederation, hand in hand with change and development. Thus, except for Pollett's detailed article, "The Manly Sport of Handlining," published in September, 1947, in which Pollett --being out at sea and away from the changing village- presented a favourable view of outnott life, no other article based on this trip was published until March, 1949. By the fall of 1947 he had either received his very first notice of rejection of his other material, or else had been given by Young some form of excuse for not seeing these other articles in print. This situation, lasting for a seventeen-month interval between publications, might have induced Pollett to submit his work to the Saturday Evening Post at this time. Here too his material was rejected. Pollett's son recalling that the event "piqued my father a little bit." By 1948, Pollett thus would have been alerted both to the highly specific nature and narrow appeal of his Newfoundland material and to the fact that his prospects of creating a new future as a writer were uncertain.

His health, too, gave increasing grounds for concern. Returning to his work refreshed and invigorated after his trip, his family and friends recall that he spoke frequently of retining to Newfoundiand, Ahe Snow noting however that "he never said exactly what he'd do there, he was too busy working." But by 1948 another heart attack interfered with Pollett's routine, this one perhaps the most severe he had yet experienced. He was forced to leave his demanding work at AD Fress permanently, thus losing the professional prestige accruing to legal printers. Financial circumstances at once prompted him to give up his habit of racetrack gambling, though evidence

suggests that he never completely lost his long-nurtured passion for it. Pollett's charghter recalls heated argaments which took place regularly between herself and her father when he simply changed the radio channel from her favourite programs to news of the daily racing results. The tax one circumstance which caused her to view Pollett as a stern father, a man about from family life, keeping much to himself. Perhaps it was his interest in horse-racing that led him, when he was able to return to work, to seek employment at Armstrong Publications, printers of <u>The Green Sheet</u>, a standard reserrack betting form. If the one lost interest in this work, however, and was to change jobs frequently in the next several years. His life now would appear to have been at something of a standstill, for he was a master of the linotype trade who could not obtain work satisfile to his killi; he was a gambler who did not gamble; and he was a writer who was neither writing nor publishing.

Despite these setbacks, Pollett maintained a strong interest in Newfoundland, and continued to think that he could make a contribution to the island's literature. In a letter to his editors published in the Atlantic Guardian in October, 1948, he commented on a number of short stories printed in the magazine that year that had made use of Hollywood-styled crime story formulas. while simply inserting Newfoundland settings and occasional native dialectical phrases. He assessed such stories as "The Fugitive" (August 1948) and "The Crucifix" (Sentember 1948), as "too, too lugubrious and not Newfie at all." Despite his awareness of the renewed interest in short fictional literature by native-born authors. Pollett said, "I am still hoping to see something in a short story that could be written only of Newfoundland...as only a Newfoundlander can interpret it. "66 In this same published letter of October, 1948, he suggested the small way in which he already had begun to realize that goal, and why he thought such contributions important; namely, to give a literature based on local traditions back to the people who lived them. By way of illustration, he described the reaction of one of his fans, a Mr. Mercer, who in the summer of 1947 undertook to haul a car in which Pollett and his son had been driving from a ditch near New Harbour. When Mercer learned who they were, he declined payment, saving, "I got so much fun reading 'The Cat with the Yaller Face,' I am well paid for this." 67 In 1948, though he had no

other literary outlet suited to his concerns, Polist nevertheless seems to have been acident timid nor retiring in voicing his opinions about the development of a native literature; on the contrary, he maintained a good-natured attitude, and an enthusiastic willingness to be part of this development, however humble his contribution.

Pollett's sentiments about the new drift of the magazine appear to have affected Ewart Young minimally. The pair maintained some ties, mainly financial ones, for throughout 1948, Young published monthly advertisements for Pollett's first collection. The Ocean at My Door and Other Newfoundland Outport Stories, and evidence suggests sales were brisk. In addition, late in 1948. Young mailed him 100 copies of the book, and Pollett's son recalls that he helped his father sell 95 copies in one evening at a dance at Prospect Hall. 88 This success might have been effective in boosting Pollett's morale. The incident also might coincide with Pollett's writing of an unsigned article that later appeared in the Atlantic Guardian, describing a Newfoundland-style square dance at the hall in Brooklyn. 50 It compares closely with some of Pollett's later writings particularly in voicing a somewhat critical assessment of the non-traditional style of dancing performed by the second generation of Newfoundlanders in attendance. A photograph of Dave Green, Pollett's former co-worker in the A.N.D. Company's Orders and Shipping department at Grand Falls, accompanied the article; and it might well be the case that Pollett sought out at least this one opportunity at the local dance to seek out materials he might be able to publish in the Atlantic Guardian. In addition, during the period, information for an article by Brian Cahill in the Atlantic Guardian was supplied by Pollett, who perhaps was too ill to organize and write the piece himself. 70 But there can be no doubt that the Atlantic Guardian had changed, and not much Pollett's liking. Brian Cahill, soon to leave his position as contributing editor, notes that the Atlantic Guardian, owing to Young's political bias, was moving increasingly away from its original purpose as a national magazine of general interest. 71 As one result, by 1949 a large number of firms had withdrawn their advertising. Perhaps as something of a response to this trend, in February, 1949. Young provided a five-year survey of the Atlantic Guardian's accomplishments, and some of these statistics, given the time and place, seem remarkable. Most

noteworthy were the circulation figures—since January, 1945, it was reported, 250,000 copies had been printed "which have been read (allowing what our circulation manager tells us is a conservative estimate of four readers to each copy) by 1,000,000" readers.⁷² It can thus be argued that the Adlantic Guardian, along with a number of newer local magazines (these, nevertheless, for less cophisticated, and often short-lived²⁷), was now encouraging the development of a popular local reading public interested in local issues. But it was an audience Pollett could not reach through the Atlantic Guardian until the Confederation issue was settled.

Events in 1949 brought a change for the better in Pollett's prospects. In March, he was given the title "contributing editor" at the Atlantic Guardian. It was, for the most part, merely an honorary title, one which he held along with authors Arthur R. Scammell and Ted Meaney, but perhaps it aroused in Pollett a feeling that he now had the opportunity to exercise a decided influence in literary matters and editorial policies. Such was not the case; although in the March issue, the last issue before Confederation, he finally had an article published, "On Going Home Again." By way of introducing Pollett as a new contributing editor, and summarizing his earlier publications, Brian Cahill wrote, "He returns to Newfoundland frequently and his stories of life in the outports have an authenticity and a skill in presentation that has made them widely read in and outside the island."74 In "On Going Home Again," however, this 'skill in presentation' is most difficult to appreciate. This fault lay not with Pollett, but rather, it would seem, with Ewart Young, who as editor may have decided deliberately on an oddly disjointed layout for the piece. Paragraphs in which Pollett criticized the recent commercialization and erosion of traditional lifeways were sandwiched among articles written by the nine premiers of Canadian provinces welcoming Newfoundland into Confederation, and various other pieces promoting the development of Newfoundland's resources and other notential innovations resulting from the political change. The effect of this choppiness was to minimize Pollett's role as something of a champion for an opposing point of view.

The key idea which was to concern Pollett in his non-fletion for most of his career nonetheless still can be gleaned from this particular article. He was now an advocate for gradual transition, and welcomed only innovations that could be adapted well to the traditional Newfoundland millies. For this reason, Joseph R. Smallwood's platform for the future of the new province—"Develop or Perish!" — never engaged Pollett's full support. Nor, for that matter, did a rather belated attempt by Geof Sterling in 1940 to away an estimated 75,000 Brooklyn-Newfoundlanders toward the alternative idea of economic union with the United States." In time, as Pollett perceived the growing impact of Smallwood's ideas upon the attitudes of Newfoundlanders, his non-fictional, descriptive writings would be characterized more and more strongly by sentimental adherence to traditional life and sacrasm toward the new, though in his autobiographical essays and in his fiction he would develop on another tack. At any rate, by June, 1940, the Confederation issue now settled, Young seemed ready to print everything Pollett wrote. In 1949-52 Pollett enjoyed his most productive years as a writer.

Pollett's new sketches and stories, among them "Epitaph for Downey" (June, 1949), "The Tongue that Never Told a Lie" (November, 1949), and "The 'Dicky-Bird' Dish" (December, 1949). evince qualities that were to distinguish many of his later writings. "Epitaph for Downey." one of the first new pieces published after the seventeen-month hiatus, does appear at first glance to be little more than an ordinary, sentimental account of a boy's loss of a beloved pet. Yet in it Pollett begins to create a vivid outport setting, mingling his extensive knowledge of a distinctive outport landscape with the native folkspeech that best describes it. He was at last evidently making use of some materials he had reacquainted himself with during 1947, and now did so not to criticize but to illuminate that other world. His ability to evoke the outport setting would soon be praised by one reader, who said of his works. "It has been a great joy to go back over one's boyhood with Ron Pollett, to smell once again the clean fragrance of birch billets and taste the sharp tang of marshberries."77 Pollett's sentimentalism in this piece is, as well, redeemed by engaging in humorous editorializing. For instance, he describes how he and his work dog Downey tried to shun the many difficult tasks inherent in getting and hauling the family's wood supply, thereby bringing forth his father's timeworn admonishment: "Hard work never killed anybody:" and to this Pollett now appends, rather chummily, "which of course was not true then any more than

today. "** But he continues the narrative by describing how, imitating his father's grulf voice, he as a young boy would use the same "saw" to his dog. This suscephilatested yet intimate humour was to prove very endearing to the local readenhip of the day. The timeless quality of an outport setting rich in beauties and lore was one he schieved again that year in "The Tongue that Never Tods a Lie" and "The 'Dicky-Bird' Dish' by focusing on more aspects of his own boybood. In steeping his imagination with the materials from his youth, Pollett was perhaps again becoming more attuned in 1946-50 to pursuing his long nutrured plan for a full-length book telling of the outport world "as sees through the eyes of a boy." Indeed in the March, 1949, issue of the

Atlantic Gancilan Brian Calaill wore, from recent information received from Ewart Young, that Pollett planned soon to write such a book. ⁷⁰ Whether or not such antobiographical sketches and short stories as he now was publishing were the seeds of it, the plan would bring forth potentially new developments both in Pollett's life and in his career.

•

Anids the literary ups and down he experienced since 1946, Pollett no doubt was made aware how closely his fortunes were linked both to the <u>Atlantic Guardina</u> and to its editor, Ewart Young. Sometime in 1980, an opportunity to forge even closer ties presented itself to Pollett. He received a promising offer of employment both as a printer and as a writer for Guardina Associates Limited, the parent company of <u>Atlantic Guardina</u>. ⁶⁰ Ewart Young, managing director of the company, was in the process of moving its operations to St. John's from Montreal, and he travelled to New York partly to try and persuade Pollett to join his permanest staff. Young had recently received a loan of \$50,000, and prospects for his new ventures seemed bright; as well, to facilitate Pollett's move Young had located a suitable apartment for the Pollett family in St. John's. These plans, together with the enthusiastic nature of Young's character for which he is well remembered, no doubt swelled Pollett's own hopes. Though he was not a not polunge hilindly into new undertakings, he now eagerly broke the news to his wife. He learned, however, that Caroline damanatly refused to accompany her hubband to Newfoundhad and to "iters up or

roost in Brooklyn. "He was going!" she now maintains, "If not for me and the children, he would've gone back there then." a Perforce Pollett settled for an alternative. In 1950 he invested substantially in the <u>Adlantic Guardian</u>, a choice which suggested his confidence in a new Newfoundland venture." He would continue to be involved with the intellectual and industrial future of his homehand—but, as before, from afar.

Perhaps as a result of Young's offer to reestablish himself on the island, by mid-1950 Pollett's writings began to show a new emphasis upon the benefits of Confederation. In his extended essay "There's No Place Like an Outport" (July, 1950), he emphasized such novelties as the convenience of a new system of highways (a major project of the new government), these making it possible for natives and tourists alike to enjoy the indelible, natural beauties and wonders of the island. Pollett wrote of "epicurean delights" and "pristine beauty" as one could guess a travel agent might, and indeed the Newfoundland tourism industry was one key emphasis of the new provincial government. 83 The uneasiness he had felt in 1947 toward aspects of the new money economy and local commercialization now were referred to as merely matters of personal taste; and in fact, in his short story "Rum in the Puddin" (December, 1950), Pollett presented an uncharacteristic panegyric on the changing post-Confederation lifestyle. He emphasized its benefits, particularly for elderly Newfoundlanders: "Now, with their social security pensions," he wrote, "they were finished with scraping the pennies and were living like decent people should instead of just existing. The fact was they couldn't afford such luxury until a time a while back when they went to sleep Newfoundlanders and woke up Canadians."84 Though he focussed upon mundane matters such as financial security and the peace of mind which Confederation brought specifically to elderly Newfoundlanders, in this story Pollett was able to achieve both humour and a fairly realistic portrait of hard-working elders receiving their just deserts. In recreating a family incident that had occurred in 1933,85 the story became a depiction of new lifestyles on the island, which reflected his lingering desire to retire there and his positive vision of aging in that changing, though in many ways still timeless, world.

These new sentiments appear to have prompted Pollett to write a number of articles on the

contrasting ways of life among Newfoundlanders in New York, which were published between 1950 and 1951. He first of all actively set about researching aspects of the emigre experience, doing so through interviews, study of census data, and reading surveys about New York. Perhaps quite unintentionally at first. Pollett seized on a topic that as yet remained largely unrecorded in the annals of American social history. For he progressed from a short biographical article, "Captain William J. Connors" (January, 1950)- an account of one Newfoundland sea captain and Brooklyn dockyard official- to vivid accounts of the entire Brooklyn-Newfoundland community in articles such as "Summer Madness" (July, 1950) and "Memories of 'Didder Hill'" (February, 1951). The factual information provided, such as the large contribution made by Newfoundlanders in building the Empire State Building and the George Washington Bridge, focussed for the first time in print on the ethnic identification of the workers rather than, as usual, on the New York monuments themselves. Brooklyn neighborhoods where Newfoundlanders lived and places they congregrated were delineated, for the first time in print distinguishing the group from the neighboring Irish residents. As well, Pollett drew on the general sentiments and outlook of the Newfoundland émigré that at times could serve to distinguish them from the larger population of some eight million other city residents. In "Summer Madness," for example, he wrote this of Newfoundlanders sitting in Central Park in spring: "Where city-bred people...see only rowboats on the park lake we envision tall-sparred schooners, white sails spread... for the spring airing, mirrored in the harbour calm." 26 And a three-part series of articles entitled "Up in the Big City" provided the broad scope in which to emphasize in detail the demanding routine imposed by city life (everyday "shooting with both barrels until you are out of breath" 37) as a clear antithesis to the outport idyll. Possibly as a result of writing these articles, the simpler pleasures he imagined of an island retreat again loomed as large as they had in 1933, when his physical condition was also weak and his finances unstable. In addition, his eager desire once again to return to his homeland was made keeper by recent reports be received from local friends who had travelled to Newfoundland since Confederation, and who claimed that life there had changed "1000 per cent for the better,"88 Their sentiments, corroborated by many letters and journals Pollett read, provided the

incentive to plan a family trip to New Harbour in 1951.

Pollett embarked on this journey in August, 1951, as enthusiastically as he had on any before. Despite the superficial changes he was prepared to perceive in the post-Confederation outnort he nevertheless had great expectations of finding life in the outport idville in comparison to the exhausting tempo of life in the city. He was planning to impart "the wonderland of fun" of his own outport childhood to his ten-year-old daughter Ethel Mae, who "had never been off the city payements."89 Taking her fishing and berry-picking, and showing her how to skip stones at the landwash, could enable him to relive exactly the same pleasures of youth he had once known. She recalls nonetheless, very possibly because of his illness, that he spent relatively little time with her during their month-long stay, 20 Other limitations on his plans were equally disappointing. He had looked forward to lounging on the veranda of his boyhood home and sharing with his wife the pleasure of viewing the scenic panorama around the harbour, which he termed the "Newfoundland Riviera."91 Certainly it remained beautiful, but in an unsettling new way: widespread and rapid economic changes had resulted in the erosion of the inshore fishery. "The beach has become as bare as before John Cabot." Pollett later wrote. 22 Though as early as 1947 he had written of the villagers that "the new generation and most of the old are occupied with other things, away from the shore,"93 adaptation to the new lifestyle now seemed decisively complete. "I was living in the past: scarcely anyone...gets up that early anymore:" indeed "there are scores of villagers...who never dip a codhook all summer long!," Pollett wrote with unsuppressed amazement as he described his difficulty in locating someone to go fishing and angling with. 24 All of these experiences no doubt led him to the disquieting realization that he was now at home, and yet home was a strangely unfamiliar world.

There was one pleasing consolation that he could dwell on when he grew retelest. Many residents of New Harbour now subscribed to the <u>Atlantic Guardian</u> and so knew of Pollett's publications. He was indeed seen as something of a local celebrity, both as a former Villager and as a former islander, for news of his travelling to the island had been the subject of a broadcast on radio from St. John's, and Pollett himself, after his arrival, also cave talks on radio from the magazine's offices in St. John's. Clearly he thought that his role as a native-hora author and the attention he received elevated his prestige locally. He wrote, "And who knows—you may even rate as ask' to join these private picnic excernions of the village elite! Here you can gorge on [a variety of delicacies] while sitting on the same roadside rock where you nibbled your tom cod and hardtack and sipped your switchel as a boy. Oh, a lot of these things can happen to you now." "At his involvement with writing could provide an escape from what he perceived as his lowly status as a faberman's son. After some thirty-five veras, this still mattered.

He now pursued his interests in writing by gathering materials for what eventually became a series of five articles based on this month-long trip. One nephew, Charles Thorne, recalls that Pollett "was always writing his stories,"98 and it is apparent that the author deliberately set about interviewing local villagers about their changing lifestyles. He noted that "the old folks don't go over the hill to the poorhouse anymore. They go over a hill, all right, but it's down to the post office to collect a monthly pension cheque from the government:" then he visited local merchants to ask how the monies were spent or saved. 97 He learned from them that old and young customers alike demanded only the most modern merchandise from mainland catalogs. So too he interviewed, among others, women whose husbands now were employed in salaried positions at distant centres, and then described his findings. Much of his writing, in short, attempted to take the form of detached reportage, while in other parts of his articles he developed an equally noticeable tendency to preach about simpler times and traditional values. Of children's former pastimes, he wrote, "Maybe there are new things just as good to replace all that. But I failed to note any last summer. Here endeth the lesson,"98 Observing and writing themselves no longer seem to have been activities that enabled Pollett to escape from the contemporary world. For documenting in great detail the former rich and distinctive way of traditional life taxed the imagination; while describing the new led him to imagine, often ironically, the total demise of outport culture in the very near future. He wrote, "Grandpa sits around on his Old Age Security, his mouth watering for a meal of scrod as he snags a frozen fillet from the grocery store. And his grandchildren if they ever hear mention of rounder or leggie probably think it is some sort of bird."99

Pollett apparently returned from his holiday in New Harbour determined to achieve a new purpose in his writing. In late 1901 and 1902, he embarded on his most ambitious literary project, Peter the Grate: An Outport Character Study. This work was a sixty-odd gedetailed account of a typicial outport figure, one Pollett had himself scalier termed the "outport millionaire"—a grown man or, rather, "an overgrown boy" who lived all his days "with a gun in one hand and a trout pole in the other," 100 The character was not only an expert outdoorsman, but also, though illiterate, a superb storyteller and songmaker. "Peter" was in fact an agglomeration of personalities Pollett had known during his youth; and the figure's habits and traits were ones, so Pollett thought, that could not withstand the sweeping changes in the contemporary with lage. The death of Pollett's own father Jim in April, 1902, no doubt reinforced his desire to preserve on paper his people's distinctive traditional lifeways. Though writing of Peter as seen progressively through the eyes of a young boy, an adolescent, and briefly, an expatriated adult, Pollett was able to review outport life during sime span of some fifty years. "As any folkforist will recognize," he said in his introduction to the work, Peter "provides a most convenient peg on which to hang a mirror of the times," 111

It was, as well, a highly imaginative work that attempted, in using a first person narratorostensibly Pollett himself—to relate in a traditional storyteiling style a "tree" story of Peter's
life. In part, this particular narrative technique might account for the immediate success in sales
of the completed book, especially in the Trinity Bay region. In New Harbour, however, the reception was hostile. ¹⁰⁰ Opinion in the community opposed Pollett's interpretation of outport life and
of Peter, and the book was thought little short of recardalous. Some readers were outraged that
Pollett, as a former villager, could publish a work whose artistry was based on esoteric village
gossip about an actual towarman. In retrospect, it is perhaps revealing that Peter, the central
figure of the book, has been identified by towarmen as distinctly one particular individual, but at
least three separate names have been identified to support this claim. Nonetheless, at least a few
villagers took, and some still do take, considerable pains to ensure that the look should remain
concealed from a former community member, who is ostensibly the daughter of the long deceased

"Peter the Grate."

Whether Pollett was made aware of the local furor he had created is not known. If he was, it evidently did not thwart his ambitions. Possibly his long concentration on Peter, who "never punched a timechock in his life" and "always mixed in where there was company and fun," long reinforced Pollett's own plan to retire to New Harbour. As his wife recalls, having received his father's legacy, now "he wanted to build his own house there," despite her arguments that there was nothing for them to do in the village. 3th shouse there," despite her arguments that there was nothing for them to do in the village, 3th shouse the rapidly declining health, this in view of the limited medical services available in the outports. For by 1952 Pollett was not able to work steadily as a printer, and in the next few years he earned a meagre disability pension of about \$100 per month. 3th Nevertheless, despite chronic illness, his literary output had been rather extensive; and evidently be thought that with this source of revenue added to his pension and savings, he could return to live in the village and partake of the simplicity and humble values still obtainable in New Harbour. Possibly he hoped to savour what he thought was yet left of the old lifestyte—fishing, the natural surroundings, the cammaderie at the bridge.

Such an Illusion served him well as his fortunes and health declined considerably in the early 1950s. It was a rude shock to receive a telegram from his editor and publisher Ewart Young, by the summer of 1952, urging him to send money in order to keep Guardian Associates afloat. 1967 The bankruptcy of the firm by November, 1952, was a blow to Pollett, for he lost an investment of more than a few thousand dediars. Only now, as his wife recalls, would Pollett agree with her that he had been wise not to become entangled more intimately with the venture. 197 Yet the failure of the business caused the <u>Atlantic Guardian</u> to remain out of print from November, 1952, to April, 1955, removing his usual outlet as a writer. However, hospital stays were now becoming more frequent and difficult, for he often could not rest there or at home until the received an injection of morphine; and thus it is doubtful that Pollett was now willing or able to seek out other literary outlets or to experiment with other subject material. All communication was difficult, although his son recalls Pollett's adamant refusal of his family's offer of a television set, giving as his reason, "You can't talk to the people" while a television set was playing. 198

His writing had become one of the few sources of personal and intimate communication left to him, and NewYoundland life his major topic in conversation and in literature. His friend Abe Saow indeed recalls how difficult it was to visit Pollett, for even on the day before his death, he "saitli believed everything would turn out fine. He was so optimistic it hurt to see him, and hear him talk above going back to NewYoundland when he got better." ¹⁰⁰

One event late in 1954 no doubt further spurred him on to visualize himself in those new circumstances. In a teny-zear review of the Atlantic Gunzilania' achievements, Pollett was selected, on the basis of fammail, the magazine's "Favourite Storyteller," and, as well, his story "Run in the Puddini" was voted the readers' favourite work. 190 This was the story in which Pollett had best conveyed his positive sentiments on aging and cultural change in the outport, and perhaps as a result of earning such praise, he attempted to resume publishing and to plan for the future. He worked only intermittently, for he had suffered several massive heart attacks by 1955; in mid-June, 1955, however, his plans for writing were extensive. In an appendix to a letter to his editor Ewart Young, Pollett sent summaries of more than a dozen articles and stories he intended to write. Beginning his appendix with the heading "Suggested Subject Material That Might Help AG [i.e. Atlantic Gunzilan | Come to Life, Pollett enthusiastically went on to describe the new kinds of material he was "burning" to deal with. 110 Despite his enfeethed condition, his outdook remained, as Walter Blackmore recalls, "keen and full of idea." 112

Pollett died on September 9,1955, as his Brooklyn home. According to the accounts of his family and friends, his idealism, his optimism, and his decisiveness, all were traits that distinguished Pollett's outlook even as death neared. The only indication that Pollett was aware of the seriousness of his condition was suggested in a letter to his brother Jim, written in mid-July, 1955. Pollett said, "Bor, if I could crawl at all, I would be in New Harbour this weather and out on the Rock [beside New Harbour Bridge]... But the way it looks, I doubt I'll get off the room at all," ¹¹³ He noted that he was unable to travel without the assistance of his wife, for he needed her to administer injectious, and he added, "I look like hell in a gale of wind; my hones rattle like a windjack. In fact, I really don't want to go anywhere smong people I know looking the way I do

now." A Alarmed by the letter, Jim and his wife travelled to Brooklyn in August, visiting him each day in Brooklyn Hospital, where he had been conflued all that month. Bertha Pollett recalls that he wanted to talk only about New Harbour, who are several days after their leavetaking in early September, he asked to be taken how here for some days, deliriously, he continued to speak of events that had occurred some fifty years before. It was such a circumstance that has caused his son to reflect, and his wife to corroborate, "that if he had survived... he'd be there now on the bridge with his split bamboo fishing road."

Pollett's embracing of that simpler way of life was the focus of various obituary notices and tributes; and, a year later, the compiling of most of his published works in a large volume entitled The Ocean at My Door was itself a response to a widespread desire for preserving his vision of the Newfoundland outport. At the time of his death, a comment by Ewart Young, in a published editorial suggested that the intimacy which Pollett had striven for in conveying his unique view of the outport had been achieved. Young wrote that the "news of Ron Pollett's sudden passing, must have brought a sense of personal loss to many thousands of people in Newfoundland and abroad."117 The editor of the Grand Falls Advertiser, in praising Pollett's ability to portray Newfoundland outport life, corroborated Young's assessment and furthermore noted: "His was a gift rarely encouraged, and his passing is a great loss to Newfoundland literature." 118 But laudatory as this comment was, it is worth noting that few writers of these tributes, which as Young reported were given wide and sympathetic attention in various media, mentioned any details about Pollett's personal life. Ewart Young, in a personal letter of condolence to Pollett's wife, said that he himself was "shocked" by the news of Pollett's death, for he had never guessed the seriousness of Pollett's illness. 119 Indeed the majority of writers of Pollett's tributes disregarded his life as an author and instead focussed on the nature of his writings, and of the outport world which he rendered with "a color, a vividness, a pulsating alive-ness," 120 As well they should have. For his writing was, largely, an escape from the urban world and the pain and frustration of daily life. He wrote mainly of quite another world, the outport he knew, idealized, broaded over, and hoped to know again. For Pollett's success in New York had been fleeting and sporadic: and despite his own better judgement, he seemed, to the very end, determined to disprove the notion that "you can't go home again."

DOLLETTIS WORKS, DECARTIBING AN OUTPORT HERITAGE

Ros Pollett pursued a career as an author only late in his life, and his output was relatively small. he produced two volumes of prose in his lifetime, and in a third volume, entitled The Ocean at Mt Door, thirty-three of his known published pieces were reprinted posthumously in 1956. The extent of Pollett's genzre and his lifeting interest in authorship are described in the first section of this chapter. Pollett's three major publications comprise autobiographical writings evoking his sentiments as an finging Newfoundlander; and the second section of this chapter examines the development of the literary style in which he defined the nature of his experiences at home and abroad. All of Pollett's work appeared in print during a period of cultural transition in Newfoundland. How Pollett's writing relates to that context, but was also touched by literary influences from the world beyond Newfoundland, it discussed in the third section of the chapter. The fourth section of the chapter provides a concluding critical assessment of Pollett's achievements as a writer.

ī

Although most of Ron Pollett's approximately forty known pieces of writing were produced between 1946 and 1955, from an early age he had an interest in storytelling. "Eeling," a poem he composed when an adolescent in New Harbour, was among his first literary efforts; it described an incident which occurred among some local boys who set their eel lines off New Harbour Britise. Its humour and exactases of localized detail are qualities towarmen recall of "Eeling," The poem played some role in local eatertainment, for such pieces once were recited frequently by Pollett and his contemporaries. In Eeler the Grate: An Outgort Character Study, a semi-decionalized autobiographical work first published in 1952, Pollett recorded single stanzas of several rhymes which he attributed to himself and other local towarmen, referring to them as "strictly homespun." Among the examples is a slight rewording of one stanza of "Welcome Robhi," a

poem contained in the Grade Two <u>Borral Brader</u> school text; and Pollett is depicted as a boy of about seven as he retikes. "Elopping o'er the carpet/Picking up the crumbal/Peter knows the Polletta/ Feed him when he comes." None of Pollett's poems composed in New Harbour was published during his life there, however, nor was "The Exploits Hotel," a poem he wrote (but did not print) in Grand Falls during the early 1920s. There, too, he is said to have established his reputation as a writer after privately circulating his poems. A third known item, entitled "A Brief for Appellants," was a prose vignette featuring fellow workers in his printing office, composed and printed during his years in New York; it was circulated privately, and no copy is known to exist. These three known examples of ephemers indicate that Pollett maintained an occasional interest in the pustime of writing throughout his early years.

In 1924, Pollett spoke of a book he intended to write "dealing with the homelife of his prople in NewYoundland" in an interview published in the Brooklan Dailt. Eagle. * This evidence suggests that Pollett did have plans to pursue writing actively in his early twenties, but how seriously we can take this youthful decharation remains questionable. As early as 1919 he is said to
have written occasional newspaper articles while living in Grand Falls; the same was said of him
when he settled in New York. Specific articles attributed to Pollett have not been located,
although an absence of a known by-line might occount for this gap.* Lack of information on the
identity of a "Special New York Correspondent" whose work appeared occasionally in The
Newfoundland Warkk (Boston) may further impede a study of Pollett is involvement in journalism in the mid-1020s. During these years, when Pollett is believed to have worked at the
Brooklan Dailt Eagle and to have acquired the habit of reading a minimum of five metropolitan
newspapers per day, articles about Newfoundlanders in New York, some first appearing in the
Eagle, were sent to The Niewfoundland Weekly for respirating; and in the April 25, 1925, issue,
one article in particular suggests comparisons with Pollett's known writings.* The article
of the property of the particular suggests comparisons with Pollett's known writings.* The article
of the property of the property of the property of the pollett's known writings.* The article
of the property of the proper

acewpapers per day, articles about Newfoundhaders in New York, some first appearing in the Earle, were seat to <u>The Newfoundhand Weakh</u> for reprinting; and in the April 25, 1925, issue, one article in particular suggests comparisons with Pollett's known writings. The article described a lecture and slideshow entitled "A Night in Newfoundhand," presented by Major R.H. Tait; and the exactness of detail and sophisticated vocabulary, describing the delight and "voiferous appliance" of expatriated Newfoundhanders viewing lanters slides of their particular birthplaces, resemble somewhat the extant records of Pollett's minutes of vestry meetings at English Harbour. Furthermore, its description of these pictures, which enabled members of the audience "to rekindle the memories of boyhood days spent in the old homeland," is remarkably similar to a passage in Pollett's article "Memories of 'Dolder Hill," published in 1951. Though this notion of expariated Newfoundlanders sharing a special relationship to the island characterized all of Pollett's later works, so typical might this sentiment once have been among the group be lived with in the United States that suggesting other attributions in The Newfoundland Weekly seems unjustified. It is known, however, that Pollett's extensive involvement in the Grand Ealth Advertiser, beginning in 1950, was based on a sophisticated knowledge of the techniques of journalism. Pollett contributed highly detailed critiques of articles printed in the Advertiser, and the thought lingers that these could only have come from somebody experienced, not just in printing and reading newspapers, but in writing for them.

The publication of Pollett's vignette "The Outport Millionaire" in the <u>Atlantic Guardian</u> in July, 1946, Isaached his second career, that of an author, and one who was essentially a spokenman for the <u>émigra</u> Newfoundlander. This sketch described the seasonal pastines of a typical "lucky" figure, one who "seeds no money" and has an "evivable capacity for eigyment of simple things," and as such represents a clear antithesis to the mundane routines of hundreds of outport immigrants "who have spent their years roving around the world and worrying about the little things." In subsequent early vignettes and essays. "A Born Trouter," "Let's Look at the Squid," and "Where Christmas in Christmas," he chronicled in sensous detail outport customs and traditions, and the pascity of comparable erban experience again was emphasized. All of these publications established for Pollett an important relationship with the editors of the <u>Atlantic</u>

Guardian, who sought to use the fledgling magazine as a vehicle for "Making Newfoundland Better Known at Home and Abroad." While the magazine's monthly column "Newfoundlanders Abroad" emphasized achievements of expatriated Newfoundlanders, Pollett's focus on the feelings of the alienated immigrant was a response that at once emphasized the tenacity of outport

upbringing and seemed to corroborate the editors' own view that Newfoundlanders could in fact

live anywhere and remain "true Newfoundlanders." His approach to Newfoundland topics thus was both welcome and timely.

Pollett's works proved to be so popular that in mid-1947 the editors undertook to reprint his first six efforts in one small volume entitled The Ocean at My Door and Other Newfoundland Outport Stories. This publication was a milestone in Pollett's early career. The book sold well, and received unreserved local acclaim. Furthermore, a translation of his earliest vignette, "The Outport Millionaire," was undertaken by a German-speaking visitor to the island and appeared in a Swiss magazine. These achievements encouraged Pollett to develop an emphasis upon the rich outport world-its customs, values, and dialect. Indeed, two of the most popular items included in the volume-"The Cat with the Yaller Face" and "The Ocean at My Door" -- simply present a boy's fresh, uncritical perspective upon objects normally given scant attention. On the publication of his book, he indicated that the outport world as seen from a boy's perspective would be his major topic henceforth; and he concluded the autobiographical essay "The Ocean at My Door" with the notion that he had a good deal yet to say. He wrote that during his years in New Harbour "the many interesting things that happened in our family life and out into the wonderful world of this colorful Newfoundland outport- around the wharves in the hoats at school and among the woods and streams-- would make a long, long story."9 Shortly afterwards, in his extended essay "There's No Place Like an Outport," he explained further that "it would take a couple of thick books to cover in detail the few years I spent in the village." Pollett continued to speak of such a book frequently in essays and letters between 1947 and 1950. By 1949 he had earned the title of "contributing editor" at the Atlantic Guardian, where he continued to publish his work. But he never produced quite the book he envisioned, one in which he would tell his own story and tell it "as seen through the eyes of a boy."11

Between the publication of The Ocean at Mc Boor and Other Newfoundland Outport Stories, and Pollett's second book, entitled Peter the Grate: An Outport Character Study (1952). The produced twesty-three pieces, comprising seventeen descriptive essays and six short stories. Although in 1947 Pollett pictured "The Ocean at My Door" as the opening chapter of his prejected book, these publications could not be regarded as subsequent chapters. In the essays he continued to focus on the experience of the native Newfoundianders as contrasted to that of his own in New York, although he added new dimensions to these accounts by describing in greater detail the immigrant's urban lifestyle and by chroniciling the changes he perceived in the outprot during two summer excursions to New Harbour in 1947 and 1951. The breaking down of community customs and traditions was emphasized in his essays, but in his short stories he resurrected the vanished past, the virtues of traditional outport life bring generally his theme.

Cally rarely did Pollett's works during this period between 1947 and 1952 convey a positive outlook on the contemporary outport world.

Given Pollett's personal situation and attitudes in the early 1950s, his second book, <u>Peter</u> the <u>Grate</u>, <u>An Outport Character Study</u>, published in 1952, is not as curious a work as it first appears. Many indications suggest that this was a book Pollett had long intended to write. It was divided into fourteen short chapters, and he included much detail from previously published essays and stories, particularly material from "The Outport Millionaire," "A Born Trouter," and his most recent series of articles based on life in the post-Confederation outport. Although several chapters depict the "millionaire" figure, Peter, from an outport boy's perspective, the book is in fact a sequential narrative spanning some 50 years. It concludes at Peter's death, which coincided with Confederation, an event Pollett increasingly saw as a threat to traditional outport community life. <u>Peter life Grate</u> was published as a "contribution to the slinds' folklore." ¹²

Pollet's fanal eight pieces, published in the <u>Atlantic Guardian</u> between 1982 and 1985, all are attempts to grapple with a changed outport world. He assumed the role of recorder of tradictional content of the property of the property of the Dark," and "The Passing of the Stagehead" all display a propensity for cataloguing a rich variety of custom and lore that, to Pollett's mind, had disappeared quickly as a result of recent modernization. The acclaim he won from <u>Atlantic Guardian</u> readers in 1984—he was elected as their "Favourite Storyteller" and his "Ram in the Puddia" (December, 1980), their favourite story—might have encouraged him to redirect his attention toward fiction and to attempt, as he had

only once, in "Rum in the Puddin'," a gentler treatment of the theme of the outport in transition.

"Uncle Ben's Adventure" (September, 1964) and "The Matchmaker," published posthumously in
February, 1956, succeed in this emphasis. The first was an exagerated depiction of a notion Pollett had mentioned frequently in essays—the importance of local radio programs to older outporters. Although he previously had suggested that nightly broadcasts had replaced traditional

storyteilling in the village square, in "Uncle Ben's Adventure" in described how an older man,
using navigational skills he had learned on sealing voyages fifty years before, successfully fought
his way through a blinding snowstorm in order to be on time for the nightly "Doyle News Bulletin." The second piece, "The Matchmaker," also affirmed the breaking down of custom in the
contemporary outport, particularly the custom which "chained a childless spouse to the headstone
of his dear departed for at least a year." ³⁸ Events are narrated by a romastically-inclined
thirteen-year-old girl, perhaps making "The Matchmaker," Pollett's most unusual work.

Only a few months before his death, Pollett indicated that he had a good deal yet to say about the changing times. A letter to his editor, Ewart Young, included summaries of approximately twelve articles he intended to write, on topics ranging from the novelty of an excursion on The Newfle Bullet' as compared to mainland trains, to the relationship between mink farming and the traditional whaling industry in Blaketown, Trinity Bay. 14 "The Passing of the Stage-head" (October, 1955), however, is the only one of these articles Pollett managed to produce before his death in September, 1955, and though there is evidence that he intended it to be humorous, 16 it is merely a record of a painful and failed personal attempt to reconcile himself to the stark physical changes which had occurred in the contemporary Newfoundland outport. As he saw it now, "Gone up in chimney smoke or drifted out to sea" was all evidence of the traditional inshore fishery; the setting was "as bare as before John Cabot." These comments suggested that all traces of the special outport world be had known and later tried to chronicle had disappeared.

Pollett's third major publication appeared posthumously. The Ocean at My Door, published in September, 1956, contained thirty-three of his pieces, most of which had appeared in the

Atlantic Guardian in 1946-55. Pollett's autobiographical essay "The Ocean at My Door" (1947) and his character sketch Peter the Grate (1952), both published previously in separate volumes, were included in this collection, the whole being advertised not by its published title but rather as "The Ron Pollett Memorial Book," 17 Pollett's material was selected and arranged by his former editor, Ewart Young, who also oversaw the book's production. The 363-page book had three variant issues-- one, a white leather-bound "deluxe" issue, one cloth-cover "special" issue intended for use in schools, and one other brown paper-bound issue intended for use in libraries and homes; the latter two issues, as a result of popular demand, underwent second and third impressions in the course of less than one year. The sole distributor of The Ocean at My Door was Guardian Associates Ltd.; and in its magazine, the Atlantic Guardian, full page advertisements for the book appeared each month until late 1957, reminding readers that Pollett had been one of the magazine's most prolific and popular contributors. After Pollett's death, the editor was quick to assume the responsibility of promoting Pollett's name. Ewart Young wrote in an obituary on Pollett that "our sadness at this moment is somewhat tempered with a feeling of pride that we were able to have a part in the creation and preservation of the classics of Newfoundland literature that poured from his gifted pen." Is Indeed, the editor is owed credit for fulfilling Pollett's goal of completing a book, though this compilation was not quite what Pollet had had in mind, Nevertheless, the magazine itself, with its articulated intention of promoting all aspects of Newfoundland life, played no small part in shaping the materials Pollett did produce in the brief span of ten years, for it provided an outlet in which, in various forms, Pollett could depict the essential nature of his immigrant experience.

H

Pollett's first complision, The Ocean at Mr Boor and Other Newfoundland Outport Stories, comprises early examples of the three main literary forms, vignette, journalistic essay, and short story, in which he rendered his own experiences as an <u>émire</u>? Newfoundlander. Two principal themes permeate there six pieces, namely, the struggle and alienation of the urban immigrant and the remembered pleasures of outport inheritance. Characteristic of this collection is Pollett's conception of an idyllic quality of traditional outport life, a recreated, documented, or imagined, timeless quality, which in each of these six pieces is defined and endorsed.

"The Outport Millionaire" and "A Born Trouter" are Pollett's first two reminiscent vignettes; and in them he uses literary devices which, if limited, nevertheless convey his perception of the unbridgeable contrast between the delights of his outport youth and the struggle of his contemporary experience. In each vignette, Pollett reduces the expatriated Newfoundlander's urban lifestyle to stark, key concepts: "clockpunching," worry, frustration, fatigue, illness; and manages to emphasize his sense of loss through a device of reiterating whole sentences composed of such negative ideas in the opening and concluding paragraphs, like refrains of songs. 19 Use of this device simply underscores his principal thematic concern of the exile's daily struggle, and provides a visual frame surrounding his recollections based almost exclusively on the familiar, earthy but rewarding, pastimes of the "lucky" native outporter. The two key figures Pollett focuses on are the rooted, penniless, but content "millionaire," who was "born lucky," and himself as an enthusiastic, naive boy. The starker the contrast the lives of these figures provide with his present day routine, the more pleasing is the recreated memory. These are unsophisticated, simple pieces of literature, requiring no special analysis; yet as autobiography the contents are significant, in that they suggest how Pollett was beginning to embark on the process of "fixing" memories of his past.20

"Let's Look at the Squid" and "Where Christmas is Christmas" are examples of journalistic essays in this first collection, and are similar to at least one dozen other such articles Pollett was to write in his attempt to attune his memory well to that other world he had left behind. They are descriptive folklore collections about distinctive and, to Pollett's mind, pleasurable, outport customs. In "Let's Look at the Squid" be describes the community tradition of squid jigging and customary work techniques associated with this activity; and in "Where Christmas is Christmas" he provides a compilation of family holiday practices, including house decoration and food preparation, and as well describes traditional community customs such as jameying. As recorded

documents, whose implicit aim is to preserve, such detailed essays are crude by literary standards.²⁶ Yet he does provide in them some wry, moving aneedotes about living apart from the culture he records. In "Let's Look at the Squid" he notes the ireory of the outport immigrant sgaing into a fish store window, imbusing the squid "with tradition and romance," while some other more practical ethnic residents actually buy the fish for supper; and in "Where Christmas ig Christmas" the physical surroundings of outport revelers are compared noctalgically to a holiday greeting card.²⁶ In his later efforts to depict the outport setting, Pollett would often refer to windows, to pictures, and also to travelogue films and scenes—all of which suggest the modes in which he as an exile could receive or visualize information from home.²⁶ If searliste essays are, however, his least accomplished publications in terms of literary style and technique. Pollett's voice, for the most part, is that of a disengaged folklore collector, and, as is typical of authors attempting to record folklore materials rather than to develop them imaginatively. "Let's Look at the Squid," "Where Christmas ig Christmas" and other similar articles provide little more than a description of millien.

Two other pieces in this collection, "The Cat with the Yaller Face" and "The Ocean at My Door," perhaps provide a more sophisticated evocation of that traditional outport world; and Polett achieves this transformation by telling of that world exclusively through the eyes of an active, enthused boy. "The Cat with the Yaller Face" is the first example of the episodic short story formats which Pollett was to use in most of his fiction, this being a grare well suited to a boy's narrative voice and to the themes of learning and growing up. In this story, Pollett focuses on a nimple incident— a boy's loss of an orarry yellow cat— in order to suggest the particular pleasures and constraints that would be known to a typical outport boy in a traditional family home. In contrast to his earlier reminiscent vignettes, is his short stories the world outside the outport is never referred to, so that no strident authorial emphasis is present to disturb the narrative by making points with the reader. As well, he does not provide mere catalogues of aspects of traditional life, but manages in faction to create as minaginative world by using only such information as relates to the action of the stories. In "The Cat with the Yaller Face," for example, a list of weekly tradi-

tional fare is presented in terms of what the boy's finicky cat will and will not eat, and a boy's domestic tasks are described in terms of how this household pet plunges its owner into mishaps that interfere with the entire family's daily work routine. Pollett attempts here to tell the story in native dialect, a technique indicating his intention to describe a very particular place. But perhaps more accomplished is the pace of the story, which has been described by one critic as a "headlong rush"24-a feature which imparts a sense of immediacy and urgency appropriate to a boy's world. In this story, Pollett seems no mere praiser of his past, but a fledgling artist. But it is perhaps in the title piece. "The Ocean at My Door," that one finds the first sustained example of Pollett's distinctive voice. It is enthusiastic, musing, and engagingly anecdotal; by writing retrospectively of an earlier time, ostensibly "as seen through the eyes of a boy." he re-acquires a childlike outlook that sees nothing that meets his eye as trivial or ordinary. There is something childlike and wholehearted in most of his mature pieces. This is part of his unique quality. Here his keen interest in the tales his grandmother tells, and his admiration of his father's masculinity, strike us as in keeping with this boyish viewpoint, which is sustained throughout the sketch. Again we see Pollett the artist, revising, selecting, and recreating aspects of his own early outport life to evoke that idvllic quality he favoured.25

Pollett's second publication, <u>Peter the Grates An Outsort Character Study</u>, is in a quite different medium—the extended character sketch—and it is autobiographical in the sense that in it Pollett traces his own changing attitudes over a period of some fifty years toward an "outport millionaire" figure and toward a besign quality of outport life which the character, <u>Peters</u> represents. Its theme is the unrealized potential of a nonconformist in a traditional outport community, and the study's intended focus is on the eccentric, now deceased, "millionaire," an illiterate songmaker and storyteller. This choice of theme suggests that Pollett was moving toward challenging the nostalgic view of the outport that he had earlier attempted to recreate. Whereas the telling and singing of stories, lore, and songs are pastimes that in this work as in many of Pollett's other pieces he usually associates with camaraderic and "fun," he soow also describes how among most outporters the creative impulse is used only to sanction the customs

and mores of traditional life. Midway through the study, for example, numerous tail stories are used to chastise Peter, a backelor who rarely undertakes difficult labour, loves busting and fishing as pastines, and who has never travelled away from the harbour, as family men must to obtain supplementary wages. "Tender good Lord," an outporter back from seasonal work in the lumber-camps, says in Peter's hearing, "Waan't the rabbits in thousands down there in Glenwood this winter....Every time I swung an axe I chopped the head off one." Each of a group of men contributes, with slight variation, a story of pleutitude; as Peter gapes, the narrator indicates that the stories are a form of verbal punishment, for the men who had laboured now use them to take a "psychological whack" at Peter who they think has merely sat by a hot store all winter. "But the narrator here has been given an interpretive function; by drawing on his own wider experience, he attempts to posit a fuller story of Peter's life than members of this community, absorbed with the exigencies of daily life, can know. This deliberate questioning stance toward cultural constraints represents an important development in Pollett's rendering of outport life. For Pollett now seems to have an ability to evaluate "the wonderful world of the outport." If the outport is not a distance and to achieve, to some extent, the stance of an outsider, so necessary to the author of fiction.

Peter the Grate, Policit's most extensive work, is prehaps also his most ambitions attempt to do what he set out to accomplish as an author to present a celebratory, detailed potrrait of traditional outport life, and in the process to define his own experiences in and away from that particular millies. It can be argued that the main flaw in the piece is an uncertain narrative point of view. For although in each of the fourteen chapters he adheres firmly to a chronological enactment of episodes which shed light on Peter's character, there is always an intrusive "I"—Pollett himself constantly confirming, deaying, and assenting many perspectives on a figure who is, after all, a fetive one. Astobiographical intrusions is, perhaps, intimical to the artist's task of boring into the nature and character of other people. The narrator here also tiresomely pensits in maintaining that this is a "true" story of Peter, a figure whom he claims to have known best through "stories handed down to us." ²⁰ pervasive gossip referring to Peter's lackadaisical and irresponsible nature, and his own close observation. Again we feel the storyteller's heavy hand,

interrupting the narrative flow. Pollett's narrative "!" would always be too close to the real Ron Pollett to allow a full release of his narrative instincts.

Yet perhaps we can look at this lack of literary sophistication in another light. All of these seeming literary flaws suggest standard aspects of an oral storytelling technique. The work's reliance on the narrator's personal experience establishes the teller's veracity; and the reiteration of statements regarding the "truth" of the narrative, as well as the addition of frequent digressive incidents and cultural items supposedly known to the listeners, all help to suggest a familiar place and recognizable characters and thus create a credible narrative. 51 Aspects of Pollett's use of these techniques can be described as follows: in his earliest chapters, he chronicles rather indiscriminately his own earliest recollections of Peter, both by describing his habits and appearance as a boy could know an adult figure, and also by supplying much detail about Peter's family and the exact location and appearance of his house. He thus establishes his own narrative role and the nature of the central figure in the story who has a normal history and a connection with a specific place and people. What attracts young Pollett, in part, are Peter's extraordinary skills at hunting and fishing and the uncommon lives of his family, particularly of Peter's father, whose hair turned white and who became a perpetual recluse upon digging for treasure at Petty Harbour. Pollett notes with boyish wonder that Peter's father "was the only one left alive to tell the tale. Only, he never told it "32 Repetition of this particular incident suggests Pollett's fear that the "untold story" might be lost; indeed, telling Peter's untold story is the basic premise motivating the narrative. The artistic, essentially impoverished, bachelor is scorned by the community, he is ridiculed because he lives differently from his neighbours; that too Pollett needs to reiterate and thus establish a "truer" story of Peter. Thus he wishes, in common with many rural storytellers, to preserve the life history of a remarkable, though misunderstood local figure, and to vindicate him by telling his story properly. He can do this best by claiming his own special place as an observer, and restating it. When we examine Peter the Grate in this way, we are less likely to locate "flaws" than to see it as a piece of literature with strong connections to the oral tradition which Pollett knew in his childhood.

In Pollett's third major work. The Ocean at My Door, are collected basically in chronological order most of his published pieces from 1946 to 1955, the major exception being Peter the Grate, which is placed as the final selection. Of these, his essays on the city-among them "Summer Madness," "Memories of 'Didder Hill," and a three-part series entitled "Un in the Big City"-display an adult consciousness somewhat different from the perspective in his earliest pieces in The Ocean at My Door and Other Newfoundland Outport Stories, His themes of alienation and absence of reward are the same, but in these essays, written between 1950 and 1951, he poses as an objective chronicler, relying on records, statistics, interviews, and personal anecdote to explore the sense of place and belonging (of lack thereof) which he and other Newfoundland immigrants experience in the city. Generally Pollett evokes the city as a yast milieu of fleeting foreign images. Wall Street, for example, is merely a maze of office buildings, each one large enough to house entire populations of towns in Newfoundland.33 Of the hugely populated and various metropolis, he says that each city block is as crowded "as Grand Falls station at train time on Sunday night-except on weekends, when one can "not see enough people to haul up a punt."54 Pollett's apt analogies to home suggest how he and no doubt other outport Newfoundlanders in New York viewed the concrete city and adapted to its way of life. Not that the attractions of life for the outporter in "the wonder city" so ever wholly faded. But when, as in "Memories of 'Didder Hill." he explores the absence of community, and alludes to the immigrants' yet unfulfilled dreams, his tone seems tired and jaded. In "Up in the Big City" he says that no one finds gold in the streets of New York, "because there isn't any there."26 To emphasize the folly of his once held vision of novelty and wealth. Pollett frequently employs a device of using boldface headings which underline the contention he now holds, namely that the so-called vast differences between city and outport life could be reduced to purely monetary terms: "Fish 754 a Pound" and "Fishhooks \$55 Each" are examples of such headings in "Up in the Big City."37 Irony also pervades his new attitude to urban delights. An occasional splurge at a "swank" restaurant where dinners cost \$30 and higher, he notes, might include "lobster, brook trout, and partridge-not uncommon fodder in Newfoundland."38 This renewed sense of the disappointments of city life leads Pollett to reasess the hettic daily struggle simply to "make ends meet." In "Up in the Big City" he expresses his bleakest vision of the outcome of one's labour. "all the plodding city man can hope to leave to posterity are the sweatband off his hat and a pimple of dirt in some lonesome big graveyard." In these essays we find an outharbour vocabulary of belittlement, and homely, earthy description, applied (with sometimes striking effects, as in "pimple of dirt") to the metropolis. These touches, together with the ironic, weary tone alluded to earlier, are new developments in Pollett's writings, hinting at rich possibilities for his work if he had lived to produce a more extensive occurre.

The idea of return, or, retreat, permeates a large number of Pollett's descriptive essays. Whether his focus is on the city or the traditional village, his theme in them is the unique appreciation which an immigrant has for his outport inheritance. As Pollett had shown in his earliest pieces, the outport could be seen as a rich source for literature; and in "The Manly Sport of Handlining"--his first piece based on his visit home in 1947--he enforces that sentiment by suggesting how the outharbour's natural surroundings inspire a poetic vision. From his father's punt, he describes in detail the "verdant" forests he could see on the shore, then turns to the movement of waves, flowering "the headlands like white horses dashing along the shore;" at dusk, he notices "the changing moods of the evening sky as the sun glided into a gold-fringed cloudbank." 40 These naturalistic passages are charming, but there is little in them that is distinctive. Yet they help to underscore a timeless sense of belonging, as does Pollett's habit of evoking a series of visual maps with each scene he views when he comes home. Unlike an ordinary tourist, he knows that each local site has a history, whether it be his father's fishing berth, "Peter's Rock," or other specific landmarks he associates with boyhood experiences or ghostly legends. And he often shares the immediacy of his experiences as a returned immigrant by creating tall stories, a sense of traditional lore he evidently favoured. Of his first night in his boyhood home during his trip in 1951, he writes

The mice operated a complete sawmill in the bedroom upstairs. Laden with augers, gimlets, chisels, cross-cuts, and portable buzz-saws, they climbed the chimney casing and started work the minute my head hit the pillow...

The first night in the bedroom I did hear something [else]. It was a persistent pat-pat on the chapboards just outside the window. The sound was not the knocking kind the goats make bumping their horns against the lee side of a house where they often seek shelter from the wind and rain. In any case, I knew there were no goats in our well-fenced yard; we were on the windy side and the sound was too far up anyway.

As the taps grew louder, the floor-length flimsy white curtains spread half across the room in a sudden gust through the open window. The night had been stark calm just before the tapping began.

But, in a minute or two, the noise subsided and the curtains folded back like hands on a lap. The ghost had passed through the room! My wife must have heard and seen all of this, through she uttered not a word. It was not time for talk.

In the morning I looked out the window into the cheery sunlight and saw an apple tree alongside. It was loaded with fruit, and a big branch hugged the clapboards. I broke the branch off, killing the ghost for good. 4

Again we see the oral tradition influencing "formal" literature. In stories such as these he suggests how his identity and qualities have been shaped through his early cultural training. He is well-suited to become a trusty narrator in <u>Peter the Grate</u>.

In such late short stories as "The 'Dicky-Bird' Dish," "Ship Ahoy," "The Tongue that Never Told a Lie," and "Rum in the Puddin'" Pollett hints that the enduring quality of a distinctive rural community with its own set of practices and values will thrive despite contemporary change. 42 One exception contradicting this hopeful theme, however, can be found in an extended character sketch entitled "Johnny the Pear." It resembles Peter the Grate insofar as it too attempts a balanced and realistic depiction of traditional outport life. But its narrative technique differs strikingly; and, to turn to theme, the effacing quality of time, not, as in Peter the Grate, the single event of Confederation, is presented as the destroyer of outport culture. Traditional customs devoid of meaning or appreciation, fragments of oral art forms, the whole forgotten owing to disuse, are elements in the story denoting Pollett's view of the effects of time on practices in traditional life. In the story a youthful narrator merely spins an adventuresome yarn set in the "rum-running days of the 1920s," a yarn which culminates in his identification of a stranger, "Johnny the Pear," a government liquor board official, 43 The narrator tells his story, but only in passing is an earlier time, rich in lore, evoked. Two central events of the tale are a group of young outporters' "lucky" find of a case of contraband rum and their subsequent mischievous thievery of barrels to be used at Bonfire Night, an acknowledged custom of which "the significance meant no more to us than to a 8y on the wall." It be device of stories within stories, their unifying motifs being stealing, greed, and informing, underlines the parallels between the two events which the naive narrator does not fathom-he is, after all, too proccupied with the fast-paced, adventuresome events themselves. But it is the disintegration of custom rather than its enriching quality over time that is emphasized. As the narrative concludes, a new generation of children is somewhat irrelevantly singing one refrain from one of the many songs Johnny the Pear had unaght their fathers. These children, the new breed of carefree youths, essentially are uninterested in the significance of all the events told in the narrative. Johnny the Pear will not be one of the immortals.

In Peter the Grate Pollett's last major piece of writing, he succeeds in summing up his many poses as an immigrant stoyrteller and chroniciers—among them, the panegyrist of the distant past, the alienated urban spokeman for the values of the traditional outport community, and the exacting recorder of a passing way of life. Telling about the outport past, describing its true nature, fluding lessons in that time and place, all were objectives that Pollett had purued consistently in his writing. In this last extensive work the overriding idea seems to be that traditional outport life clearly had passed away. Yet the meaning even here is equivocal: there are traces of nostalgia; there is still hope, pitted against the bitter foretaste created by the develop or perish mentality of the new Newfoundland. Indied, his subsequent renderings of outport life resume rather than develop his same, well-used, and limited poses, which, as such, provided limited scope to develop his skills as an author and interpreter of the Newfoundland cultural milieu. At bottom, he remained a sentimentalist. A nostalgic remembrance of outport life in the distant past, informed by his experiences as an immigrant, remained characteristic of Follett's writing from beginning to end.

111

Despite this lack of development in his writing, both in terms of experimenting with literary form and technique and expanding his limited range of emphases, Pollett's works contributed to a timely trend participated in by a number of other native-born authors whose writing on the Newtoundland experience reflected two popular issues of the day-broadly, tradition and transition. The post-war years fostered a spirit of nationalism, when sostalgia in the face of imminent cultural change was an attitude shared by many, and Newfoundland's past, present, and future became the major topic of discussion, socially and in literature. A new wave of literature appeared in a spate of fledgling journals that developed during the mid-1940s. Among these were the Atlantic Gazardina, (solutited "the Magazine of Newfoundland"), and the more exclusively literary The Newfoundland Story. The Newfoundland Writer, and The Newfoundland Companion. Most significantly, in these journals ordinary Newfoundlanders now took to

expressing themselves in print; and in what had been essentially (though not exclusively) an oral culture, this material represented quite a new cultural development. Authors uses of traditional oral stories and their devices, motifs, and functions have not yet been the object of an extensive study; nor has any study suggested how this transition from oral to written culture might account for an awkward use of literary techniques among some new authors whose only previous experience had been school compositions and correspondence. Wevertheless, each of these scoplyres embarked on a new avocation or career with one sound principle in mind: each drew on what he knew, or remembered, or imagined, of Newfoundhald life, making use of the common body of cultural knowledge he knew best. And the <u>Atlantic Guardina</u> was for many the most available vehicle for their work, for its editors lavited expressions of popular opinion on the lishaders' legacy and their future prospects and gave roughly equal space to opposing views in each listue.

As a regular reader, Pollett could keep pace with the contemporary issues, but those writers in the <u>Atlantic Guardina</u> who shared with him a nostalgic view that eschewed change appear to have influenced him most as a storyteller of outport cultural values and traditions. One of these was Arthur R. Scammell, a Newfoundland-born schoolteacher living abrond and a popular, regular contributor who appears to have been something of a model in Pollett's career. Pollett's early autobiographical vignette, "A Born Trouter," in which he describes drollishly the makely of "trout fever" contracted by every Newfoundland boy, is ostensibly a lighter treatment of the

conflict rendered in Scammell's first short story, "Sea Fever," involving a boy's choosing to become a fisherman rather than a schoolteacher. 49 Eventually, Pollett's short stories, all set in the traditional outport milieu suggest (though do not develop) similar conflicts; mere homesickness for a day of trouting can, in Pollett, flower into a lengthy reworking of his decision to abandon the traditional ways of life. And in "Ship Ahoy!" young Ernie Tarn, whose father had drowned. ignores his mother's urging to prepare himself for the far less rigorous profession of schoolteaching when an uncle convincingly teaches him about the many rewards and challenges that a skilled fisherman enjoys. The link with Scammell is clear. But the traditional outport world of Scammell and Pollett is not always portraved sentimentally. Both could render its customs and values with irony and humour. In Scammell's short story "Buyer Must Be in Good Condition." when the erratic starting mechanism of an outboard motor demands more labour than the traditional sailboat, its owner decides to sell, and composes an advertisement from which the story takes its title. 50 There is similar laughter over modern machinery and devices in Pollett; both men watch with pleasure an old breed of men tinkering with the novelties of the twentieth century. Pollett admired Scammell's tolerant attitude to the new age, although his own brand of humour was usually more sardonic. He often used the traditionally popular tall story as a tool for contrasting values in a new era of transition and materialism. In several instances set in the distant past, a merchant who "went to church barely once a month to spare on collection and even then sang only one verse to save his breath," is presented as a local example of notorious greed. [3] Pollett's frequent descriptions of a new generation of outporters that ignores traditional chores and custom. and squanders large sums on novelties, make the traditional frugality of the ordinary outporter a comparative virtue, even when, for instance, he exaggerates his telling of how, when a fly fell into an old man's sugarbowl, "he picked the fly up and shook the sugar off its back into the bowl before letting it go."42 Pollett, one feels, had the makings of a satirist in him; not so Scammell, whose sketches of the outport are prevailingly gentle and lyrical.

But Pollett needed to define a distinctive traditional setting to enhance his treatment of acquisitiveness and frugality, misapplication and skillful prowess; and in drawing this, he appears

to have been influenced by at least two local noets, apart from Scammell. He admired the work of Eli Miles a Newfoundlander living in Boston and Rose Sullivan both of whose noems or rhyming sketches, appeared regularly in the Atlantic Guardian; and their eye for detail was a quality he thought enhanced their special ability to "catch the spirit of their locale," 58 Sullivan and Miles took for their subjects ordinary incidents, such as seasonal preparations for sealing and for work in the woods, an excursion on the "Newfie Bullet," and traditional Christmas celebrations. They were literary cataloguers, and Pollett was as well, to a degree. His own approach to the same milieu was perhaps more journalistic (or even pedestrian) than poetic. His vignettes and articles resemble the notes of an amateur local historian of any small community. Like the folklore collector P.K. Devine, he amassed information on custom and practice; and his articles and vignettes relate how a traditional Christmas table was set, provide an inventory of his family's weekly fare or give a detailed catalogue of local flora and their traditional medicinal and dietary uses. While such exercises were doubtless important in reminding him of materials he would draw on and interpret in his stories, what their significance was for readers of the Atlantic Guardian is hard to determine. 54 Presumably, they functioned to reawaken an enthusiasm for ways of life that for some were no longer practised. This was a part of Pollett's contribution to what was an essentially nationalistic, popular literature of the period, seeking to define the islanders' distinctive character and traditions.

Yet Pollett also sought to direct his collective audience's eye, not only toward the remembered past but also away from present day commercialized change and noveity. He was a sort of homely moralist. In his fiction using a boy as the protagonist, and in his autobiographical accounts featuring himself as an outport boy busily occupied with learning important out-of-door skills and community values, Pollett appears to have drawn on a late Victorian diductic tradition of popular writing for and about children. Authors of this fiction sought to impart "wholesome," moral and practical lessons; indeed it was from this wealth of juvenile literature that the <u>Royal</u> <u>Readers</u>, Pollett's and his audience's main early school texts, were compiled: ⁴⁰ One full-length novel in particular— <u>Smith and his Friends</u>, only one of several hundred novels by Evelyn Everett-Green- appears to have had some influence on how Pollett set about depicting his own outport boybood although it was written in the author's well-used formula- a combination of adventurous exploration nature study and historical wonders to South's holiday to the Swiss Alps opens a door of beauty, intricately detailed by the author, and of folk belief, among these fables of "ice maidens" who lure unwary travellers to mountain precipices. These are relayed to Squib by his friend Herr Adler an erudite and pious man whose fables and anecdotes always have a message for the credulous youth: in them he is urged to excel, to sympathize with the weak, and, most importantly, to appreciate the heauties in nature and his fellow men. An outport world, superficially commonplace in contrast to this foreign setting, could be made to seem extraordinary by adopting just such a boy's limited perspective. Pollett's reading of a second author appears to have encouraged him in the idea that he too had an important story to impart. His enthusiasm for the autobiography of American author Alice Marble may be explained in part by her ability to turn into literature ordinary, rural family life much like his own. Like Marble, he too wished to evoke a setting where "Days came and went in an unchanging pattern," but yet contained a great deal to explore and learn from. Seeing the uncommon in the seemingly commonplace- the hidden stairway in his house, the delicate mica stove front that somehow does not burn, the incongruous pattern of the wallpaper in his room- was to be a source of inspiration for Pollett, as it had been for Marble. Reviving the perhaps rather outdated formula conveyed in sources such as his Royal Readers, the works of Everett-Green, and Marble had the effect of giving Pollett's work an old-fashioned, musty look. It was also an approach that was perfectly in keeping with his own and his audience's first experience of literature-stories that taught one to search out a lesson in both the commonplace and the foreign.

While Pollets awere completed his project of telling about the rich output world "as seen through the eyes of a boy," seither did he wholly abandon it, his short fiction being interspersed with accounts of his contemporary urban filletyle; in effect, the local issues of tradition and transitions were transferred to the urban milies. In his recurring focus on materialism and the abbesser of a sense of community, his view of city life has elements of an American tradition of urban

literature. One literary historian, surveying the works of Hawthorne, Melville, Fitzgerald, and other classic American authors, has concluded that "the city" in literature most often is depicted as an "abstract receptacle for displaced feelings about other things"-namely, ambivalent feelings about "the transformation of society and of culture of which the emerging city is but one manifestation "58 Such an attitude often fosters an attempt to imbue the stark reality of the city either with pastoralism or some other alternative vision that clearly is not "of" the city. Pollett's reading of the autobiographical novels of Thomas Wolfe would have given him an illustration of this kind of phenomenon in writing about urban America. Undoubtedly Wolfe had an influence on Pollett. In Of Time and the River, for instance, Wolfe presents an extensive sensuous catalogue of commodities available at the Faneuil Hall food markets in Boston, and tells how the lonely Eugene Gant peers at stores and warehouses in the evening, perceiving the place as only a southerner could understand it.59 The window of memory through which only an immigrant could see is an important image in Pollett's first book. In both "A Born Trouter" and "Let's Look at the Squid," immigrants are depicted as sombrely gazing into actual city windows, in the first vignette with noses flattened against a restaurant window watching a "gorgeous show" of trout, and in the second musing over squid in a fish store window, thinking of home despite the variety of attractions in New York, and endowing the "lowly squid with tradition and romance," 60 As in Wolfe, the city provided Pollett with a wealth of images. Pollett's inability to find beauty or comfort in the city expresses itself in a hostility to the metropolis that is perhaps characteristic of much of American urban literature, though he finds a vocabulary to describe it that is "of" his former seaside home. His frequent ironical natural images allude to outport life: the harried "subwayite" hurrying down stairs toward the dank underground is compared to a diving builbird; and indistinguishable, locked doors in huge apartment complexes are "like periwinkles on a wharfstick." But unlike more accomplished and prolific American authors, Pollett appears to have relatively little of real substance to say about the impact of the urban experience upon himself or other Newfoundlanders. He produced no sustained attempt describing the years between the initial enthusiasm that led them to New York and their decline in middle age. The full story remains

untold.

A cultural community uprooted by circumstance was an idea which certainly had notential for a novel. It was one which Pollett came upon frequently in his numerous re-readings of two particular American novels, John Steinbeck's The Grapes of Wrath and Martha Mitchell's Gone with the Wind. While these authors' influences on Pollett's works appear limited, his reading of them might well have encouraged him to embark on a study of the effects of time and urban experience on a transplanted people and a community. He comes closest to beginning to render this idea in literature in "Memories of 'Didder Hill'," where he describes the unique heritage shared by Newfoundlanders living abroad and denotes their special character and place in the city- their distinctive dress, speech, skills, attitudes- and even the recording of a local place name legend of "Didder Hill," the men of Little Newfoundland's favoured meeting place, all would seem to enrich a segment of the urban milieu. 62 Nevertheless, after initially telling of this community's bold vision of the promise of America, Pollett resisted telling each man's story, and resorted instead to the pose of objective chronicler. The latter half of the article is a mere compilation of names, occupations, and activities, mere reportage that fails to render "the densely woven fabric of all their histories," such as Wolfe attempted, 53 In narrative technique, he would appear to have been influenced by the American journalist and critic Edmund Wilson, who, in Memoirs of Hecate County, creates an inscrutable narrator who is always commenting distantly on the folly or the accomplishments of a group of thinly-drawn characters who share a common bond merely because they live in a distinct place- Hecate County- but have little in common beyond this choice of residence. 64 The narrator never personally raises questions about their lives or actions; and therefore what one cannot help but gather from this book (or group of looselydrawn stories) is merely that over time, things, sometimes extraordinary but more often ordinary things, just happen. Pollett, too, in his modest attempt to universalize the Newfoundland immigrant experience in America, fails to find an appropriate literary form suitable for the making of fiction. Although he was partially influenced by American literary sources in the rendering of his topic, he can only be said to have a very minor place in American literature, one likened to writers of ephemera or colorful local sketches whose works find a voice in any daily newspaper.

He turned to the outport of his imagination for his most sustained work. It is doubtful that Pollett was reading the Canadian writing of his day, yet his concerns in Peter the Grate parallel those of writers of Atlantic Canada; Ernest Buckler's The Mountain and the Valley was published in the same year (1952), following Hugh MacLennan's Each Man's Son (1951), and preceding slightly Charles Bruce's The Channel Shore (1954). All have in common the examination of seemingly unchanging rural communities, and focus on inarticulate artistic or ambitious figures set against a parochial, traditional background. In David Canaan's house in The Mountain and the Valley, his grandmother weaves used clothing once worn by all members of the Canaan family, each piece imbued with intimate associations, into a rug; this rug is a symbol of unity, emphasizing the difficulty of breaking away from custom--a breaking away which David longs for, for he sees the traditional exclical and seasonal form routines not as tiring but tiresome 65 As in The Channel Shore, this frustration is not vented; Hazel McKee, in Bruce's novel, wants to pursue a career as a singer and knows "it lisn't practical. Music has no place on the Shore except in church concerts."66 Individual inherited obligation is as much a problem as individual choice; to diverge is a threat in unified, traditional communities where members depend on one another. Pollett's resolution to these sorts of problems is no more satisfying than that suggested by Maritime novelists of his day-namely, death, sometimes untimely; emigration; or, as MacLennan unconvincingly manages for his central character. Alan MacNeil, adoption into a wealthy, learned family. The factor of a similar cultural history, which perhaps alone distinguishes the literature of the Atlantic region, suggests a Canadian context in which many other of Pollett's works can be assessed and appreciated.

Peter the Grate stands as Pollett's last major work, and in writing it be appears to have asked: despite change, what did remain essential in outport life? He had no answers. While in this work Pollett progressed somewhat from his earlier characteristic culling and rendering of pleasing boyhood memories, in most of his subsequent publications he took his own early works for his model. Important messages he originally sought to impart about the value of traditional life mixed with ordinary middle aged nostalgia as he catalogued lost outport traditions and customs; and by the end of his career he would seem to have progressed in his art not at all. One result of this uniformity was that his own works, being perceived as "merely setting down his own tender reminiscences." for inspired imitations of dubious literary merit. For instance, in 1957 Theresa Collier published a sentimental imitative vignette entitled "One of the Millionaires." which begins. "The day I hired him to saw firewood for me I thought to myself, 'This is one of the outport millionaires so fascinatingly described by Ron Pollett in his interesting book The Ocean at My Door, Pollett's notions about the unequal balance of the quality of life in urban and rural worlds, once typified in the figures of "clockpuncher," and "millionaire," but which he neglected to develop himself, now were ignored, Collier's focus being on a mere derelict worthy somehow of praise in flowery and stilted prose. In Pollett's wake, the urban experience also became a focal point for some local authors, and an article appearing in 1955 on Maritimers living in Boston resembles Pollett's writing on city life. Entitled "Two Maritime Advocates." this item too is flawed by a similar reticence 40 The fact that this group occasionally socialized together elsewhere appears as the main motivation for writing the piece, but it has very little to say about their experiences. Pollett's few followers evince an inability to distinguish between his works' merits and flaws, and their writing remained in the realm of the provincial.

In returning to the idyllic outport of his imagination toward the end of his career, Pollett sought to preserve a pastoral vision that increasingly was being eschewed by his contemporaries. Indeed the issues of tradition and transition were now being handled quite differently by some new writers. In one characteristic article appearing in 1948, entitled "You Cast' to Home Again," an author in a few key sentences reduced to backwardness and doubtful accomplishment the history of outport traditions over hundreds of years. "In Still, many other writers of the day created "typical" outports and made them colorful by locating in them drama between Hollywood-style gangiters and stereotypical country bumpkins, the latter's unaccoloral language and quaint thought being likened to dolithhees." In Pollett's renewed if simplified effort to record in his final articles and stories a richly textured past was, in contrast, a more pleasing vision, and

inspired one reader to dedicate a poem to him in 1954, when he was publishing his last pieces. It was entitled "Come Back Ron Come Back," and it urged him to "tell us all about your Newfoundland....Ere the happy mem'ries fade-"2". The parase "your Newfoundland," whether intentionally or otherwise, nicely suggested that Pollett's outport was an artificial construct, and pointed up the ironic contrast between his delight in the now idyllic, static outport of his imagination and his own youthful impatient longing to leave his real outport bome to explore the world outside. It was perhaps for other, later Newfoundland writers to explore the cultural changes in the world they knew, and provide a more complete and perhaps more balanced version of outport life as they perceived it.

٠..

In his writing Pollett attempted to preserve a record of the outport experience as he saw it. The collapse of Guardian Associates Ltd. by 1957, however, was an event that seemed destined to send Pollett's name and work to oblivion. Yet there was a small attempt to ensure his fame.. In 1956 Pollett's publisher praised The Ocean at My Door as "worthy of a place in any home and school;" and evidence suggests that it was sometimes used in the schools as a supplement to world history texts and literary classics. 78 As native source material, it could provide accurate detail for study of cultural history. In focussing on the everyday past, Pollett researched neither the mainstream political nor cultural events of the first half of the twentieth century in Newfoundland, but rather suggested how great events affected the lives of ordinary outporters, from the excitement of seeing the first trains of the Bond and Morris years to the enthusiasm of Confederation brought about by "Joey" - "The Man with the Bow Tie." His work is an accurate record of the common man's history; and his particular scrupulousness in providing detail presents something of a treasure-trove for the modern day folklorist. The book was never added formally to the provincial school syllabus, however, an omission perhaps indicating that local administrators and teachers were unwilling or unable to locate a place for it in the traditional courses of study, or that the world evoked in it was too close in living memory to be of interest to students. It is thought that

Pollett's publications were nevertheless partly responsible for the formal recognition the <u>Atlantic</u>
<u>Guardina</u> received in 1955 for preserving the region's "cultural assets," this farther afield, on the
mainland of Canada. In Newfoundland, as in the rest of Canada, pedagogical reaction toward
undertaking formal study of local art and culture was, at best, conservative.

Pollett was, primarily, a chronicler, though he could also be an engaging storyteller. Like some other writers of his day, he succeeded in capturing in literature a sense of the traditional outport way of life in its various manifestations, such as its religious tensions, its dialects, and its distinctive exaggerated mode of humour. In his popular short story, "Rum in the Puddin" for example. Uncle Bill jiggles his wife's elbow, tipping the contents of a rum bottle into a Christmas nudding meant to be served to their dinner quests, four Salvation Army soldiers. Bill knows that his guests think of the liquor as "the devil's brew." but tells his wife. "Don't spare the horses. Martha, me maid... Christmas only comes once a year."76 She later pulls off the pudding's calico cooking wrapper, and watches the raisins in "the rum-soaked enigma" stare out at her "like a hundred accusing eyes."77 Such small events in the lives of ordinary outporters were frequently the ones that engaged Pollett's attention; but unlike his contemporaries such as Arthur R. Scammell and Ted Russell. Pollett more often does not simply seize upon such an event as the main focus of his stories or articles. In Scammell's fiction, the ordinary events, such as a fisherman's son's act of manfully and charitably parting with his first hard-earned dollar, are similar to the sentimental themes of much of Pollett's juvenile fiction; 78 what is absent in Scammell's stories is a vivid rendering of the particular quality that distinguishes the outport from communities elsewhere. Pollett's ability to evoke a distinctive place, in contrast, is achieved by his particular care in providing naturalistic and sensuous detail: it was his attention to the physical surroundings, no less than to folk customs and lore, that provoked one enthusiastic critic to comment that "even if you were born in the city you can visualize and appreciate the outport, even if you have never before seen an outport."79 Pollett would, of course, have agreed with most of Scammell's stated opinions about the outport, perhaps even with this rather smug statement in the essay "Outport Heritage": "Newfoundlanders in their little communities have built up something worthwhile,"

Scammell writes, adding that "hundreds of little communities, far removed from the rush and bustle of city life, are rich in true social and spiritual values,"80 Pollett in his writing attempts to make a similar affirmation, but in contrast to Scammell, his way is to linger over details and lovingly explain all that meets his eye rather than issue theses. Within his vignettes and essays there is much anecdotal material that has the potential for weightier literature; we may note, for instance, his allusion to the fisherman who could be as tight-fisted and secretive at sea as he could be generous and companionable on land and who also had inherited the talents of a "seventh son,"81 Nor is this the only character who could do with longer treatment. But there are few eccentrics or bold "characters" in Pollett's community; and in this he is unlike his contemporary. Ted Russell, who began writing seriously and broadcasting his work some years after Pollett and Scammell had already published the bulk of their work in the Atlantic Guardian. Russell's attitude towards the outport is similar to Scammell's as expressed in "Outport Heritage," but his work evades sentimentality through his use of such idiosyncratic "characters" as the members of the Noddy family and its equally eccentric goat, or the shrewd and philosophical narrator of the chronicles. Uncle Mose. 82 These figures and their antics are grounded in a specific place, and indeed they are remarkable; so much so that Russell need not explain the appearance of this "typical" outport in which they live, or the routines which characterize their lives. It is the individual "character," not the place, that matters in Russell. Pollett, however, in his amassing of detail in fictional and nonfictional prose, undertook to describe the distinctive physical layout and cultural history of the outport, and to tell of the aspects of his people's ordinary lives. In Pollett, the people collectively are the center of interest, how they work, where they live,

Any final assessment of Pollett's contribution to local literature must take into account the literary milieu and critical climate of his day. His works appeared when no publishing house existed on the island, and literary culture, such as it was, was confined to a small group of creative writers and journalists living in St. John's. Their criteria for judging literature was runimentary; they examined authors' use of native culture in literature and proceeded then to bandy about notions of 'authenticity.' Questions of literary style and form took second place to examine ing bald content. Indeed, one reviewer of Pollett's posthumous compilation The Ocean at My Door noted that its content was little more than "simply reminiscent and descriptive," yet at length emphasized the work's intimate recapturing of the local flavour of bygone days; "Here is the taste of figged loaf, excursion biscuits, barksail bread, *84 This list is more reminiscent of Pollett's own predilection for cataloguing than of the techniques of literary criticism. Still, the critic saw something of Pollett's quality, namely his engaging narrative voice, and his ability to evoke a richly textured past. The intimate and sensuous treatment of his topic, making his reader "see one and every outport," was also praised by the novelist Margaret Duley, who hailed Pollett's publications as "Newfoundland gems."85 Duley, herself an author of serious fiction in which the outport way of life was central, albeit portrayed as sadly insular and parochial, first called for the collection of his pieces in a book "so that his portrayal of the blood and bones of Old Newfoundland will not be lost to posterity." She also found in Pollett's works a vividness in depicting outport life that could aid one in the understanding of the poetry written by Newfoundland-born E.J. Pratt;87 and as such her comment represents one of the very few hints to be found in contemporary criticism that an authentic literature of Newfoundland indeed existed prior to the literary stirrings of the mid-1940s.

The Newfoundhard resuling public in the 1900 and 1900 was perhaps larger than is supposed, the most popular item being fiction for adults. In Pollett's works (and of course in Scammell's) the country's readers were possibly seeing for the first time in prist a focus upon the everyday life they knew, and reactions to his publications ranged from the reader who "eagerly looked forward to each one," to another who declared: "he wrote of the outport with such feeling that it are into one's soul." In his enthuisant to record all he could remember, Pollett often resorted to the images of "panorama," "picture," and "scene," at times enjoining his readers to personalize these with details from their own wealth of recollection. Indeed it is difficult to recapture the enthusian ordinary resident, living during a time of war and upbeaval, must have feld in discovering a literary recreation of their own world. They responded to his voice; and Pollett himself once said that in his desire to reach the broadest range of readers he deliberately tried to

keep his narratives "simple and spiced with hyperbole."90 Clearly, he did not have exaggerated notions of his own talent; he would refer to his writing as "a talent for stringing words together."91 On occasion he indulged in self-mockery, such as when he described poetically the nymph-like splashings of his wife at the "sylvan retreat" of a local waterfall, only to be told by her, "it's about the best place ever for people who don't have a bathtub at home." Pollett responded, typically, "And so it goes. I breathe rhetoric and hyperbole; I create immortal lines at the snap of a finger- and my wife reduces it to soap!" The artlessness of most of Pollett's prose involved his readers not only in his own personal story, but in his view of the timeless, quality of the outport experience. His was a subjective, nostalgic view, although in depicting it and winning for it considerable acclaim, the importance of Pollett's specific audience of the day cannot be underestimated. If in writing of the city he thought of his readers as ordinary folk "picking out the words by lamplight in some far-off fishing cove," then he knew that he had to exercise the greatest caution when recording aspects of their familiar milieu. Published letters from both readers and writers in the Atlantic Guardian suggest the scrutiny given publications in each issue by readers. This occurrence alone made the journal a lively forum of debate, allowing anyone to have a say in sanctioning or deriding writers' handling of Newfoundland topics. Realism, above all, was demanded. And Pollett supplied it. When memory failed, he asked older outporters to supply the forgotten detail; and perhaps in this context of gauging authenticity, his greatest success in rendering outport life can be supposed from the response of one reader who was questioned about whether she read Pollett's writings- she said, "I lived them." Beyond the sentimentalism of his works. Pollett had recaptured a good part of the cultural fabric that had shaped his own past. It also seemed to have been that part which his critics and readers wanted to recall as their own.

But Pollett also wrote of another, distinctly different place and time, and, is the years since his demine during which there have been no dearth of stories and accounts immortalizing the traditional outport, his careful focus on the <u>éminte</u> experience is perhaps Pollett's most significant contribution to the island's literature. This experience, though little studied, is also an important part of Newfoundland society and culture. A small number of islanders have produced short, descriptive items to describe the experience of the city, but these pieces fail to bore into the nature of daily life known to the expatriated Newfoundlander; indeed, most are written by mere Newfoundland tourists. Autobiographical tales of youthful adventure-seeking in Manhattan and Brooklyn such as those briefly snun by Violet Randell in Thursday's Child, and by Joseph R. Smallwood in I Chose Canada, are presented as the mere sojourns that they were, and end with the sure knowledge of return to the familiar homeland. 95 The Rev. Philip Tocque, whose preoccupation with the details of monuments and other city structures resembles Pollett's city essays. was also a visitor, not a dweller, in Manhattan, and his "Incidents of a Visit to New York," gives one no sense that there are inhabitable sites in the city 26 In contrast Pollett's depictions of the city provide a truer portrait: he provides as much statistical data as any interested tourist, but goes further in attempting to describe the daily interaction between Newfoundlanders and other dwellers in "the big city." Adaptation to urban lifeways was demanded of them, and he documents the attempt. That the possibility of a return to the truly familiar way of life within their new milieu was inconceivable, he knew, "Land enough to build a wharf, stage, flake, house and cabbase sarden-about an acre-would run to half a million dollars on the New York waterfront." Pollett wrote."47 His usual emphases on struggle, alienation from home, and unrealized dreams, do not make for a pleasant picture; however, his work is the single personal account of the uprooted Newfoundlander that exists in literature. As such, it has been said "that part of Pollett's value as a writer is that he speaks for a mute multitude of emigrant Newfoundlanders."98 His writing reflects the émigré experience in its many aspects; the unrewarding experience of the city which in turn fostered his creation of the outport idyll; and the loneliness in the absence of a familiar community in the major centre in which he felt both attracted and alienated. Writing for him was cathartic: it was also a way of responding to his desire to bridge the distances of time and space that separated him from his native land. As a Newfoundland author in a foreign land, he dwelt on his cultural heritage, and exercised his literary talents, in an effort, perhaps, to feel more at home. In so doing, he expressed a part of the cultural history of his people.

ABBREVIATIONS

A list of	abbreviations	need in	Notes	and	Ribliography:

ACHL- A.C. Hunter Library (Newfoundland Section), St. John's

- CNS- Centre for Newfoundland Studies, Queen Elizabeth II Library, Memorial University of Newfoundland
- MHG- Maritime History Group, Henrietta Harvey Library, Memorial University of Newfoundland
- MUN- Memorial University of Newfoundland
- MUNFLA- Memorial University of Newfoundland Folklore and Language Archive
- PANL- Provincial Archive of Newfoundland and Labrador
- SPG- Society for the Propagation of the Gospel

NOTES

Preface

Yasmuel Johnson, <u>The Rambler</u>, no.60, Saturday, 13 Oct.1750; rpt. in <u>The Yale Edition of the Works of Samuel Johnson</u>, vol. 3, Eds. W.J Bate and Albrecht B. Strauss (New Haven: Yale Univ. Press, 1969), p.322.

² Oral history as a research technique is increasingly being accepted as an invaluable tool for historical and biopaphical studies, particularly in the study of subjects where written records as one-existent. One staunch advocate of this technique is Edward D. Ives, and research methods outsiden in his book, The Tange-Econocide Interrise (Knowlike, Tenn. Livis, "of Tennessee Press, 1980), have been used for the interviews undertaken for this study, See David J. Mitchell, "Living Decements: Of all linety and Bengarby's. In Biogramia," o. d. J Fell 1989, pp. 26-6, which procures to a continuous proposal control of the procure of the procure

I. New Harbour Boyhood

- ¹ Garry J. Cranford, with Raymond Hillier. <u>Potheads and Drumhoops: A Folk- History of New Harbour, Trinity Ray</u> (St. John's Harry Caff, 1983), p.7. I am indebted to Mr. Cranford for loaning me the lengthier, unedited manuscript of this book, much of which documents eighteenthand early nineteenth- century New Harbour life.
 - ² Op. cit., p.8. The original source is the Trinity Census, 1800-1801.
- ³ Op. cit., p.9. See also the research on the early history of New Harbour and the Newhook family in the Nimshi Crewe Collection, P4/13, Box 9, Files 227 and 228, PANL.
- 4 Newfoundland General Assembly. <u>Abstract Census and Return of the Population of Newfoundland and Labrador</u> (St. John's: Ryan and Withers, 1845) (hereafter cited as <u>Abstract Census</u>).
- ⁶ Cranford, <u>Potheads</u>, p.10. A more substantial treatment of the economically depressed 1830s in New Harbour is provided in an unpublished essay by Cranford entitled "The Dirty Thirties," in his private collection.
- ⁹ I am indebted for these references to Garry J. Cranford, from material supplied by Monsignor Raymond Labey, whose research at the S.P.G. Archire, Loadon, England, covered a broader range of materials than is available in 'Newfoundland. See also the aumerous pleas for meal 'for the relief of the poor in this Place,' in the correspondence of New Harbour merchants and officials of the period to Magistriae Robert Planest, Harbour Grace, in Harbour Grace,

Newfoundland, Magistrate's Office, Cull-5, File 19, CNS Archives, MUN. Other strata of local society apparently were affected: the circuit praceder straing New Harbour, fave, W. Billock, for example, wrote to Archdesson Edward Wix on Nov. 13,1833, of the "miscrable destitution to which the clumb in this shades of newford." Of this prathoness, who were expected to provide a which the clumb in this shades of newford. Of this prathoness, who were expected to provide a three past years, I cannot but satisfyade a full measure of suffering for my own large and helpless famility's fees Fig. ("O" Mag. 20, 14/20, A1022, p.6, Fayl.).

- ⁷ See James C. Faris, <u>Cat Harbour</u>: <u>A Newfoundland Fishing Settlement</u>, Newfoundland Social and Economic Studies, no.3 (St. John's; MUN, 1972),pp.160-1, for a discussion of Christmas "times." For a description of "times" similar to that obtained from informants in New Harbour, see Aubrey M. Tizzard. On Sloping Ground: Reminiscences of Outport Life in Notre Dame Bay. Newfoundland (St. John's: Breakwater, 1984), pp.124-8. The English traditional custom of "ianneying" or "mumming" in its characteristic Newfoundland patterns and ritual activities is described in Herbert Halpert, "A Typology of Mumming," in Herbert Halpert and G.M. Story, Eds., Christmas Mumming in Newfoundland: Essays in Anthropology, Folklore, and History (Toronto: Univ. of Toronto Press, 1969), pp.34-61. Ron Pollett records various local Christmas customs practised in New Harbour, in "Where Christmas is Christmas," Atlantic Guardian Dec. 1946; rpt, in The Ocean at My Door and Other Newfoundland Stories (St. John's: Guardian Associates Ltd., 1956), pp.38-44 [hereafter, cited as Ocean (1956)]. These included annual parades by local members of the Loval Orangemen's Association and Society of United Fishermen, community carolling and church teas for children. See also a student essay by C. George Brown, "Traditions which Exist in the Woodman Area of New Harbour, Trinity Bay in General and the Repertoire of Aubrey John Woodman in Particular," Ms.72-94, pp.14-22, MUNFLA.
- ⁸ S.P.G., <u>Report of the Year 1830</u> (London: C.J.G. and F. Rivington, 1831), p.184 (Hereafter as SPG. <u>Report)</u>; and Government of Newfoundland, <u>Journal of the General Assembly of Newfoundland, Jay (38)</u> (St. Johns: W.R. Shea, 1845), p.185, in which it is noted that £15 worth of materials to finish the schoolhouse's construction still were needed.
- ⁹ Abstract Census, 1869 (St. John's: R. Winton, 1870), passim; and SPG, Reports, 1830-60, passim.

<u>Harbot Grace Standard</u>, 25 Sept. 1880, p.2, in which it is stated: "The only thing one desiderated was a larger attendance of parents and friends, who in other place generally muster in good force on such festal occasions as these. The people in these parts are singular somewhat in this respect: for some reason or other they have little interest in what goes on around them—except it be to try to catch fish, or jis quids."

- ¹¹ Lloyd George, in "Some Heritage Values of the Area [from New Harbour to Blaketown] Served by the Lions Club," lecture, Lions Club 11th Annual Charter Night, New Harbour, Trinity Bay, 28 March 1981, provides a survey of local constables and their usual duties from 1823 onwards; the date and stipulations of compulsory education are noted in Frederick W. Rowe, The Development of Education in NewOodmodland (Toronto-Ferron, 1984), a Development of Education in NewOodmodland (Toronto-Ferron, 1994), p. 18
- ¹² Anglican Church of Newfoundland, "Anglican Ministers' Registry," Queen's College Archive, St. John's. The evangelical nature of tasks accorded to local missionaries is outlined in annual reports of the SPG. Elford, the first schoolteacher in New Harbour, is referred to in these reports alternately as "schoolteacher" and "catechist."

- ¹⁸ Information obtained from Hiran Silk, personal interview, Grand Falls, Nide, 26 May 1983, and from Gyr R, Brassad, "Ferr, Arther C. Wagshore (1851-1900)", Canotina Botanical Anton, Billetina, "See evident botanics, and Anton Trans. Brasses-1900, "Canotina Botanical Anton Silk, "See evident botanics, and this," "See collected perhaps close to 10,000 specimes in Newfoundhad and Labrador, which eventually made their way to most major herbaris in Europe and York America." Waghore's spublications infined "Christmas Customs in Newfoundhad". Journal of America Bolkore, vol. 6 (1983), 683; and "Death Sigas and Westher Sigas from Newfoundhad and Labrador," (Jerman) of American Selficer, vol. 18 (1990), pp. 277-9.
- ¹⁴ Much of this information on the year 1901 is from the <u>Census and Return of the Population of NewYoundland and Labrador, 1901 (St. John's: J.C. Withers, 1903). (annual complete censuses, hereafter cited as <u>Census</u> 1.</u>
- ¹⁶ Ros Pollett, "The Ocean at My Door," in <u>The Ocean at My Door and Other Newfoundland Outport Stories</u> (Montreal: Guardian Associates Ltd., 1947), p.6 [hereafter this volume referred to as <u>Outport Stories</u> (1947)].
- ¹⁸ Information about the Newhook and Pollett families obtained from Crewe Collection, PANL, and Cranford, "Potheads and Drumhoops."
 - ¹⁷ Pollett, "The Ocean at My Door," <u>Outport Stories</u> (1947), p.7.
- ¹⁸ Information about Jim Pollett obtained from Church Registry, Church of St. Augustine, New Harbour, and personal interviews with Ralph Higdon, New Harbour, 1 June 1983, Elsie Thorne, New Harbour, 30 May 1983, Jim Thorne, Bishop's Falls, 27 May 1983, Reg Woodman, New Harbour, 30 May 1983, and other informants.
 - 19 Church Registry, Church of St. Augustine, New Harbour.
- ²⁰ Information about Norah Pollett obtained from Jesse Fogwill, St. John's, telephone interview, 22 June 1983, W. Bramwell Pollett, New Harbour, personal interview, 30 May 1983, and other informants.
- ²¹ Information about the Pollett family obtained from Church Registry, Church of St. Augustine, New Harbour.
 - ²² Pollett, "The Ocean at My Door," <u>Outport Stories</u> (1947), p.8.
- ²⁵ Pollett, "There's No Place Like an Outport," <u>Atlantic Guardian</u>, July, 1950; rpt. in <u>Ocean</u> (1956), p.135.
- ²⁴ See Joseph R. Smallwood, <u>I Chose Canada</u> (Toronto: Macmillan, 1973), pp.377-98, for descriptions of wages and living conditions in the lumbercamps.
 - 25 Pollett, "The Ocean at My Door," Outport Stories (1947), p.9.
- ²⁶ W.J. Lundrigan, Corner Brook, letter to author, 19 July 1983, and Mildred Johnson, Etobicoke, Ontario, letter to author, 17 July 1983.

- ²⁷ Jesse Fogwill, telephone interviews, 19 and 22 June 1983, Mildred Johnson, letter to author, 10 Jan. 1984, W. Bramwell Pollett and Margaret Pollett, personal interview, 9 Sept. 1982.
 - ²⁸ W.J. Lundrigan, letter to author, 19 July 1983.
 - ²⁹ Pollett, "The Ocean at My Door," <u>Outport Stories</u> (1947), p.12.
 - 30 Pollett, "There's No Place Like an Outport," Ocean (1956), p.132
 - 31 Op. cit., p.143.
- ⁸² Pollett describes these tasks in various essays. See, for example, "The Ocean at My Door," <u>Outport Stories</u> (1947), p.10; "The Garden on the Point," <u>Atlantic Guardian</u>, Oct. 1952; ppt. in <u>Ocean (1956)</u>, pp.283-.
 - ⁸⁸ Pollett, "There's No Place Like an Outport," <u>Ocean</u> (1956), p.128
- ³⁴ Pollett, "A Born Trouter," <u>Atlantic Guardian</u> Aug. 1946; rpt. in <u>Outport Stories</u> (1947), p.24.
 - ³⁵ Jesse Fogwill, telephone interview, 22 June 1983.
 - 36 W. Bramwell Pollett, personal interview, 30 May 1983.
- ²⁷ New Harbour Bridge, located between the two central areas of residence in the village, was a traditional medicap place for all villagers. It was a focal point of activity dering Pollett's lifetime, and every local informant reminisced about it. Pollett describes activities on the bridge in artinous easiey, See, for example, "I'm Bridge," Atlantic Guardina, May 1961; pt. in Ocean (1956), pp.179-97.
 - ³⁸ Pollett. "Afraid in the Dark!" <u>Atlantic Guardian</u>, (Oct. 1954); rpt. in <u>Ocean</u> (1956), p.297.
 - 39 Op. cit., p.301.
 - 40 Op. cit., p.300.
 - 41 Pollett, "The Bridge," <u>Ocean</u> (1956), p.217.
- ⁶ Bonfire Night was another occasion Pollett works about ofters. See especially "shohany the Pear," a fictional character sketch in which Bonfire Night is one of two central events informing the action of the aurantive. An academic treatment of the significance of Bonfire Night, its traditional rivata and functions, may be found in Catherine Ann Schwedermann. "An Exploration of the November Fill Bonfire Celebration in Brigas, New-Southaded" (MA. Thesis, MIXI) 1981). For descriptions of local adaptations of these annual rituals in New Harbour, see C. George Brown, ms.72-919,353-5, MINTLA.
- 43 Jim Thorne, Bishop's Falls, personal interview, 27 May 1983, and Charles Hillier, Grand Falls, personal interview, 24 May 1983. This rather extraordinary occurrence has been corroborated by Reg Woodman, New Harbour, personal interview, 30 May 1983.

- 44 Poilett, "Johnny the Pear," Ocean (1956),p.189.
- 46 Pollett, "There's No Place Like an Outport," Ocean (1956),p.145.
- ⁴⁶ Hannah Brown, New Harbour, personal interview, 10 Sept.1982, and Charles Hillier, personal interview, 26 May 1983.
- ⁴⁷ Jesse Fogwill, telephone interview, 22 June 1983, W. Bramwell Pollett and Margaret Pollett, personal interview, 31 May 1983, and Harold and Alfreda Rowe, Green's Harbour, 31 May 1983.
- ⁴⁹ Information about Colin Jones, the tasks and responsibilities of Newfoundland teachers, class size, and subjects of study, obtained from Church of England Boards, <u>Report of the Public Schools of Newfoundland</u> (St. John's: Herald Printers, 1905-11) (an annual publication, hereafter referred to an <u>Report</u>).
- ⁴⁶ William B. Blackall, "Inspector's Report," in Church of England Boards, <u>Report</u>, 1907, p.114. Information about Jones's part in raising attendance levels and the quality of teaching is from Blackall, "Inspector's Report," in <u>Report</u>, 1906, p.105.
- ⁵⁰ Ralph Higdon, New Harbour, personal interview, I June 1983, and Reg Woodman, personal interview, 30 May 1983. Many other local residents have commented on Pollett's unusually tidy dress and of his speech, the general view is, "you wouldn't know he was from New Harbour."
- ⁶¹ Blackall, "Inspector's Report," in <u>Report</u>, 1910, p.46; and "Inspector's Report," in <u>Report</u>, 1911, p.44
- ⁴² See especially a detailed, three-page questionnaire devised by Blackall, estitled "Local Nature" Observations," in appendix to Egnett, 1908. A former Angliens eshool superintendent, R.L. Andrews, has noted, in a letter to the author, 17 Aug, 1988, Blackall's ardent promotion, spanning over twenty-they areas, of nature sody in Church of England schools, in teachers' college, and in publications, such as F.A. Evston, and Vorman M. Johnson, The Braton Memorial Berger and Section of the Church of
- ⁴⁰ Bernal Readern, Twois (London: Thormas Nelson and Sons, ad.). In an ensy in 1957 by Dr. A.C. Huster, "The Old Broral Readers," in Cycle Rose, Ed. Raffine (Wind and Title, An Antholory of Newfoundland Poetry, Prose and Drams (Portugal Core, Nidd: Breakwarter, 1973), pd.7, it is noted that the texts, designed in 1870, were still in print and their mid-Victorian emphases sever updated. A humorous account of the impact on twentieth-century Newfoundland children of the vestiges of early Victorian precepts contained in the Reader may be found in Cyril F. Poole, "The Royal Readers," in his in Search of the Newfoundland Smil (St. Johns: Harry Cell 1981), pp.5-14. For an extensive treatment of the genesic of these and similar school texts, and their surprising commercial success, see J.S. Bratton, The Impact of Victorian Children's Extend (London: Cronn Heim, 1981).
- ⁵⁴ See Bratton, <u>Victorian Children's Fiction</u> (1981), pp.39-59;103, for an interesting discussion of the functions and popular appeal of "miscellanies," particularly for young boys.

- 55 Royal Reader, No.1 (Preface); and Royal Reader, No.4 (Preface).
- ⁴⁹ Pollett mentions there books in various seasys such as "The Garden on the Point," Occan (1905), p.28x, and Peter the Grate at Output Character Study (St. John's Coardian Associates Ltd., 1923), p.12. Jesse Foyull, in a telephone interview, 19 June 1983, recalls a large personal collection of books owned by Pollett, most of which were more sophisticated in subject matter than texts used in school, and which he frequently lent to her. She emphasizes that Pollett was a vocacious reader; and this comments has been corroborated by Pollett's sixter. Florence Pinsach, Dildo, personal interview, 10 Sept. 1983, and by his contemporary, Ralph Higdon, New Harbour, personal interview, 11 June 1983.
- ST Pollett, "The Bridge," Ocean (1956), p.211. Wordsworth's poem, "Lucy Gray," appears in Royal Reader, No.2, pp.103-5.
 - ⁵⁸ Marion Gosse, telephone interview, 2 Apr. 1984.
- ⁴⁹ Council of Higher Education, Newfoundland. <u>Report of the Examinations</u>, <u>1912-1914</u> (St. John's: 1912-14) (annual publication).
- ⁵⁰ Op., cit., Some statistics worth noting: in 1012, 1973 students and for Primary examinations, and 449 paneed, 890 fairled (freet of the Examinations, 1012, 1015); in 1013, 1031 students and for Preliminary examinations, and 477 passed, 454 failed (Recent of the Examinations, 1013, 285); and in 1914, 599 students as for the Intermediace examinations, and 340 passed, 207 failed (Recent of the Examinations, 1914, p.83). In addition, records of grades kept by the Department of Education confirm Pollet's shiftless—information from these records obtained from Dabbook, Department of Education, St., Iohn's. The information about Pollett's teachers obtained from Church of England abouts, Regard, 1014.
 - 61 Pollett, "The Bait Punt," Atlantic Guardian, Aug., 1954; rpt. Ocean (1956), p.291.
- ⁶² Clarence Cranford, St. John's, personal interview, 29 June, 1983; Max Cranford, New Harbour, personal interview, 1 June 1983; Jesse Fogwill, telephone interview, 22 June 1983. Pollett frequently mentions this routine in essays such as "There's No Place Like an Outport," <u>Ocean</u> (1986), p.134.
- ⁸⁸ Pollett, "The Bridge," <u>Ocean</u> (1956), p.210. The strict adherence in New Harbour to the custom of not undertaking work on Sundays has been discussed frequently in interviews, particularly in terms of examples of punishments incurred when community members did not abstain from labour. But see Faris, <u>Cat Harbour</u>, p.134, where it is noted that the stricture is more a folk than a Christian taboo.
 - ⁶⁴ Caroline Pike, Brooklyn, N.Y., personal interview, 14 Mar.1983.
 - 66 Reg Woodman, New Harbour, personal interview, 30 May 1983.
 - 66 Pollett, "The Garden on the Point," Ocean (1956), p.284.
 - 67 Pollett, "A Born Trouter," Ocean (1956), pp.34-5.
- 68 See Bratton, <u>Victorian Children's Fiction</u>, who provides details throughout his study about the incentives for giving reward books, the kinds of books given, and the practice itself,

which according to him dates from the early nineteenth century.

- ⁶⁹ Pollett, "Where Christmas is Christmas," Ocean (1956), p.41.
- ⁷⁰ See Pollett, "The 'Dicky-Bird' Dish," <u>Atlantic Guardian</u>, Dec. 1949; rpt. in <u>Ocean</u> (1956), pp.94-115.
 - ⁷¹ Pollett, "There's No Place Like an Outport," Ocean (1956), p.132.
- ⁷² Patrick O'Flaherty, "The Myth of the Happy Outport: Ron Pollett and Art Scammell," Newfoundland and her Writers videotape series, No.19, MUN, Educational Television Centre.
- ⁷³ Frederick W. Rowe, <u>Education in Newfoundland</u>, p.126, notes that the required qualifications of teachers remained basically the same between 1892-1916. These qualifications are outlined in Church of England Boards, <u>Report</u>, 1905, Appendix A, pp.126-9.
 - ⁷⁴ Clarence Cranford, St. John's, personal interview, 29 June 1983.
- ¹³ Maurice Barke, in letter to author, 27 Apr. 1983, and in various informal conversations with the author, has noted that the training of teachers in the outport was usually understated through an informal agreement between the pupil and the local teacher. Although teachers' training colleges existed in St. John's well before 1914, mining a flower for mining the contract dates from the outports. The Education Act of 1916, bailed by Rowe for increasing grants and salaries to Newtondiand schools and acaderse, provided sums for teacher training, but only limited sumbers of grants were made available to selected prospective teachers from the outports to travel to St. John's, where they would each receive 1910 per annum. See Rowe, Education in Newfoundland, pp. 120-7, and Governement of Newfoundland. The Education Act of 1916, Sections 49-53 and Schedelt 6, uncatalogued newspaper 816, ACHI.
 - ⁷⁶ Blackall, "Inspector's Report," Church of England Boards, Report, 1914-16.
- Tillie Thorne, New Harbour, personal interview, 30 May 1983, Leah Thorne, New Harbour, personal interview, 1 June 1983, Jesse Fogwill, telephone interview, 19 June 1983, and other informants.
- ⁷⁸ R.T. McGrath, "Why Recruiting is Low Outside St. John's," <u>The Cadet</u> (Nfld.), March 1916, pp.28-9.
- 7º Frederick Rowe, & History of Newfoundland and Labrador (Toronto: McGraw-Hill Ryerson), p. 372. Goldon-G W.L. Nicholson, in <u>The Eighting Newfoundlander: A History of the Royal Newfoundland Regiment</u> (S. John's: Government of Nida, 1964), pp.199-30, describes various recruiting drives undertaken between 1914 and 1915, and the rather stringent requirements for recruits.
- ⁸⁰ W. Bramwell Pollett and Margaret Pollett, personal interview, 31 Aug. 1982, Florence Pinsent, Dildo, personal interview, 10 Sept. 1982, Ronald J. Pollett, Brooklyn, N.Y., personal interview, 3 Mar.1982, and other informants.
- ⁸¹ These duties of the Supervising Officer are outlined in Council of Higher Education, Newfoundland, Report of the Examinations, 1912, p.56.

- ⁸⁰ Maurice Burke, letter to author, 7 Apr. 1983, includes a xeroxed reproduction of Pollett's creation papers of Jan. 1917. Burke has noted in informal conversations with the author that in lites of formal examinations required of a pupil-teacher in St. Johns, a local outport teacher's personal recommendation of the candidate was often sufficient for an outport pupil-teacher to obtain his teacher's certification.
- ³⁸ Caroline Pike, Brooklyn, N.Y., personal interview, 3 Mar.1983, W. Bramwell Pollett and Margazer Pollett, personal interview, 9 Sept.1882, and Jense Fogwill, telephone interview, 19 June 1983. The large number of deaths in New Harbourd during 1916 is mentioned frequently in Pollett's writings, and the fact is confirmed in the Church Registry, 1916, Church of St. Augustine, New Harbourd.
- ** See Reid-Nild. Correspondence, "Heart's Content Branch Line 1911-17" file, GN 2/9, Box 6, 48/4/5, PANL; and Melvin Rowe, Have Touched the Greatest Ship: A Seas of Heart's Content (St. John's: Town Crier, 1976), pp.170-3, for a history of the line, and its local impact.
 - 85 Pollett, "Johnny the Pear," <u>Ocean</u> (1956), p.179.
- ⁴⁰ Jim Thorne, personal interview, 27 May 1983, corroborated by Raymond Hillier, Rosemend. California, letter to sathere, 6 Cet. 1983. "Eeling" is mentioned in a biographical sketch of Pollett by Ewart Young in "Gauerian Angles." <u>Athantic Gauerian</u>, Sept. 1966, p. l. it is stated that the power was printed when Pollett was a schoolopy. Pollett's adjustly to create reprinting a stories has been discussed by Marinu Gome, beliphase interview. 2 Apr. 1984, Rajah Higdon, matter, 1, Jan 1985, Leep Gought, theplowas networks, 27 Apr. 1984, Rajah Higdon, matter, 2004.
- ⁸⁷ Bertha Pollett, in a letter to author written for her by her niece Mildred Johnson, 10 Jan. 1984. The song "Nicholas Tobin" is discussed in chapter four of this thesis, p.73.
- ⁸⁸ Ralph Higdon, personal interview, 1 June 1983, and Reg Woodman, personal interview, 30 May 1983. Jesse Fogwill, telephone interviews, 19 and 23 June 1983, Mildred Johnson, letter to author, 17 July 1983, and other information.

II. Learning his Trades: The Transient Years, 1917-24

- ¹ This estimate has been derived from the <u>Census</u>, 1911 and 1921.
- ² Information on the early history of English Harbour may be found in a student essay by Ian Penney, "Research Paper on English Harbour, Trinity Bay," File E-6, MHG.
- ³ Penney, in "Research Paper," pp.32-5, notes the extent of emigration of persons between the ages of 30 and 30 during the period [319-03, adding that the largest age group in the village comprised persons who were between 50 and 70 years old. His source is the Centus, 1911 and 1921. Local men's enlittenest to fight in The Gerst Wer may also account for the decrease, although there were other incentives for leaving, In 1933, Commissioner William Warren Mackenie Amulters aggrested that the inception of the Newfoundand Railway and its branch lines had had serious effects on Newfoundhand community life: "Men were lared away from the thinty in the hope of regular and less ardroom sout," See Great Brains Newfoundhand Royal Commission.

Report (London: His Majesty's Stationery Office, 1933), p.40 (hereafter, <u>Amulree Report)</u>. For a personal account of the effects of emigration on the people from the English Harbour area, see Violet Randell, <u>Thursday's Shild</u> (New York: Carleton Press, 1975).

- 4 Pollett, "The Garden on the Point," Ocean (1956), p.287.
- ⁵ Penney, "Research Paper," p.1.
- ⁶ Information about the Frank Penney family, the general tasks of local families, and the high regard accorded to local teachers obtained from personal interviews in English Harbour with Gertrude Bugden, 29 May 1983, Harvey Jones, 29 May 1983, Willis Penney, 29 May 1983, and telephone interview with Jo Ann Penney, 29 May 1983.
- Information about Pollett's tasks in English Harbour obtained from personal interviews with Ray Fowlow, Port Rexton, Trinity Bay, 23 May 1983, and Gertrude Bugden, English Harbour. 29 May 1983. The vestry records may be found in All Saint's Gutrerh, English Harbour.
 - 8 Church of England Boards, Report, 1917, p.14; and Report, 1918, p.14.
 - 9 Jo Ann Penney, telephone interview, 29 May 1983.
- ¹⁰ Information about the schoolhouse and furnishings is from Church of England Boards, <u>Report</u>, 1917, p.14, where Blackall noted that these were "hardly satisfactory," and from personal accounts obtained in interviews with Gertrude Bugden, 29 May 1983, and Willis Penney, 29 May 1983.
- ¹¹ Church of England Boards, <u>Report.</u> 1917, p.14. The rating system was used regularly, and an outline of its terms of evaluation may be found in each annual report.
- ¹⁷ The day of Superintendent Blackall's annual visit was an important occasion. According to the accounts of Willis Pearsy, Gentred Bugden, and other, students were expected to fress in their "Sunday best," and impress Blackall by demonstrating their progress and good behavior. A bunneous account of the trepidation left out the occasion and shared by statesters and teachers alike may be found in Cyril F. Poole, "For the Love of Men's Soul's," in In Search of the Newfoundhand Soul, p.21.
- ¹³ Church of England Boards, <u>Report</u>, 1917, p.14. The next year, in the <u>Report</u>, 1918, p.42, it was noted that the school board, whose chairman was the minister, Arthur Pittman, had allowed a loan and gift totalling \$100 to be used for the local school building.
 - ¹⁴ Blackall, in Report, 1918, p.iii.
 - ¹⁵ Debbie Butler, Queens, New York, telephone interview, 17 Mar. 1983.
 - ¹⁶ Gertrude Bugden, personal interview, 29 May 1983.
 - ¹⁷ Jo Ann Penney, telephone interview, 29 May 1983.
- ¹⁸ Information about strictures on local activities obtained from Gertrude Bugden, personal interview, 29 May 1983, and Willis Penney, personal interview, 29 May 1983. Church teas,

regular meetings of a local temperance society, and "sales of work" (i.e.,handicrafts), were among the limited social activities in the village.

- ¹⁹ This photograph is in the possession of Gertrude Bugden, English Harbour.
- 20 Gertrude Rugden, personal interview, 29 May 1983.
- ²¹ Pollett's salary is noted in Church of England Boards, Report, 1919, p.19.
- ²² Gordon S.A. Cox, in Edik Music in a Nerdennalhad Ontson, National Moseum of Man, Mercury Series, Gandlain Castries for Sic Culture Studies, Paper no. 23 (Ontsure National Museum of Canada, 1980), pp. 503-545;117, implies this traditional relationship between the New Harborn and Green's Harborn communities, Information about Polletic's place of residence and familiarity with the Green's Harborn community obtained from Bertha Burt, Green's Harborn and Polletic Place of Polletic Place of Series (1982), and Charles and Laury Thomes, New Harborn, personal interview, 2 Sept. 1982.
- 25 The physical description of Green's Harbour is based on Cox's depiction, $\underline{\rm Folk}$ $\underline{\rm Music.}$ p.16.
 - ²⁴ An approximation based on the <u>Census</u>, 1921.
 - 25 Church of England Boards Report, 1919, p. 19.
 - ²⁶ Bertha Burt, letter to author, 27 June 1983.
- ²⁷ See Charch of England Boards, Report, 1919, p.iv, where it is noted that the majority of Newfoundland schooliditized leid school by age 12 and before they had passed Grade & Martin G. Rowe, in a student ensay, "The History of Green's Harbour, Trainity Bay," McG-28, p.77, MSIG, notes that this community, shirtorically, had had an especially high illnerary rate; and in Charch of England Boards, (Eggart, 1965-12, in particular, the Green's Harbour school and its state-desiculous obstander from Eay Bennett, Creen's Harbour, personal interview, 31 May 1983, and Harbour, personal interview, 31 May 1983, and Harbour, personal interview, 10 Spec. 1982.
 - 28 Bertha Burt, letter to author, $27\,$ June 1983.
- 29 Harrison Hillier, personal interview, 31 May 1983, and Bertha Burt, letter to author, 27 June 1983.
- ²⁰ S. J. Noel, Politics in Newfoundhand (Toronto Univ. of Toronto Press, 1971), p.129, notes that the effect of the Great Way on the Newfoundhand comonly was not "exceptional prosperity" from the trade in fish. Though the "bom" was short-lived, be says that as a result of it, "Habits developed in this free-spending era, and expectations were aroused, that lingered long after the means of satisfying them had vanished." Teachers' salaries were not affected during the rea.
- 31 Jesse Fogwill, telephone interview, 22 June 1983; the incident was frequently related to her by Pollett's sister, Clara (now deceased).

- ³² Ralph Higdon, personal interview, 1 June 1983, W. Bramwell Pollett, personal interview, 30 May 1983, and other informants in New Harbour.
- ³⁵ Education Act of 1916, Section 46(d). The requirements for Grade Two certification also are included in Section 46.
- ³⁴ As a result of the Education Act of 1916, more formal requirements had been prescribed for all grades of teacher certification, and a nummer school for teachers had been instituted in St. John's. There is no evidence that Pollett attended any such school (attendance roles are listed in Church of England Boards, <u>Regest</u>, 1919), and it cannot be accertained whether his non-attendance affected his advancement in the teaching profession.
 - ³⁵ Bertha Burt, letter to author, 27 June 1983.
 - ³⁶ Church of England Boards, Report, 1919, p.19.
- ³⁷ Bertha Burt, letter to author, 27 June 1983, and Harrison Hillier, personal interview, 31 May 1983. These particular texts have not been located.
- 38 Bertha Burt, letter to author, 27 June 1983, Harrison Hillier, personal interviews, 10 September 1982 and 31 May 1983, and other informants in Green's Harbour.
 - 29 Bertha Burt, letter to author, 27 June 1983.
 - 40 Census, 1921.
- ⁴¹ For background details on the Harmsworths' agreement with the Robert Bond government, and a study of the Grand Falls paper mill's inception (which transformed the Newfound-land forest industry into a pulp and paper industry) and its production prior to Confederation, see James Hiller, "The Origins of the Pulp and Paper Industry in Newfoundland," <u>Acadiensis</u> vol.2 (Spring 1982), pp.42-68.
- ⁴² George Hicks, "Newfoundland from Shadow to Sunlight," ms.(1968), p.76, CNS. The citation is from a speech Lord Northcliffe gave to Church Boys' Brigade, Grand Falls, 1909.
- ⁴³ Joseph R. Smallwood, Ed., <u>The Book of Nurfoundland</u>, vol. 1 [St. John's: NewGoundland Book Publishers Lid, 1937], pp.236. Various informants interviewed in Grand Flath kave cited additional leisure activities made available by Northcliffe, such as movies (Northcliffe donated a movie projector prior 1920), and all of these attractions were available by the time Pollett arrived in Grand Falks. Personal interviews in Grand Falks with Robert Arkeley, 27 May 1983, John Blackmore, 28 May 1983, Charles Edward 25 May 1983, and other informants.
 - 44 Smallwood, The Book of Newfoundland, vol. 1, p.23.
- ⁴⁶ Walter Blackmore, Grand Falls, personal interview, 24 May 1983, Jack Brown, Grand Falls, personal interview, 26 May 1983, Hiram Silk, Grand Falls, personal interview, 26 May 1983, and W.T. Howell, Grand Falls, letter to author, 29 June 1983.
- ⁴⁶ Caroline Pike and Ronald J. Pollett, Brooklyn, N.Y., personal interview, 6 Mar. 1983, Walter Blackmore, personal interview, 24 May 1983.

- ⁶ Information about the Exploits Hotel obtained from personal interviews in Grand Falls with Louise Bugdes, 55 May 1983. Chairse Edwards, 25 May 1983, Abm Blackmere, 28 May 1983, and Walter Blackmore, 24 May 1983. Mr. Edwards, former town mayor, emphasizes the differences between various bactes in Grand Falls: the Calob House and the Erin House were similar to the Exploits Hotel in extering to men with skilled positions, whereas a Workman's Hotel careered to general workers, and facilities there could be compared to the bankmoss men lived in when they worked in Immbercamps. A number of student essays at the Maritime History Group point to other appears of this historic division between blooming classes in the company town: see for example, Anne George, "The Grand Falls Pulp and Paper Mill; its Origin, Growth, and Impact on the Surrounding Area", mis. BE-124, MHG.
 - $^{\rm 48}$ Charles Edwards, personal interview, 25 May 1983.
- ⁶⁰ The three phases of the town's growth are outlined in [George Hicks], "Anglo-Newfoundhand Company Ltd.; Exclaimment and Fifty Yearn of Progress," no. [in possession of Walter Tucker, Grand Falls), p.10. Being a "closed town" in which Northcliffe owned all property rights, Grand Falls grew according to the A.N.D. Company's needs.
 ⁵⁰ Walter Blackmore, telephone interview, 10 May 1983 and personal interview, 24 May
- 1983, Charles Edwards, personal interview, 25 May 1983, W.T. Howell, letter to author, 29 June 1983.
- ⁵¹ Information about the mill's production obtained from [George Hicks?], "Anglo-N\u00edd. Company," pp.10-12.
 - 52 Op. cit., pp.11-17.
- ³⁵ Op. cit. Figures for manual labourers obtained from this manuscript have been confirmed by John Blackmore, personal interview, 28 May 1983, Charles Hillier, personal interview, 26 May 1983, and Alphonsus Power, 25 May 1983—these men noting that such labourers worked a tenhour day.
- ⁵⁴ See a student essay by John A. Southcott, "Church and School History of Grand Falls (1905-1940's)," ms. G-18, MHG.
 - 55 W.T. Howell, letter to author, 29 June 1983.
 - ⁶⁸ Walter Blackmore, telephone interview, 19 May 1983.
- ⁸⁷ The specific details of Pollett's poem about the Exploits Hotel were obtained from W.T. Howell, letter to author, 29 June 1983. Other comments and impressions about Pollett as a writer during the Grand Falls years were obtained from Howell, and from Michael Blackmore, telephone interview, 25 May 1983, and Walter Blackmore, personal interview, 24 May 1983.
- ⁵⁸ Michael Blackmore, telephone interview, 25 May 1983; corroborated by Alfreda Silk, personal interview, 26 May 1983.
 - 59 Michael Blackmore, telephone interview, 25 May 1983.
- ⁶⁰ Information about Emma Hackett, and Pollett's relationship with her, obtained from Walter Blackmore, personal interview, 24 May 1983, Caroline Pike, personal interview, 15 Mar.

1983, Ronald J. Pollett, personal interview, 20 Mar. 1983, Alfreda Silk, personal interview, 26 May 1983, and Hiram Silk, personal interview, 26 May 1983.

- ⁶¹ W.T. Howell, letter to author, 29 June 1983. A photograph of Pollett, along with other charter members of the Loyal Orange Young Briton Association, is in Howell's possession.
- ⁶² See [Hicks?], "Anglo-Newfoundland Company," pp.13-25, for a detailed account of the conditions leading to the strike. Similar information appears in a student essay by Patrick Melvin "Grand Palls Past Present and Future" ms. G.16 no. 19-21. MRIG.
 - 68 Church Registry, Church of St. Augustine, New Harbour,
- ⁶⁴ Information about the relationship between Sophia and Jim Pollett obtained from personal interviews with Caroline Pike, 17 Mar. 1983, Ronald J. Pollett, 20 Mar. 1983, Ronald J. Pollett and Caroline Pike, 6 Mar. 1983, W. Bramwell Pollett and Margaret Pollett, 31 May 1982, and other informants.
- ⁶⁶ Jim Thorne, personal interview, 27 May 1983; Jesse Fogwill, telephone interview, 22 June 1983. Other similar information about Sophia Pollett obtained from personal interviews with Raujah Higdon, 1 June 1983, W. Bramwell Pollett and Margaret Pollett, 9 Sept. 1982, Alfreda and Harold Rowe, 31 May 1983, and other informats.
- S.J. Noel, in <u>Politics in Newfoundland</u>, p.144, notes that the remarkable period of prosperity during 1914-10 was short-lived, and that in 1919 Newfoundland was "on the brink of economic catastrophe." See his discussion of the political and economic conditions in Newfoundland during the aftermath of the Great War, pp.144-87.
- ⁶⁷ See [Hicks?], "Anglo-Newfoundland Company," pp.18-29, for details about the ending of the strike, and an implied comparison between the remarkably thriving nature of life in the town and most other places in Newfoundland.
- ⁶⁸ See, for example, two notices by R.A. Squires, Colonial Secretary, and by the company, urging more men not to embark on journeys to Grand Falls, in <u>Evening Telegram</u>, 21 Aug. 1921, p.4.
 - ⁶⁹ W.T. Howell, letter to author, 29 June 1983.
- ⁷⁰ Walter Blackmore, in a personal interview, 24 May 1983, notes that not until the inception of the <u>Grand Falls Advertiser</u> in 1936 were printing facilities and machinery available in the town. Even then, there were no opportunities for the training of printers.
- ⁷¹ Government of Canada, Department of Labour, <u>Labour Gazette</u>, <u>1922</u> (Ottawa: Government Printing Bureau, 1922), p. 263.
 - 72 Op. cit., p.846.
- ⁷⁸ Pollett, "Memories of 'Didder Hill," Atlantic Guardina, Apr. 1981; rpt. in Ocean (1995), p. 198; and "The Outport Millionaire," Atlantic Guardina, July 1946; rpt. in Ocean (1996), p. 24. A rather dismal portrait of living conditions known to 80 per cent of the population of Montreal during the 1920s is provided in Terry Copp. The Anatomy of Powerty: The Condition of the Working Class in Montreal, 1887; 1922 (Protontic McClelland and Stewart, 1974), pp. 70-8.

¹⁵ Peter Neary, in "Canadian Immigration Policy and the Newfoundlanders, 1015-1309," Auditainst vol. (Spiring 1882), pp. 674, does increasingly restrictive Canadian immigration policies applied to Newfoundlanders by 1020 (whereas previously this group had been treated much the same as "restruct" Canadiana, and among the stiplications of an Order-Go-Ouncil inseed in decimals in Canada, Policit support of the New Policy of the Po

²⁶ The terms of apprentice training for linestype newspaper work are set forth in "Montreal Apprentice Agreement, International Typorophysical Union, 18", "Labout Gazette 1912 (Ottawa C.H. Farmeler, 1912), pp.546-7; later revisions are set forth in Labout Gazette 1921, pp.892-3, and Labout Gazette 1922, pp.128-7. This training for envespaper work differed dramatically from the usual apprenticeships for very young men, whose training required a five-year apprenticeship, Information about the particular skills which interpolate water equired to learn obtained from Gunsar-Janger, New York, and Thomas Kopeck, Colorado Springs, Colo.), each lesson of which all international appearatices was required to master.

⁷⁶ Information about the linotype machine may be found in Alexander Lawson, <u>Printing Tyrner</u>, an <u>Introduction</u> (Boston: Beacon Press, 1971), pp.6-35; and Victor Strauss, <u>The Printing Industry: An Introduction to its Many Branches, Processes and <u>Products</u> (Washington, D.C.: Printing Industries of America, Inc., 1967), pp.62-33.</u>

⁷⁷ Thomas W. Kopeck, letter to author, 13 July 1983.

⁷⁸ See Government of Canada, <u>Labour Gazette</u>, 1921-24.

⁷⁹ Government of Canada, <u>Labour Garette 1924</u>, p.861. It is noted in the <u>Labour Garette 1921</u>, p.799, that New York City was "the only important city to make the change [to the 44-hour work week] without conflict."

No This estimated figure derived from Government of Canada, Census of 1921 (Ottawa: F.A. Acland, 1929), figures in Government of Canada, Labour Gazette 1921-1924, and in Government of Canada, Fourteenth Annual Report on Labour Organization in Canada (Ottawa: F.A. Acland, 1925), passim.

⁸¹ Jack Williams, <u>The Story of Unions in Canada</u> ([Toronto?]: J.M. Dent and Sons, 1975), pp.10-11.

Thomas W. Kopeck, letter to author, 13 July 1983. It is noted in various works on American immigration policy that despite increasingly restrictive measures such a soport estrictions on American immigration during 1921-4. Canadians were given preferential treatment. See Marion T. Bennett, American Immiration (Delicies A History (Wanhajton, D.C.) Public Afairs Press, 1993), for a study of various measures during the period; William S. Bernard, Ed., American Immiration and Camanda (Press, 1995), pp. 43, and George Marion (Press, 1995), pp. 43, and pp. 43, and pp. 44, and pp. 44

⁸⁸ Pollutt is réferred to au "boss-lindayjatt," in a feature stop; "Cilea Cove Printer to Save. Lord Northeillië Legacy for Some Need of Future, Thomburn Dailt, Easle, Nov. 23, 1034, in "Pollett" and the Company, Bookhyn Dailt, Easle, Nov. 23, 1034, in "Pollett" and Company, Bookhyn Dailt, Easle, Nov. 23, 1034, in "Pollett" and Company, Bookhyn Company, Co

84 Coles and van Santvoord, Glen Cove, pp.60-4.

⁸⁰ Information about the size and local name of the neighborhood obtained from Caroline Plex, personal interview, 3 and 15 May 1838, Romath J Pollett and Caroline Plex, personal interview, 6 May 1983, Belle Faver, Brooklyn, N.Y., telephone interview, 6 May 1983, A belle Faver, Brooklyn, N.Y., telephone interview, 8 May 1983, and other informants. The particular area of Park Slope in which NewFounditaident tended to settle, according to oral report, corresponds which NewFounditaident tended to settle, according to oral report, corresponds which local light interfestes are noted to have settled. So, Julium Mitzner.

Niciabondos Study of the Park Sione Community in Brooklyn (NY: The New York School of Social Work, 1904), p. 10. No mention of Newtourdanders as a national group has been located in any of a large number of books focussing on Park Slope or other Brooklyn neighborhoods. An oversight in ceasus data documentaing settlers' anxional origins is possible nee Rowland Tappan. Berthoff, British Immirrants in Industrial America 1709- 1939 (NY: Russell and Russell, 1953).no.5-401.

⁸⁶ See "Newfoundland" file, <u>Brooklyn Eagle</u> Morgue, Brooklyn Public Library.

⁸⁷ See "Clea Core Printer," <u>Brooking Dalk Earle</u>, Information about the legacy derived from Northelife's 118 million estate is contained in this article other sources providing details about Northelife and his legacy are; John Braddock, "One Bold Venture than Started a Northelife and his legacy are; John Braddock, "One Bold Venture that Started a Northelife community and Installary and its SILI Developing an Abandwe Cade," in NY. Howell, Comp. "Mobry Leg." (Grand Falls, Ndd. Price (Ndd.) Pulp and Paper Co., Ltd., 1968), non 10-15; and WY. Howell, Letter to about 70.9 June 100.

88 "Glen Cove Printer," Brooklyn Daily Eagle.

³⁹ Op. cit. See also Raymond A. Schroth, <u>The Eagle and Brooklyn: A Community Newspaper</u>, <u>1841-1955</u> (Westport, Conn.: Greenwood Press, 1974).

90 "Glen Cove Printer," Brooklyn Daily Eagle.

91 Op. cit.

⁹² Information about Hester obtained from Schroth, <u>The Eagle and Brooklyn</u>, pp.13;54;143;185;263; and Raymond A. Schroth, Worchester, Mass., letter to author, 3 Jan. 1984.

25 Pollett, "Memories of 'Didder Hill,'" Ocean (1956), p.198.

III. 'New York-Newfoundlander,' 1925-45

¹ Information about Pollett's place of residence and a general overview of life in the neighborhood obtained from Caroline Pike, Brooklyn, personal interview, 3 Mar. 1983, Ronald J. Pollett, Brooklyn, personal interview, 5 Mar. 1983, Ale Foave, Brooklyn, kelphone interview, 6 Mar. 1983, Abe Snow, kelphone interview, 18 Mar. 1983, and other informants. See also Pollett's essay, "Memories of 'Didder Hill." (Jozan 1965) pp. 198-200.

² Pollett, 'Memories of 'Didder Hill,' Ocean (1956), p.201.

⁸ Op. cit., p.199.

4 These functions of earlier immigrants from Newfoundland, and local institutions usually organized by or participated in by them, would appear to conform to Raymond Breton's concept of "institutional completeness." a development which enables immigrant groups to maintain their ethnic identity by not depending on services and institutions maintained by the receiving society. See Raymond Breton, "Institutional Completeness of Ethnic Communities and the Personal Relations of Immigrants," American Journal of Sociology, no.70 (1964), pp.193-205. Such a development accounts for what it termed the "beaten paths" theory of migration, in which word-ofmouth information obtained from earlier immigrants about jobs and other necessities provides inducements for later immigrants to emigrate to the same communities. See Peter Morrison. "The Functions and Dynamics of the Migration Process," in Alan A. Brown and Egon Neuberger. Eds., Internal Migration: A Comparative Perspective (N.Y.: Academic Press, 1977), pp.61-72. However, despite the commonplace knowledge in Newfoundland that Brooklyn was a key centre for immigrants- many people have at least one relative there, in what is referred to as the "Boston States" (Brooklyn is included in this term)- a study of the Newfoundland community in Brooklyn has never been undertaken. Studies of Newfoundlanders abroad are: Edward Vincent Chafe, "A New Life on 'Uncle Sam's Farm:' Newfoundlanders in Massachusetts. 1846-1859" (M.A. Thesis, MUN., 1982); David William Abbott, Rural to Urban Migration Adjustment; A Study of Newfoundlanders in Toronto (Toronto: York Univ., 1975); Ann Elizabeth Martin, "Up-Along: Newfoundland Families in Hamilton" (M.A. Thesis, McMaster Univ., 1974); and Larry Orton. "Newfoundlanders in Toronto," in Perspectives on Newfoundland Society and Culture: Book of Readings, Maurice A. Sterns, Ed. (St. John's: MUN, 1974), pp.82-3.

 5 Pollett, "Up in the Big City," Part I, <u>Atlantic Guardian.</u> June, 1951; rpt. in <u>Ocean</u> (1956), p.222.

⁶ The term, "the jigging ground," is in popular thought associated with the camaraderie and gibbermen who converge on a particular fishing ground to "jig," or, fish for, the squid, a bait fish. See a popular song by AR. Scammell, "The Squid-Jigging Ground," in his collection, Mr Newfoundland [Montreat: Harvest House, 1969], pp. 118-9. Pollett's reactions to Manhattan and Brooklyn oblined from Abe Sout kelenkoen interview. 18 Mar. 1986.

⁷ Pollett, "Memories of 'Didder Hill," Ocean (1956), p.199.

⁸ Information about Pollett's work routine and work-related activities obtained from Caroline Pike, personal interviews, 3 Mar. 1983, 14 and 15 Mar.1983, Ronald J. Pollett, personal interviews, 20 Mar. 1983, and Abs Snow, telephone interview, 18 Mar. 1983.

- ⁹ See Government of Canada, <u>Labour Gazette 1912</u>, p.547, in which an agreement between LT.U. and Local No.176 states: "Operators shall receive price and a half for all alterations"—sets changes in sepling and punctuation. This clause was unaltered in a new agreement in Montreal in 1921, and must have been in force as well in New York, since all agreements and policies of the union were standardized.
- ¹⁰ An estimate of Pollett's salary obtained from figures in Charlotte E. Morgan, <u>The Origin and History of the New York Employing Printers' Association</u>, (N.Y.: AMS Press, 1998), p. 127, and from xerozed reproductions of contracts from the period between LT.U. Local No. 6 and Printers' League, obtained during a personal interview with Gennar Janger, New York, New York, 19 Mar. 1983. Snow's comment obtained during leiphone interview, 18 Mar. 1983.
- ¹¹ Information about Caroline (Gilbert) Pollett and her relationship with Ron Pollett obtained from her, now Caroline Pike, in several personal interviews on 3 Mar. 1983, 15 and 16 Mar. 1983.
 - ¹² Caroline Pike, personal interview, 3 Mar. 1983, Ronald J. Pollett, personal interview, 20 Mar. 1983, Bellle Feaver, telephone interview, 6 Mar. 1983, and Abe Snow, telephone interview, 18 Mar. 1983.
- 18 Information about Pollett's wedding obtained from Caroline Pike, personal interviews, 3 and 14 Mar. 1983.
- ¹⁴ Pollett, "Up in the Big City," Part II, <u>Atlantic Guardian</u>, July, 1951; rpt. in <u>Ocean</u> (1956), p.233.
- ¹⁶ Abe Snow, telephone interview, 18 Mar. 1983. As Snow notes, and Caroline Pike corroborates (personal interview, 18 Mar. 1983), this move was often facilitated through Newfoundland-born friends: Pollett, for example, obtained his partners on 28 and Street with the help of his close friend John Pike, who lived near to Snow in Bay Ridge. Both Snow and Pike formerly lived in Park Slope.
- ¹⁰ A study of the rapid change in the Park Slope neighborhood during the late 1920s may be found in Julius Mitzner, Neighborhood Study of the Park Slope Community in Brooklyn, pp.5-10, and Timothy O'Hanlon, "Neighborhood Change in New York City: A Case Study of Park Slope, 1850 Through 1980" (M.A. Thesis, N.Y.U., 1982), pp.125-30.
- ¹⁷ Information about Pollett's homelife obtained from Caroline Pike, personal interviews, 3 Mar. 1983, 14 and 15 Mar. 1983.
- ¹⁸ Information about Pollett's changing character and new domestic routines obtained from Caroline Pilke, personal interviews, 14 and 18 Mar. 1983, and Abe Snow, telephone interview, 18 Mar. 1983. A series of photographs dating from the period is in Ronald J. Pollett's personal collection. Information was also obtained from the "Bay Ridge-Fort Hamilton" file, Long Island Historical Society, Recollyn.
- ¹⁹ Details of the card-party activities obtained from Caroline Pike, personal interviews, 3 and 17 Mar. 1983, and Abe Snow, telephone interview, 18 Mar. 1983.
- ²⁰ Figures on Pollett's income obtained from Morgan, <u>Printers' Association</u>, p.127; and LT.U., No.6 contracts of the period. The estimated income thought sufficient during the period is

cited in Caroline Bird, The <u>Invisible Scar</u> (N.Y.: McKay, 1966), p.40. Her source is a public speech by Senator Paul McKay. Bird notes that prior to Oct., 1929, more than two-thirds of American families were living on \$2500 per year or less.

²¹ Caroline Pike, personal interview, 3 Mar. 1983.

22 Pollett, "Up in the Big City," Part I, Ocean (1956), p.224.

²³ Information about Pollett's activities during the period obtained from Caroline Pike, personal interviews, 3 Mar. 1983, 15 and 16 Mar. 1983.

²⁴ Caroline Pike, personal interview, 3 Mar. 1983.

²⁵ Pollett, "Memories of 'Didder Hill," <u>Ocean</u> (1956), p.198.

²⁶ Ronald J. Pollett, personal interview, 6 Mar. 1983.

 27 Caroline Pike, personal interview, 3 Mar. 1983. Other information about Pollett's working conditions obtained from this interview.

²⁰ Caroline Bird, in <u>The Invisible Scar.</u> pp.81;79, provides a detailed explanation of "shared employment." Raymond A. Schroth, in <u>The Eagle and Brookivn</u>, pp. 13:24, suggests that the <u>Eagle's</u> fortunes were better than most other local newspapers, although during the 1930s it began to lose its considerable prestige nationally.

²⁹ Pollett, "Summer Madness," Atlantic Guardian, Aug. 1950; rpt. in Ocean (1956), p.122.

³⁰ Pollett, "Memories of 'Didder Hill," Ocean (1956), p.202.

³¹ Information about Pollett's new quarters and routines obtained from Caroline Pike, personal interviews, 3 Mar. 1983 and 14 and 15 Mar. 1983. Caroline Bird, in <u>The Javisible Scar</u>, notes throughout her book that similar practices of sharing and scrimping were typical among American families during the Decression.

⁵² Information about Pollett's health condition obtained from Caroline Pike, personal interview, 3 Mar. 1983, and Ronald J. Pollett and Caroline Pike, personal interview, 6 Mar. 1983.

²³ Information about Pollett's newly acquired habits obtained from Ronald J. Pollett and Caroline Pike, personal interview, 6 Mar. 1983, Ethel Mae Barnes, Centerreach, N.Y., telephone interview, 19 Mar. 1983, and Abe Snow, telephone interview, 18 Mar. 1983.

³⁴ The figure is from Caroline Bird, <u>The Invisible Scar</u>, p.64. She notes that pari-mutuel betting on horse races was legalized in 1933.

³⁵ Ronald J. Pollett and Caroline Pike, personal interview, 6 Mar. 1983.

³⁶ Abe Snow, telephone interview, 18 Mar. 1983.

- ³⁷ Information about Pollett's plan, his trip to Newfoundland and return to New York, obtained from personal interviews with Caroline Pike, 3 Mar. 1983, and Ronald J. Pollett and Caroline Pike, 6 Mar. 1983.
- 58 See <u>Amulree Report</u>, p.78; various other references to the dire conditions of the populace are to be found in this report.
 - 39 Raymond Hillier, Rosemead, Calif., letter to author, 6 Oct. 1983.
- ⁶⁰ Caroline Pike, personal interviews, 3 and 6 Mar. 1983. Some New Harbour residents indeed thought of their stay as an extended holiday, as noted by Mildred Johnson, letter to author, 17 July 1983, Max Cranford, personal interview, 1 June 1983, and other informants in New Harbour.
- ⁴¹ For an examination of political and economic conditions leading to Britain's instituting of Commission of Government, see Noel, <u>Politics in Newfoundland</u>, pp.188-214. A graphic description of the "poverty and misery" suffered by Newfoundlanders of the period may be found in Joseph R. Smallwood, <u>J. Chose Canada</u>, pp.180-96.
- ⁴² Amultee Resort, po.767. In addition, the lack of remittances (totalling approximately \$1 million per year) formetly provided by immigrants exacerbated families' financial situations. These remittances, as noted in R.A. Mackay, <u>Newfoundland, Economic, Diplomatic, and Farteetic Studies</u> (Tovanic: Oxford Univ. Press, 1946)p.72, were reduced by about 75 per cent during the period.
- ⁴³ Information about AD Press obtained from Michael Weinberg, N.Y.,N.Y., telephone interviews in Grand Falls with Walter Blacke, of "law printers" were described in personal interviews in Grand Falls with Walter Blackene, 24 May 1983, and Bruce Fudge, 25 May 1983; and with Ted Long, St. John's, 27 June 1983.
- ⁴⁴ Information about Pollett's new home, his family's situation, and his new domestic activities obtained from personal interviews with Caroline Pike, 6 and 15 Mar. 1983, Ronald J. Pollett and Caroline Pike, 6 Mar. 1983.
 - ⁴⁵ Caroline Pike, personal interview, 3 Mar. 1983.
- ⁴⁶ Pollett, "Memories of 'Didder Hill," <u>Ocean</u> (1956), p.199. Information about Prospect Park obtained from "Prospect Park" file, Long Island Historical Society, Brooklyn, N.Y.
 - 47 Ralph Higdon, personal interview, 1 June 1983, Max Cranford, personal interview, 1 June 1983, and other informants in New Harbour.
 - ⁴⁸ Mildred Johnson, letter to author, 17 July 1983.
- ⁴⁰ The inception of the town's newspaper and the ambitions of its young owners are described in Laura Blackmore, "A Boyhood Dream Comes True," <u>Grand Falls Advertiser</u>, T July 1980, pp.1-3; and in Fraser Lush, "The Press in Central Newfoundland," <u>Journal</u> [of Business and Commerce, St., John's] [Nov. 1987], pp.29-32.
- ⁵⁰ Pollett's active participation in the development of this newspaper was described by Michael Blackmore, telephone interview, 25 May 1983, and by Walter Blackmore, telephone

interview 19 May 1983, and personal interview, 24 May 1983.

- 81 Walter Blackmore, personal interview, 24 May 1983.
- ⁵² Walter Blackmore, personal interview, 24 May 1983, although Pollett's name is not mentioned in various articles describing the success of the paper, such as L. Blackmore, "A Boyhood Dream," and Lush, "The Press in Central Newfoundland."
 - 58 Walter Blackmore, personal interview, 24 May 1983.
 - ⁵⁴ "Glen Cove Printer," <u>Brooklyn Daily Eagle</u>, Nov. 25,1924.
- ⁵⁶ Walter Blackmore, personal interview, 24 May 1983. Ronald J. Pollett, in a personal interview, 20 Mar. 1983, notes that Pollett and other local Newfoundlanders shared a custom of trading island newspapers they either received in the mail or bought from various local newstands in Brooklya and Manhattan which regularly carried Newfoundland newspapers.
 - 56 Ronald J. Pollett, personal interviews, 6 and 20 Mar. 1983.
- ²⁷ Ronald J. Pollett, personal interview, 6 Mar. 1983. The full influence of these popular radio programs on Pollett's later wining can only be conjectured. He would acknowledge the surprising popularity of the "Doyle News Bulletin' very frequently in his prose (see, for example, "On Gong Home Again," Attaintic Ganzalian, Mar. 1906, pp. in Ozcara (1996), pp. 204-61, Bothes of Dark Adventure," Allantic Ganzalian, Sopt. 1904; pp. in Ozcara (1996), pp. 204-61, Bothese of an "unusual overgation" in New York in "Up in the Big (1907, Part II) Ozcara (1996), pp. 202.
 - ⁵⁸ Ronald J. Pollett, personal interviews, 6 and 20 Mar. 1983.
 - 50 Caroline Pike, personal interview, 3 Mar. 1983.
- ⁶⁰ This, apparently, was a remark Pollett often made. It has been relayed by Ethel Mae Barnes, telephone interview, 18 Mar. 1983, Caroline Pike and Ronald J. Pollett, in various interviews, and Abe Snow, telephone interview, 19 Mar. 1983.
- ⁶¹ Pollett, "The Manly Sport of Handlining," <u>Atlantic Guardian</u>, Sept. 1947; rpt. in <u>Ocean</u> (1956), p.49.
- ⁶² Ronald J. Pollett, personal interview, 6 Mar. 1983; it is also described by Pollett, in "Seven League Boots" Atlantic Guardian, Dec. 1951; rpt. in Ocean (1956), p.256.
 - 63 Op. cit.
- ⁶⁴ The strong impression of affluence Pollett conveyed is recalled by Alfreda and Harold Robert Personal interview, 31 May 1983, Max Cranford, personal interview, 1 June 1983, and other local informats.
- ⁵⁶ Pollett, "Summer Madness," <u>Atlantic Guardian</u>, (Aug. 1950); rpt. in <u>Ocean</u> (1956), p.121. This opportunity for the returned immigrant nevertheless has been given some derisive attention in local prose. See, for example, Helen Porter, "They Do it Every Summer," <u>Atlantic Guardian</u>

Jan.-Feb. 1954, p.31; Jesse Beaumont Mifflin, "The Return of the Native," <u>Newfoundland Quarterly</u>, vol.70 (Spring, 1984), pp.43-4;48; and a rhyming sketch entitled "Mary Phool," by Rev. Caon G.H. Earle, Ms. 76-93, Tape C3439, MUNFLA.

- ⁶⁶ Caroline Pike, personal interview, 6 Mar. 1983. Other information about their subsequent activities obtained from her, in personal interview, 17 Mar. 1983, and from David Gilbert, Grand Fpersonal interview, 28 May 1983, Mildred Johnson, letter to author, 17 July 1983, and other informants.
- ⁶⁷ Michael Blackmore, personal interview, 25 May 1983, corroborated by Walter Blackmore, personal interview, 24 May 1983. The brothers' impressions of Pollett's abilities were obtained during these interviews.
 - 68 Caroline Pike, personal interview, 6 Mar. 1983.
- ⁶⁰ See Peter Neary, comp. "U.S. Bureau of the Cenus, Newfoundland-born Population," ms, 1971, CNS, which provides statistics of numbers of Newfoundlanders in New York relinquishing their citizenship during the Irozbo. Of the approximately 6000 Newfoundlanders said here to reside in New York City, by 1920, 48.7 per cent of the males, and 50.5 per cent of the females habecome naturalized.
- ⁷⁰ Information about Pollett's new quarters obtained from Ronald J. Pollett and Caroline Pike, personal interview, 6 Mar. 1983.
- Ti Casoline Pike, personal interviews, 14 and 15 Mar. 1983. Other information about his citizenship bearing and political adilitions obtained offinity these interviews, and from Abe Soow, telephone interview; 10 Mar. 1983. Snow notes Roosevell's particular popularity among the entire Brooklyn-Newdondand community. Details of Roosevelt's programs during the period may be found in Elliot Roosevelt, and James Bough, A Rendervous with Destiny: The Rooseveltes of the White Hosses (NY, CP, Perman, 1975), and Caroline Britis, The Interhible Scara, passim.
- ⁷² Charles C. Sava, New York, N.Y., letter to author, 5 Dec. 1983, in which he includes xeroxed reproductions of the contents of "Ronald Pollett" file, Immigration Section, U.S. Dept. of Justice, New York, N.Y. These documents note that Pollett, prior to 1941, had lived in the U.S. as a "Permanent Resident," which did not enable him to receive federal benefits or to vote.

²³ Information about the resurgence of Newfoundinad-related activity during the period, obtained from Belle Feaver, telephone interview, O Mar. 1988; the <u>Newfoundinad Weekty</u> (N.Y.), 19040-; R.H. Tait, Newfoundinad; A. Summar; of The History and Development of Britains, 1904er; Colley Chem. 1972 to 1920; [New York]; Harringon Press, 1993, pp. 55. Aurona columns and the "Newfoundinad" life, <u>Recklyn Eagle</u> Norpe, Brooklyn Public Liberry, describe these activities. See also Pollett's account in "Newfoundinad" life, <u>Procklyn Eagle</u> Norpe, Brooklyn Public Liberry, describe these activities.

⁷³ Abe Snow, telephone interview, 19 Mar. 1983.

 $^{^{74}}$ Caroline Pike, personal interviews, 3 Mar., 6 and 17 Mar. 1983.

⁷⁵ Abe Snow, telephone interview, 19 Mar. 1983.

- 77 Pollett "Memories of 'Didder Hill." Ocean (1956), pp. 202.
- 78 Pollett, "A Born Trouter," Ocean (1956), p.35.
- ⁷⁹ Information about Pollett's new activities obtained from Caroline Pike, personal interviews, 3 and 17 Mar. 1983.
- ³⁰ See also Pollett's description in "Up in the Big City," Part III, <u>Atlantic Guardian</u> (Aug. 1951): rot. in Ocean (1956), p.240.
 - 81 This photograph is in the possession of Ronald J. Pollett.
 - 82 Caroline Pike, personal interview, 3 Mar. 1983.
 - 88 Blanche Gilbert, Halifax, N.S., telephone interview, 23 June 1983.
 - 84 Pollett, "A Born Trouter," Ocean (1956), p.36.
 - 85 Op. cit., pp.36-7.
 - 86 Ronald J. Pollett, personal interview, 6 Mar. 1983.
- 87 Information about Pollett's activities in Cape Cod obtained from Caroline Pike, personal interviews, 3 and 6 Mar. 1983.
 - 88 Pollett, "A Born Trouter," Ocean (1956), p.37.
- ⁸⁹ Information about Pollett's business venture in Buchans obtained from Mildred Johnson, letters to author, 17 July 1983 and 10 Jan. 1984, Caroline Pike, personal interview, 3 Mar. 1983, and Ronald J. Pollett, personal interview, 20 Mar. 1983.
 - ⁹⁰ Ronald J. Pollett, personal interview, 6 Mar. 1983.
 - ⁹¹ Caroline Pike, personal interview, 6 Mar. 1983.
- ⁹² Information about Pollet's activities with his co-workers and about "A Brief for Appellants" obtained from Caroline Pike, personal interview, 3 Mar. 1983, and Ronald J. Pollett, personal interviews, 6 and 20 Mar. 1983.
- ³³ Information about Pollett's activities obtained from Caroline Pike, personal interviews, 6 and 14 Mar. 1983, and Ronald J. Pollett, personal interviews, 6 and 20 Mar. 1983. The greeting cards are part of a large collection of mementos owned by Caroline Pike.
- ⁹⁴ Information about Pollett's trip to New Harbour obtained from Walter Blackmore, personal interview, 24 May 1983, Ralph Higdon, personal interview, 20 Mar. 1983, Alfreda and Harold Rowe, personal interview 31 May 1983.

IV. LOOKING HOMEWARD: THE WRITING OF THE OCEAN AT MY DOOR, 1945-55

¹ Information about Pollett's illness obtained from personal interviews with Caroline Pike, 3 and 17 Mar. 1983, and Ronald J. Pollett, 6 Mar. 1983. The seriousness of Pollett's illness can be inferred from Pike's recollection of her husband at their son's high school graduation in June, 1945; she said on 3 Mar. 1983, "Even then we knew he was dying."

² Pollett, "Up in the Big City," Part I, Ocean (1956), p.220.

3 Op. cit.

⁴ Caroline Pike, personal interview, 3 Mar. 1983. Other information about Pollett's activities during his convalescence obtained from her in various personal interviews.

Ethel Mae Barnes, telephone interview, 19 Mar. 1983, Caroline Pike, personal interviews, 18 and 19 Mar. 1983.

⁶ Blanche Gilbert, telephone interview, 23 June 1983.

⁷ Pollett's enthusiastic response to Wolfe was described in personal interviews with Caroline Pike, 3 Mar. 1983, and Ronald J. Pollett and Caroline Pike, 6 Mar. 1983. See C. Hugh Holman, The Lonelineus at the Core: Stuties in Thomas Wolfe (Baton Rongs, La: Louisians State Press, 1975), pp.xv-xvii, which notes a similar first reaction to Wolfe's novel by millions of his readers prior to the 1955.

8 See Thomas Wolfe, Of Time and the River, A Legend of Man's Hunger in his Youth (1935)tptp. Caden City, N.Y.: Garden City Books, Ind.). The author's frequent repetition of particular sentences and phrases, such as, "He awakes at morning in a foreign land and thinks of home," makes it difficult for the reader not to graps the author's major thematic concerns.

⁹ Wolfe, You Can't Go Home Again (Garden City, N.Y.: Garden City Books, 1940), p.433.

¹⁰ See the first chapter of Alice Marble, <u>The Road to Wimbledon</u> (N.Y.: Charles Scribner and Sons, 1946), pp.1-17.

¹¹ Pollett's family notes Pollett's aversion for spectator sports, in various interviews. Documents of the period, in contrast, note the general Newfoundland community's keen enjoyment of them and especially of baseball [the Dodgers were then at their winnip best) and the institution of various Brooklya-Newfoundland sport teams. See The Bulletin of the Newfoundland Assoc. Inc. of Brooklya-Newfoundland aport teams. When the Proposition of the Newfoundland Weekly (New Pork), 1940-1.

12 Marble, The Road to Wimbledon, p.167.

 15 The autographed book still is among Pollett's small collection of books.

¹⁴ A brief survey of the <u>Atlantic Guardian</u> and its editors may be found in Patrick O'Flaherty, The Rock Observed: Studies in the Literature of Newfoundland (Toronto: Univ. of Toronto Press, 1979), pp.150; 152-4; and a student essay by Joe Jennings, "The Atlantic Guardian 1945-1956;" ms., Patrick O'Flaherty collection, CNS.

¹⁵ Brian Cahill, "Guardian Angles," <u>Atlantic Guardian</u>, June-July 1945, p.2.

16 Brian Cahill, Ottawa, Ont., letter to author, 1 Aug. 1983.

¹⁷ See Richard Gwyn, <u>Smallwood: The Unlikely Revolutionary</u> (Toronto: McClelland and Stewart, 1968),pp.61-2, in which Young is singled out as a catalyst for the "golden vision" of Smallwood's leading NewGoundland into Confederation.

 18 Brian Cahill, letter to author, 1 Aug. 1983.

19 Ewart Young, "Guardian Angles," Atlantic Guardian, Sept. 1946, p.1.

²⁰ Ronald J. Pollett, in personal interviews, 6 and 20 Mar. 1983, has supplied information about this translation. Various letters written by Rose F. Eggli, Berne, Switzerland, appearing in the <u>Atlantia Guardian</u> between 1946 and 1950, suggest that it was she who undertook the

²² Information about Pollett's earnings obtained from Caroline Pike, personal interviews, 3 and 6 Mar. 1983, and Ronald J. Pollett, personal interview, 6 Mar. 1983. A statement dated 1955, from the Bank of Montreal, St. John's, where Pollett deposited most of his earnings from his writing, crorborates this estimate (Ronald J. Pollett collection).

²² Brian Cahill, letter to author, 1 Aug. 1983. The precarious finances of the magazine were noted by Don Ryan, St. John's, telephone interview, 12 Apr. 1983, Joseph R. Smallwood, St. John's, personal interview, 9 Feb. 1983, and other informants.

²⁵ Arthur R. Scammell, St. John's, personal interview, 31 Jan.1983, and Michael F. Harrington, St. John's, personal interview, 23 Feb. 1983.

²⁴ Information about Pollett's writing routine obtained from Ethel Mae Barnes, telephone interview, 19 Mar. 1983, Caroline Pike, personal interviews, 3 Mar., 6, and 19 Mar. 1983, and Ronald J. Pollett, personal interviews, 6 and 20 Mar. 1983.

²⁶ Ewart Young, "Guardian Angles," <u>Atlantic Guardian</u>, Sept. 1946, p.1. Young's enthusiasm for Pollett's work is suggested by the context of this remark; he wrote to readers, "You can depend on it—we are going to keep hammering on his door for more stuff that has a real salt-water tang to it."

25 Pollett, Foreward, Outport Stories, (1947), p.3.

27 Op. cit.

²⁸ Brian Cahill, "Books and Such," <u>Atlantic Guardian</u>, Dec. 1947, pp.26-7.

29 Op. cit., p.26.

To Guardian Associates Ltd., as noted by Harold Horwood, Annapolis Royal, N.S., letter to author, 30 Apr. 1983, was not a publishing house but a printing company which occasionally published books. Among the titles Young published after Oligont Stories [1947], are a large number of books which deal exclusively with Newfoundhand topics, such as Young's Around Newfoundhand with a Camera [1947] and Lee F. Earlish and Roo Pumbrev. Stranee Parts

<u>Newfoundland with a Camera</u> (1947), and Leo F. English and Ron Pumphrey, <u>Strange Fac about Newfoundland</u> (1948).

- 31 Abe Snow, telephone interview, 18 Mar. 1983.
- 32 Abe Snow, telephone interview, 18 Mar. 1983.
- 38 Pollett, "Summer Madness," Ocean (1956), p.116.
- 34 Op. cit., pp.117-8.
- 55 Op. cit., p.116.
- Mc Charles P. Frank, Edimand Wilson (New York: Twayne, 1970), p.138, cites various reviews of Meming of Heast County, which describe cites impressions of the book's form; for example, Malcolm Cowley thought it "a grouping of novella," and Harrinos Smith thought the book comprised "there short novels prefaced by three shorts stories." See the 1971 version of the book (NY; Doubleday, 1971), which Frank notes, pp.135-4, was unavailable (in the uncapurgated version) between 971 and 1995 in the United States, and still not for sale in New York State.
- ²⁷ Ronald J. Pollett, personal interview, 20 Mar. 1983. See similar comments in Frank, <u>Edmund Wilson</u>, pp.14;153-4, where it is noted that the book was banned in New York almost immediately after its publication. Some reasons for its suppression are provided.
- ³⁸ Pollett, "There's No Place Like an Outport," Ocean (1956), p.133. Details about their travel obtained from Ronald J. Pollett, personal interview, 6 Mar. 1983. The luxury of their means of travel is described generally in Keeneth Westort Joses, Romantic Enliwary (London: Arlington, 1971); see ch. 7, "The 'Newfie Bullet' through the Wilderness," pp.70-81, which provides an account of the train rich and airport facilities of the day.
- ³⁹ Pollett, "The Manly Sport of Handlining," <u>Atlantic Guardian</u>, Sept. 1947; rpt. in <u>Ocean</u> (1956), p.46.
 - 40 Op. cit., p. 51.
 - 41 Op. cit., pp.51-2.
 - 42 Op. cit., p.54.
- 43 Ronald J. Pollett, personal interview, 6 Mar. 1983. Similar information obtained from Max Cranford, personal interview, 1 June 1983, Hannah Brown, personal interview, 10 Sept. 1982, and other informatis in New Harbour.
- 44 Clarence Cranford, personal interview, 29 June 1983, Ralph Higdon, personal interview, 1 June 1983.

- 45 Ronald J. Pollett, personal interview, 6 Mar. 1983.
- ⁴⁰ Pollett, "On Going Home Again," <u>Ocean</u> (1956), p.76. Here and in other articles, Pollett would frequently refer to his trips as "medicine," or "tonics."
- ⁴⁷ Pollett, "Vignettes of the Village," <u>Atlantic Guardian</u>, Aug. 1949; rpt. in <u>Ocean</u> (1956), p.86.
 - 48 Op. cit., pp.86-8.
 - ⁴⁹ Ser Patrick O'Fisherty, The Rock Observed, pp.14-69, for an examination of the recommic and social effect on Newfoundation's reculling from Newfoundation's recomming an international strategic outpoot during World War Two, Information about changes in New Hambour obtained from personal interviews with Max Cranford, June 1983, W. Bramwell Pollett and Margaret Pollett, 9 Sept. 1982, Charles and Lucy Thorne, 10 Sept. 1982, Elsie Thorne, 30 May 1983, and other informants.
- ⁵⁰ Pollett's disdain of the jukeboxes has been noted in various interviews, with Ronald J. Pollett, 30 Mar, 1983. Charles and Lucy Thorne, 9 Sept. 1982, and other informants. Charles Thorne notes that during the period, even the smallest stores in the area had had the machines installed.
 - ⁵¹ Pollett, "Vignettes of the Village," Ocean (1956), p.83.
 - 52 Op. cit., p.85.
- 68 Pollett, "The Village Goes to Town," <u>Atlantic Guardian</u>, July 1949; rpt. in <u>Ocean</u> (1956), p.80.
 - ⁵⁴ Op. cit., p.77.
 - 55 Ted Thorne, New Harbour, personal interview, 31 Aug. 1982.
- 56 See Joseph R. Smallwood, <u>I Chose Canada</u>, pp.235-314, in which the chapter, "A Crusade is Born," describes the radio broadcasts and the political literature that "blanketed the country."
 - ⁵⁷ Ronald J. Pollett, personal interview, 6 Mar. 1983.
 - 58 Jim Pollett's will is in the possession of Ronald J. Pollett.
- Dellett, in these articles, emphasized these particular effects of modernization, though many other innovations concerned him. Many informants from New Harbour recall his attentiveness toward the slightest changes he now perceived. This experience proccupied him long afterward. As his friend Abe Snow noted, in a telephone interview, 18 Mar. 1983, Pollett's views on change and Conderletation were not shared by the general Brooklys-Newfoundland community. They, for the most part, welcomed new developments, this in view of the benefits for their families and friends on the island.
- 60 During the period 1947-9, Young frequently published his own editorials and articles in which his favourable views toward Confederation were described. See, for example, his article,

"The Challenge of Confederation," <u>Atlantic Guardian</u>, Sept. 1948, pp.48. A curious inconsistency has been noted in many of Policit's strictles from the period. All of these articles, mainly unreceptive to Confederation, refer to "last summer," or "1948" is brackets-which in context suggest that Policit's trip and the time of writing the articles were in the summer and late fall of 1948, actually, the relevant time was summer and fall, 1947. No explanation can be cited for this discremancy.

- 61 Ronald J. Pollett, personal interviews, 6 and 20 Mar. 1983.
- 62 Abe Snow, telephone interview, 18 Mar. 1983.
- ⁶⁵ Ethel Mae Barnes, telephone interview, 19 Mar. 1983.
- ⁶⁴ Information about Armstrong Publications and Pollett's work there obtained from Gunnar Janger, personal interview, 19 Mar. 1983; and Pollett's work file, International Typographical Union, No. 6, N.Y., N.Y.
 - 85 Pollett, letter to editors, Atlantic Guardian, Oct. 1948, p.9.
 - 68 Op. cit.
 - 67 Op. cit., pp.9-11.
 - 68 Ronald J. Pollett, personal interview, 6 Mar. 1983.
 - 69 See unsigned article, "Vet's Ball in Brooklyn," Atlantic Guardian, Feb. 1949, pp.35-6.
- ⁷⁰ See Brian Cahill, "Guardian Angles," <u>Atlantic Guardian</u>, Apr. 1949, pp.3-5. Cahill acknowledges Pollett's contributions in the introductory paragraph of the article.
 - 71 Brian Cahill, letter to author, 1 Aug. 1983.
- ⁷² Brian Cahill, "Guardian Angles," <u>Atlantic Guardian</u>, Feb. 1949, p.3. E.C. Boone, former circulation manager, has noted that the figure for readers per copy was probably as high as ten (in a personal interview, St. John's, 5 July 1983).
- 73 A brief historical review of the magazines of the period may be found in Patrick O'Flaherty, $\underline{\text{The Rock Observed}}$, pp.149-51.
 - ⁷⁴ Brian Cahill, "Guardian Angles," Atlantic Guardian, Mar. 1949, pp.12:14.
- ⁷⁶ A discussion of the slogan and the platform, may be found in Joseph R. Smallwood, <u>I Chose Canada</u>, pp.340-72.
- ⁷⁰ This particular attempt by Stirling is described in "Newfoundlanders Seek Aid for Economic Union with U.S.," <u>Brooklyn Eagle</u>, 23 Mar. 1949, "Newfoundland" file, <u>Brooklyn Eagle</u> Morgue, Brooklyn Public Library.

- 77 George Smith Allston, Mass., letter to editor, Atlantic Guardian, Feb. 1950, p.12.
- 78 Pollett, "Epitaph for Downey," Atlantic Guardian, June 1949; rpt. in Ocean (1956), p.66.
- ⁷⁹ Brian Cahill, "Guardian Angles," Atlantic Guardian, Mar. 1949, p.14.
- ⁸⁰ Information about Young's offer to Pollett, and about Young's new venture, obtained from promonal interviews with Walter Blackmore, 24 May 1983, Michael F. Harrington, 23 Feb. 1983, Caroline Pike, 3 Mar. 1983, and Brian Cashill, letter to author, 1. Aug. 1983. The final status of the St. John's firm is noted in "Guardian Press Ltd." Big, Registry of Deeds, Companies, and Securities, Government of Newfoundland, Deep of Justice, St. John's Law.
 - 81 Caroline Pike, personal interview, 3 Mar. 1983.
- ⁸² Caroline Pike, personal interview, 3 Mar. 1983. The investment is noted in "Guardian Press Ltd." file, Registry of Deeds, Companies and Securities, St. John's. Pollett is among the largest of the shareholders. His occupation is listed as "writer."
- ⁸⁵ Pollett, "There's No Place Like an Outport," Ocean (1956), p.126. Peg Godden, St. John's, personal interview, 12 Feb. 1983, notes that as a former assistant secretary to J.R. Smallwood, she wrote more than 2000 letters per year during the period to prospective tourists. See also various articles in the Atlantic Guardian and other journals of the day promoting tourism.
 - ⁸⁴ Pollett, "Rum in the Puddin'," <u>Atlantic Guardian</u>, Dec. 1950; rpt. in <u>Ocean</u> (1956), p.165.
- 86 Caroline Pike, personal interviews, 3 and 17 Mar. 1983, notes the occasion in summer, 1933, when she cooked a rum-soaked pudding to be served to a number of Pollett's relatives who belonged to the Salvation Army.
 - 86 Pollett, "Summer Madness," Ocean (1956), pp.118-9.
 - 87 Pollett, "Up in the Big City." Part III, Ocean (1956), p. 240.
- ⁸⁸ Abe Snow, telephone interview, 18 Mar. 1983. Snow notes that this was the general consensus among his large circle of Newfoundland-born friends in Brooklyn.
 - 89 Pollett, "Fun for the Family," Atlantic Guardian, Jan. 1952; rpt. in Ocean (1956), p.260.
 - 90 Ethel Mae Barnes, telephone interview, 19 Mar, 1983.
- ⁹¹ See Pollett's descriptions in his essay, "Newfoundland Riviera," <u>Atlantic Guardian</u>, Nov. 1951; rpt. in <u>Ocean</u> (1956), pp.247-9.
- ⁹² Pollett, "The Passing of the Stagehead," <u>Atlantic Guardian</u>, Oct. 1955; rpt. in <u>Ocean</u> (1956), p.303.
- ²⁵ Pollett, "On Going Home Again," <u>Ocean</u> (1956), p.61; and "The Village Goes to Town," <u>Ocean</u> (1956), pp.79-80.

- ⁹⁴ Pollett, "Seven League Boots," <u>Atlantic Guardian</u>, Dec. 1951; rpt. in <u>Ocean</u> (1956), pp.256-9.
- ⁴⁶ Pollett, "Home Grown Happiness," <u>Atlantic Guardian</u>, Feb. 1952; rpt. in <u>Ocean</u> (1956), p.267.
 - 26 Charles Thorne, personal interview, 9 Sept. 1982.
- ⁴⁷ Pollett, "On the Pig's Back," <u>Atlantic Guardian</u>, Mar. 1952; rpt. in <u>Ocean</u> (1956), pp.270-2.
 - 28 Pollett, "Home Grown Happiness," Ocean (1956), p.266.
 - 99 Pollett, "The Passing of the Stagehead," Ocean (1956), p.308.
 - ¹⁰⁰ Pollett, <u>Peter the Grate: An Outport Character Study</u> (St. John's: Guardian Associates Ltd., 1952), p.6.
 - 101 Op. cit., Author's Note, inside back cover.
 - And Accounts of the book's reception obtained from personal interviews in New Harbour with Advoordman, 9 Sept. 1982, Louise Woodman 31 Aug. 1982, Neil Woodman, 32 Aug. 1982 at the phone interviews with Carry J. Craiford, 7 Sept. 1982, Sense Foywill, 22 June 1983, and informants. For the majority of informants, <u>Peter the Grate</u> is the work for which Pollett is best remembered.
 - 108 Pollett, Peter the Grate, Author's Note, inside front cover, and p.7.
 - 104 Caroline Pike, personal interviews, 3 and 6 Mar. 1983.
 - 106 "Pollett" file, International Typographical Union, No. 6, New York, N.Y.
 - ¹⁰⁶ Caroline Pike, personal interviews, 3 and 6 Mar. 1983; Ronald J. Pollett, personal interview of Mar. 1983. In April 1933, Ewart Young in an editorial explained to his readers that the cause of the <u>Atlantic Guardian</u>, spore was the failure of Guardian Press, but that the finances of the magazine were sound, <u>Atlantic Guardian</u>, Apr. 1953, p.2.
 - $^{\rm 107}$ Caroline Pike, personal interviews, 3 and 6 Mar. 1983.
- ¹⁰⁸ Ronald J. Pollett, personal interview, 20 Mar. 1983. Pollett's daughter also recalls her father's firm refusal to acquire a television set, telephone interview with Ethel Mae Barnes, 19 Mar. 1983.
 - 109 Abe Snow, telephone interview, 18 Mar, 1983.
 - ¹¹⁰ See <u>Atlantic Guardian</u>, Nov.-Dec. 1954, pp.5;13.
 - 111 Pollett, letter to Ewart Young, 13 June 1955, Appendix, 2 pp.

- 112 Walter Blackmore, personal interview, 24 May 1983.
- 118 Pollett, letter to Jim Pollett, 14 July 1955.
- 114 Op. cit.
- 115 Mildred Johnson, letter to author, 10 January 1984.
- 116 Ronald J. Pollett, and Caroline Pike, personal interview, 6 Mar. 1983;
- 117 Ewart Young, "Ron Pollett Passes On," Atlantic Guardian, Nov. 1955, p.1.
- ¹¹⁸ [Laura Blackmore?], "Former Grand Falls Man Dies in Brooklyn, N.Y.," <u>Grand Falls Advertiser</u>, 15 Sept. 1985, p.1.
 - ¹¹⁹ Ewart Young, letter to Caroline Pollett, 14 Sept. 1955.
- ¹²⁹ Muriel McKay, "A Tribute to Ron Pollett," recorded 13 Sept. 1955, St. John's, Nfld., CJON; rpt. in <u>Atlantic Guardian</u>, Nov. 1955, pp.23-4.

V. POLLETT'S WORKS: RECAPTURING AN OUTPORT HERITAGE

- 1 Pollett, Peter the Grate, p.41.
- ² Op. cit., p.17. Compare "Welcome Robin," in Royal Reader, No.2, pp.26-7.
 - 3 "Glen Cove Printer," Brooklyn Daily Eagle, 25 Nov. 1924.
- ⁴ Information about Pollett's activity as a journalist obtained from Michael Blackmore, telephone interview, 2 May 1988, Mar. Carnaford, personal interview, 1 June 1988, Mr. Howell, letter to author, 29 June 1983, Florence Piasent, personal interview, 10 Sept. 1982, and other informants. An extensive search of newspapers such as the Brookthz Distr. Eagle. Excellan Telegram, Newfoundland Weakly (Fowton), and Newfoundland Weakly (New York), was underwanted to the Control of the
- ⁵ See "A Night in Newfoundland," <u>Newfoundland Weekly</u> (Boston), 25 Apr. 1925, p.1; and various articles by a "Special New York Correspondent," published in this newspaper between 1924 and 1927.
- 6 "A Night in Newfoundland," p.1. Compare Pollett, "Memories of 'Didder Hill," Ocean (1956), pp.202-3.

- 7 Pollett, "The Outport Millionaire," Ocean (1956), p.21.
- ⁸ The magazine editors' philosophy was stated in each issue of the <u>Atlantic Guardian</u>. The particular enthusiasm the editors felt toward Pollett's early work was stated by Brian Cahill, letter to author, I Aug. 1983.
 - 9 Pollett, "The Ocean at My Door," Outport Stories (1947), p.13.
 - 10 Pollett, "There's No Place Like an Outport," Ocean (1956), p.127.
 - 11 Pollett, Forward, Outport Stories (1947), p.3.
 - ¹² Pollett, Author's Note, <u>Peter the Grate</u>, inside back cover.
 - ¹³ Pollett, "The Matchmaker," <u>Atlantic</u> <u>Guardian</u>, Feb. 1956, p.5.
 - ¹⁴ Pollett, Appendix, letter to Ewart Young, 13 June 1955, 2 pp.
 - ¹⁵ Pollett, letter to Jim Pollett, 14 July 1955, p.1.
 - ¹⁶ Pollett, "The Passing of the Stagehead," <u>Ocean</u> (1956), p.303.
- ¹⁷ See full-page monthly advertisements for the book in <u>Atlantic Standina</u>, 59c. 1956-Dei. 1957. Also noted in these advertisements are various popular reactions to the book, statisties of sales, and descriptions of the variant issues available. Most of this information was corroborated by Caroline Pike, personal interview, 3 Mar. 1983, who arranged with Young to have the book published.
 - ¹⁸ Ewart Young, "Ron Pollett Passes On," <u>Atlantic Guardian</u>, Nov. 1955, p.1.
- ¹⁹ See Pollett, "The Outport Millionaire," <u>Ocean</u> (1956), pp.21;24, and "A Born Trouter," <u>Ocean</u> (1956), pp.35;37.
- ²⁰ The notion of "fixing" as a conscious process that autobiographers must engage in to define their changing words, is described in Goreges Guoden", "Conditions and Limits of Autobiography", in James Olsey, Ed. <u>Autobiography: Essays Theoretical and Critical (Princetor: Princetor Curier, Press, 1999), pp. 38-48. Guoden Josep, 203. Hust this process is not a distincted or era as "illusion of the past." See his essay, "Full of Life Now," in Olsey, Ed. <u>Autobiography</u>, pp. 50-4.</u>
- ²¹ See Carlos C. Drake, "Literary Criticism and Folklore," in <u>Journal of Popular Culture</u>, vol.5 (1971), pp.287-97. He raises these points, and yet notes the value of the immediacy which such collections impart, in contrast to the "detached and labored quality" of literary accounts which attempt to fictionalize these raw materials.
- ²² Pollett, "Let's Look at the Squid," <u>Ocean</u> (1956), p.27, and "Where Christmas is Christmas," <u>Ocean</u> (1956), p.38.

- ³³ Pollett's numerous references to pictures, scenes, windows, and other modes of visualizing, his thoughts and sentiments, often provide more than a singer and impart a sense of immediacy, for he often enjoins his reader to see and recall with him the virid details he remembers or imagines of the outpoor. At times, however, he lapses into the sentimental and highly personalized mentalgic memory, each as when, or dreaming of a trip home, he meetly anys that "the amperical reverse" "Summer Madaess," Czesal (1995), p. 120.
 - ²⁴ Brian Cahill, "Books and Such," <u>Atlantic Guardian</u>, Dec. 1947, p.26.
- ²⁶ See Roy Pascal, <u>Design and Truth in Autobiography</u> (London: Routledge and Kegan Paul, 1960), pp.3-10, which describes the objective of coherence which the autobiographer strives for. Indeed much of the current writing on the "art of autobiography" likens the crafting of life stories to the techniques of fiction.
 - ²⁶ Pollett, <u>Peter the Grate</u>, pp.50-2.
 - ²⁷ Op. cit., pp.46; 52.
- ²⁸ Pollett uses this phrase in "The Ocean at My Door," <u>Outport Stories</u> (1947), p.13, and in context, he suggests that no part of his experience, however novel, after leaving this home had made him doubt the rich quality of outport life he said he had known.
- ³⁰ Pascal, in Design and Truth in Autobiorraphy, p.177, raises this point about the restrictness of autobiography: "The subolographer can settle age inside other people and outside himself," but William L. Howarth expands upon the reason for this limitation, noting that although the form "welcomes all the devises of skilled narration" of historical writing, it "observes few of the restrictions—scruncy, impartiality, inclusiveness." See his article, "Some Principles of Autobiography," in Obley, Autobiography, in Obley, Autobiography, and Design Principles of Autobiography, and the principles of Autobiography, in Obley.
- ³⁰ Pollett, Peter the Grate, p.4. The insumerable times in this work in which Pollett refers to gossip about Peter ("some people said," p.5, or, "indeed there were hone who insisted," p.6) are matched by the number of times which Pollett insists that he has set out to tell the "true," if less colorful, story of the figure. See for example, "Author's Note," inside back cover; pp.5-6; p.16.
- ³¹ These supects of traditional storytelling have been outlined and serstinized in a pictorn or folklower articles and texthooks. A general discussion of the various forms a personal narrative may take, which in turn impose cerain demands on the teller's technique, may be found in Jan H. Bruwand L. B. Esting of American Galdhice, An Introduction (New York: Ww. Norton, 1988).
 The first person marrative in literature, of course, also can be a standard device serving to impart immediacy and verneity.
 - ⁵² Pollett, <u>Peter the Grate</u>, p.11.
 - 58 Pollett, "Up in the Big City," Part II, Ocean (1956), p.227.
- ³⁴ Op. cit., pp.227-8. Various informants from Grand Falls have noted that congegrating at the train station to see who was coming to town was the major source of Sunday night entertainment in the 1920s. Interviews with Robert Arkeley, 27 May 1983, Walter Blackmore, 24 May 1983, Alphoasus Power, 25 May 1983.

⁸⁸ Pollett uses this phrase rather ironically to depict a tourist's view of the city in "Up in the City," Part I. Ocean [1956], pp. 220; 220. Indeed he frequently attempts to dispel the illusion of novelty, by, for instance, describing a Broadway play as "smally nothing more than the kind of tale the village gossip whispers to her neighbor over the back fence" ("Up in the Big City," Part II, Ocean [1956], pp. 231-22).

³⁶ Pollett, "Up in the Big City," Part I, <u>Ocean</u> (1956), p.222.

²⁷ O_{D.} cin. Part III, p.339; and Part III, p.322. In the latter instance, he is of course referring to such inxury trinkets as gold-plated shooks available at exclusive shops on Fifth Arenne. Pollet underscores the irony toward the idea of a professional fisherman acquiring such commodities by including them in a list of a "few of the interesting items a fellow who's done well with the fish could buy on a trip to New York," p.332.

⁵⁸ Op. cit., Part II, p.228.

⁵⁹ Op. cit., Part I. p.221.

⁴⁰ See Pollett, "The Manly Sport of Handlining," Ocean [1956), p.52; "Seven Leapue Boots," [Ocean [1956], p.55; "NewToundland Riviera," Ocean [1956], p.52; "Bet has refreshed his memory as a result of his recent trips, and displays his ability to reader seasonably the ordinary, and anatural landscape; although in cases when some years have elapsed between trips, his writings tends to take on an ardent Pyrical quality that is uncharacteristic of his usual reminient tone. He describes, for instance, how small fish and intente preceive the piles of offul underneath the stages: "Got spleens were emeralds, the cyclealls pearls, the gills ten shades of roses," in "The Passing of the Staghendo, Ocean [1905], p.304.

⁴¹ Pollett, "Newfoundland Riviera," Ocean (1956), pp.242-3.

⁴⁷ In stories such as "Ship Aboy!" Pollett focuses on two key figures—the exanest and diligent fatheries young boy. Ernic Tran, and the weathered and knowledgeable teacher figures that for the part of the

⁴⁸ The choice of narrator of "Johnny the Pear," is an unusual one in Pollett's writing; while most of his child narrators are naive and credulous toward outport life, this one displays a jaded manner toward most events of a traditional nature.

44 Pollett, "Johnny the Pear," Ocean (1956), p.189.

⁴⁶ Op. cit., p.107. It should be noted that this story is far richer in descriptions of the practice of Bonfire Night customs, and also in the recording of the songs and activities at regular meetings on the local bridge, than is Pollett's later essay, "The Bridge," in which similar occasions are merely alluded to.

** See Pollett, Peter the Grate, p.64, where he raises the point that Peter was neither a particularly good nor harmful character, and "the cemetaries everywhere are full of people about

whom can be said the same thing." This rather flat and mundame final statement nevertheless alludes to the innumerable phrases from local gossip peppering the narrative, and emphasizes Pollett's concern to redress the simple interpretations generally held about the figure and his qualities.

- ⁶ The latter three magazines, though more "literary," were unabashedly nationalistic in tone and contest; and after Confederation, tones magazines that managed to endure chose to maintain that outlook. In 1935, Herbert Granford, the editor of The Newfoundland Storr, seized upon the recent Report on Arts, Letters, and Sciences by the Massey Commission, which said that periodicals play a vital part in the arts and represent "the nearest thing to a national literature." He used these findings in order to derifie local advertisers releasts to support tool ventures such as his, but instead had supported "those smooth-talking outsiders." See his editorial in The Newfoundhand Storr, vol.7 (1951a), p. 18.
- ⁴⁸ This thesis does not attempt to address the entire body of literature of the period from these perspectives. A brief discussion of one local author's use of traditional elements may be found in Roberta Buchanaa, "Some Aspects of the Use of Folklore in Harold Horwood's "Comerrow Will Be Sunday," Lecture, Folklore Studies Assoc. of Canada, Learned Societies Conference, Mostreal, 1890, CNS.
 - ⁶⁹ See Arthur R. Scammell, "Sea Fever," <u>Atlantic Guardian</u>, June 1945, pp.11-16.
- 50 See Scammell, "Buyer Must Be in Good Condition," <u>Atlantic Guardian</u>, July 1948, pp.27-31.
 - ⁵¹ Pollett, "Johnny the Pear," <u>Ocean</u> (1956), p.192.
 - ⁵² Pollett, "Newfoundland Riviera," Ocean (1956), p.248.
- ⁸⁰ Pollett, letter to editors, <u>Atlantic Guardian</u>, Nov. 1948, p.9. Among the brief sketches in the magazine by Miles are "The Widow I Met from Corner Brook," Oct., 1948, p.17, and by Sullivan, "Mummering at Christmas," Dec. 1948, p.12.
- ⁵⁴ Letten from readers suggest a thoroughly enthusiastic response to Pollett's collecting-to-interpreting method. The catalogues in, for instance, "Where Christmas," see concess transformed to create a more sophisticated, sensous treatment for his Christmas story. "The Dicky-Bird Dish," which was praised by one reader as "lend as a Christmas gift as any 1 boosed to find under the tree." See George Smith, letter to editors, <u>Atlantic Guardian</u>, Feb. 1950, p.12.
- 55 See J.S. Bratton, <u>Victorian Children's Fiction</u> for a detailed examination of this genre of literature and a broad discussion of school <u>Readers</u>,
- ⁶⁶ See Evelyn Everett-Green, <u>Squib and his Friends</u> (London: Thomas Nelson [c.1900]). See Bratton, <u>Victorian Children's Fiction</u>, pp.192-200s, which links Everett-Green with the trends of a later school of Victorian fiction, i.e., one that had evolved from evangelical tracts to a more general moral emohasis in literature.

⁵⁷ Alice Marble, The Road to Wimbledon, p.15.

- ⁵⁸ Leo Marx, "The Puzzle of Anti-Urbanism," in Michael C. Jaye and Ann Chalmers Watts, Eds. <u>Literature and the Urban Experience: Essavs on the City and Literature</u>, (New Brunswick, N.J.: Rutgers Univ. Press, 1981), pp.63-4.
- 50 Thomas Wolfe, Of Time and the River, p.141, although the catalogues proceed from pp.137-41.
- ⁶⁰ Pollett, "A Born Trouter," <u>Ocean</u> (1956), pp.35-6; and "Let's Look at the Squid," <u>Ocean</u> (1956), p.27.
 - ⁶¹ Pollett, "Up in the Big City," Parts I and III, Ocean (1956), pp.223;234.
 - 62 See Pollett, "Memories of 'Didder Hill," Ocean (1956), esp. pp.198-203.
 - 68 Wolfe, You Can't Go Home Again, p.89.
- ⁶⁴ See Edmund Wilson, <u>Memoirs of Hecate County</u>; among the characters who people the book are a recluse who raises turtles, and a repressed housewife, who might make for interesting stories, although the narrator is a distant and smug fellow who is preoccupied with writing a book on art history. He merely imagines some rather bizares stories shout his neighbours, but this gift for the bizare does not compensate for a complete abence of realism.
- ⁵⁵ Ernest Buckler, <u>The Mountain and the Valley</u> (1952; rpt. Toronto: McClelland and Stewart, 1961), p.11.
 - ⁶⁶ Charles Bruce, <u>The Channel Shore</u> (Toronto: Macmillan, 1954), p.10.
 - ⁶⁷ Muriel McKay, "A Tribute to Ron Pollett," <u>Atlantic Guardian</u>, Nov. 1955, p.21.
 - ⁶⁸ Theresa Collier, "One of the Millionaires," <u>Atlantic Guardian</u>, Jan. 1957, p.8.
- 69 Marion Wathen Fox, ''Two Maritime Advocates," <u>The Maritime Advocate and Busy East,</u> Sept. 1955, pp.29-30.
- ⁷⁰ Rev. Walter G. Jones, "You Can't Go Home Again," <u>Atlantic Guardian</u>, May 1948, pp.44-5.
- ⁷³ See for example, Art Harnett, "The Crucifit," Atlantic Guardian, Sept. 1948, pp.11-23; and Gladys M. Cow, "The Fugitive," Atlantic Guardian, Aug. 1948, pp.14-16. This trend in the writing proroked Brian Cahill to issue new directives to potential writers, arging them to ground their stories in the local millien. See "Guardian Angles," Atlantic Guardian, Nov. 1948, pp.4-6. A review of subsequent issues suggests that his advice west unherded.
- 72 Daisy Mae Billings, "Come Back Ron Come Back,"
 $\underline{\text{Atlantic}}$ Guardian, Apr.-May 1954, p.9.
- ⁷³ The citation is from monthly advertisements for the book appearing in the <u>Atlantic Guardian</u>, during 1956-7; its use in the schools has been noted by various informants, such as Brian and Shirley Brown, personal interview, 9 Sept. 1982, David Gilbert, personal interview, 28 May 1983, Don Ryan, telephone interview, 12 Apr. 1983.

- ⁷⁴ Pollett refers to Smallwood in these terms in, for example, "Rum in the Puddin'," <u>Ocean</u> (1956), p.178; and in "On the Pig's Back," <u>Ocean</u> (1956), p.272.
- 75 See "Atlantic Guardian Wins Prize Trophy," <u>Grand Falls Advertiser</u>, 29 Sept. 1955, p.4, which describes how the magazine was the first recipient of the Canadian Wine Institute Trophy (won by Macleans in 1956).
 - 76 Pollett, "Rum in the Puddin", Ocean (1956), p.167.
 - 77 Op. cit., p.169.
 - ²⁸ See Scammell's story, "Hard Cash," Atlantic Guardian, Nov. 1947, pp.9-14.
 - 79 Muriel McKay, "A Tribute to Ron Pollett," pp.23-4
 - ⁸⁰ Scammell, "Outport Heritage," <u>Atlantic Guardian</u>, May 1945, p.12.
- ⁸¹ See Pollett's, "The Garden on the Point." <u>Ocean</u> (1956), pp.237-8; and "The Manly Sport Of Handlining," <u>Ocean</u> (1956), pp.49-50. The seventh son, in folk belief, has inherited the gift for curing minor allments, such as toothaches.
- ⁸⁵ See Bitabeth Miller Russell, The Life and Times of Ted Buxell, [St. John's: Jesperson. 1931], pp. 1956, which records some of Russell's own seatiments toward the value of prescripts the output character in literature. She ignores Ted Russell's earliest attempts at detion, such as "The Groß Box", and "Doxy at Quid Vold," which were frequestly reprinted in the various journals of the day, and were indistinguishable from the bulk of parachial topics and annateur factional attempts reselved by suthors of this day. Russell hepath broadcasting his series "The in The Chomicles of Unick Mose (Portugal Cow., Nidd. Breakwater, 1975) and Tales from Piezon Intel (Portugal Cow., Nidd. Breakwater, 1975).
- ³⁸ A comment on the absence of a local publishing industry was obtained from Harold Horwood, letter to suther, 30 Apr. 1983. The critical reviews of new literary works that appeared in energapers and journals of the day give an indication of the literary miller. In a weekly column in the Excingin Celeram by Sylvas Well, if a sew book sullied to Newboundland or was by a type. See for example, her column "The Telegram is Wedneeday Bookshelf," Evening Telegram, 5 Spep, 1956, p. 7. This actionalistic estiments, however, was not confined to the literary realm a provincial archives (1956) and a provincial museum (1955) were now instituted—the past would seem to have takes on a new cultural significance. See Walken Stegger, "The Provincial Consenses "Link, of Turnatu Synatzeria, vol. of Commen 1974, pp. 29-2010, which addresses the encompaning entyl directions and developments in Canada and the United State. See topics 1974.
- ⁸⁴ Rev. of <u>The Ocean at My Door and Other Newfoundland Stories</u>, <u>Daily News</u>, 23 Nov. 1956, p.7. As a rule, the weekly book reviews in this paper devoted one paragraph to each book. Five paragraphs were devoted to Pollett's book.

⁸⁶ Margaret Duley, "Glimpses into Local Literature," <u>Atlantic Guardian</u>, July 1956, p.25.

- 88 Op. cit.
- 87 Op. cit., p.22
- ³⁰ See Michael F. Harnington, "Newfoundlanders Rend More Books," <u>The Newfoundlanders</u>, Feb. 1953, p.13. is the also noted the extent of services provided by travelling libraries, and think in organizations and people these facilities served, such as lighthouse keepers and church groups. Similar weekly articles appeared during 1954-6 in the <u>Previnty Teterarin</u>, in its column." Notes from the Library," many of these articles attempting to guage the response of readers to the recent inception of television on the island.
- 89 W.J. Lundrigan, letter to author, 19 July 1983, Raymond Hillier, letter to author, 6 Oct. 1983.
- 90 Pollett, Preface, "Uncle Ben's Adventure, <u>Atlantic Guardian</u>, Sept. 1954, p.5.
 - ²¹ Pollett, "The Ocean at My Door," Outport Stories (1947), p.6.
 - 92 Pollett, "Fun for the Family," $\underline{\rm Ocean}$ (1956), pp.232-3.
 - 33 Pollett, "Up in the Big City," Part I, $\underline{\rm Ocean}$ (1956), p.222.
 - 94 Jesse Fogwill, telephone interview, 19 June 1983.
- ⁹⁶ See Violet Randell, <u>Thursday's Child pp.30-48</u>; and Joseph R. Smallwood, <u>I Chose Canada</u>, pp.126-70.
- ³⁸ Rev. Philip Tocque, "Incidents of a Visit to New York [1893]," in <u>Kaleiderone Echnes</u>, Ed. Annio Tocque (Toronto: Hunter, Bonce O. Ld., 1895), pp. 188-75. This writing is similar to Tocque's earlier work, <u>A Penn at Uncle Sam's Farm</u> (Boston: Chaire H. Pierce and Co., 1851), in which Tocque seks to document the bisitory, prography, monuments, and industries of Boston and its surrounding cities: the personal touch that would give one a sense of life in any of these places is remarkably whent.
 - 97 Pollett, "Up in the Big City," Part II, Ocean (1956), p.233.
 - ⁹⁸ Patrick O'Flaherty, The Rock Observed, p.155.

BIBLIOGRAPHY

I. A CHRONOLOGICAL LIST OF RON POLLETT'S PUBLICATIONS

- "The Outport Millionaire." <u>Atlantic Guardian</u>, July 1946, pp.6-7;24-7.
 [Translated as "Der Küsten Millionär." Trans. (Rose F.Eggli ?). <u>Neue Zuricher Zeitung</u> (c.1947) not traced.
- "A Born Trouter." <u>Atlantic Guardian</u>, Aug. 1946, pp. 14-15;26;28-30.
- 3. "Let's Look at the Squid." Atlantic Guardian, Sept.1946, pp.19-21.
- "The Cat with the Yaller Face." <u>Atlantic Guardian</u>, Nov.1946, pp.20;22-31.
 [Rpt. in <u>Doveloods: Newfoundland Writings and Art Selected and Edited for Young People, Ed. Kervin Major. Portugal Core, Mid. Breakwater, 1974, pp.128-34; and <u>Communication</u> <u>Newfoundland-Sivile</u>, Narr. Clyde Rose, C.B.C. (St. John's) School Broadcasts, No. 5, Program <u>12</u>, 1974-51.
 </u>
- 5. "Where Christmas is Christmas." Atlantic Guardian, Dec. 1946, pp.20-1;23;39-42.
- "The Ocean at My Door," In <u>The Ocean at My Door and Other Newfoundland Outport Stories</u>, Montreal: Guardian Associates Ltd., 1947, pp.4-13.
- The Ocean at Mv Door and Other Newfoundland Outport Stories, Montreal: Guardian Associates Ltd., 1947. [This volume comprises items 1, 2, 3, 4, 5, 6 above.]
- 8. "The Manly Sport of Handlining." Atlantic Guardian, Sept. 1947, pp.18-22;32-8.
- 9. "On Going Home Again." Atlantic Guardian, Mar. 1949, pp.92-3;95;98;100;102;104.
- 10. "Epitaph for Downey." Atlantic Guardian, June 1949, pp.26-33.
- 11. "The Village Goes to Town." Atlantic Guardian, July 1949, pp.34-9.
- 12. "Vignettes of the Village." Atlantic Guardian, Aug. 1949, pp.20-2;29-31.
- 13. "The Tongue that Never Told a Lie." Atlantic Guardian, Nov. 1949, pp.31-3;56.

[Rpt. in <u>Doryloads</u>, Ed. Major, pp.142-7; and <u>Communications Newfoundland-Style</u>, Narr. Rose, No. 5, Program 19, 1974-5.]

- 14. "The 'Dicky-Bird' Dish." Atlantic Guardian, Dec. 1949, pp.54-63;66-71.
- 15. "Captain William J. Connors." Atlantic Guardian, Jan. 1950, pp. 20-2.
- 16. "There's No Place Like an Outport." Atlantic Guardian, July 1950, pp.10-17;46-54.
- 17. "Summer Madness." Atlantic Guardian, Aug. 1950, pp.18-25.
- 18. "Ship Ahoy!" Atlantic Guardian, Sept. 1950, pp.20-30.
- 19. "Rum in the Puddin'." Atlantic Guardian, Dec. 1950, pp.20-30.
- 20. "Johnny the Pear." Atlantic Guardian, Feb. 1951, pp.18-32.
- "Memories of 'Didder Hill." <u>Atlantic Guardian</u>, Apr. 1951, pp.25-31.
 [Rpt in <u>By Great Waters: A Newfoundland and Labrador Anthology</u>, Eds. Peter Neary and Patrick O'Flaherty. Toronto: Univ. of Toronto Press, 1974, pp.218-22.
- 22. "The Bridge." Atlantic Guardian, May 1951, pp.13-16:18-20:34-6.
- 23. "Up in the Big City." Part I. Atlantic Guardian, June 1951, pp.33-4;36-40.
- 24. "Up in the Big City." Part II. Atlantic Guardian, July 1951, pp.31-2:34:37-9.
- 25. "Up in the Big City." Part III. Atlantic Guardian, Aug. 1951, pp.25-9.
- 26. "Newfoundland Riviera." Atlantic Guardian, Nov. 1951, pp.41-6.
- 27. "Seven League Boots." Atlantic Guardian, Dec. 1951, pp.48-9;51-2;54-6.
- 28. "Fun for the Family." Atlantic Guardian, Jan. 1952, pp.27-30.
- 29. "Home Grown Happiness." Atlantic Guardian, Feb. 1952, pp.16-17;19-20.

- 30. "On the Pig's Back." Atlantic Guardian, Mar. 1952, pp.36-41.
- 31. "Spring is for the Lambs." Atlantic Guardian, Apr. 1952, pp.25-6;31-4.
- 32. Peter the Grate: An Outport Character Study. St. John's: Guardian Associates Ltd., 1952.
- 33. "The Garden on the Point." Atlantic Guardian, Oct. 1952, pp.37-43;45.
- 34. "The Bait Punt." Atlantic Guardian, Aug. 1954, pp.7-9.
- 35. "Uncle Ben's Adventure." Atlantic Guardian, Sept.1954, pp.6-9.
- 36. "Afraid-In the Dark." Atlantic Guardian, Oct. 1954, pp.27-31.
- 37. "The Passing of the Stagehead." Atlantic Guardian, Sept. 1954, pp.5-9.
- 38. "The Matchmaker." Atlantic Guardian, Feb. 1956, pp.5-11:29-32.
- 30. The Ocean at Mr. Door and Other Newfoundland Stories, St. John's Guardian Associates and John John Charles (This Vertice comprises the times above numbered 7, 8, 9), 101, 112 (in part), 13, 14, 16, 17, 18, 19, 20, 21, 22, 28-26 (appearing as one article, entitled "Up in the Big Gyr"), 26, 77, 28, 29, 30, 31, 32, 33, 48, 56 (in part), 36, 77, in all, this volume includes 33 of Pollett's publications, but note that "Uncle Ber's Adventure," which is included; in short from the table of contants.]

IL FUGITIVE WRITINGS BY POLLETT

- "Eeling." [unpublished (?); c.1911; poem; lost].
- "The Exploits Hotel." [c.1920; poem; lost].
- "A Brief for Appellants." New York: Privately printed [c.1945; prose sketch; lost].

III. ARTICLES ATTRIBUTABLE TO RON POLLETT

- 1. "Night in Newfoundland." Newfoundland Weekly (Boston), 25 Apr. 1925, p.1.
- 2. "Vet's Ball in Brooklyn." Atlantic Guardian, Feb. 1949, pp.35-6.

[It is possible that these two items were written by Pollett; but his authorship has not been proven.]

IV. LETTERS BY POLLETT

- To Atlantic Guardian editors. Atlantic Guardian, Nov. 1948, pp.9-11.
- To Jim Pollett, 14 July 1955, Ms., Bertha Pollett private collection, Toronto, Ontario,
- To Ewart Young. 13 June 1955. Ms., Ronald J. Pollett private collection, Brooklyn, New York.
- V. CRITICAL AND BIOGRAPHICAL STUDIES OF RON POLLETT AND HIS WORKS (INCLUDING OBITUARIES AND TRIBUTES)
- Anon. "Glen Cove Printer to Save Lord Northcliff's Legacy for Some Need of Future." <u>Brooklyn Daily Eagle</u>, 25 Nov. 1924.
- Anon. "Newfoundland Writer Dies." Obituary. Sunday Herald, 25 Sept. 1955, p.6.
- Anon. Rev. of <u>The Ocean at My Door and Other Newfoundland Stories</u>, by Ron Pollett.

 Daily News (St. John's), 23 Nov. 1956, p.7.
- Anon, "Ronald Pollett." Obituary, New York Daily Mirror, 11 Sept. 1955, p.55.
- Anon, "Ronald Pollett Dies in New York," Obituary, Evening Telegram, 20 Sept. 1955, p.3.
- Billings, Daisy Mae. "Come Back Ron Come Back." Atlantic Guardian, Apr.-May 1954, p.9.



Duley, Margaret. "Glimpses into Local Literature." Atlantic Guardian, July 1956, pp.21-6.

McKay, Muriel, "A Tribute to Ron Pollett," Recorded 13 Sept. 1955, St., John's, Newfoundland. CJON: rpt. in Atlantic Guardian, Nov. 1955, pp.21:23-4.

Neary, Peter, and Patrick O'Flaherty, Eds. By Great Waters: A Newfoundland and Anthology, Toronto: Univ. of Toronto Press, 1974, p.218.

O'Flaherty, Patrick, The Myth of the Happy Outport: Ron Pollett and Art Scammell, Recorded [ca.1977]. Newfoundland and her Writers videotape series. No.19, MUN.

----. The Rock Observed: Studies in the Literature of Newfoundland, Toronto: Univ. of Toronto Press, 1979, pp.154-6.

"Ron Pollett." 1974-5. C.B.C. School Broadcasts. Program 24. Tape 6. B.150-380. MUNET.A

Story, Norah. Oxford Companion to Canadian History and Literature. Toronto: Oxford Univ. Press, 1967, p.561.

Whalen, Terry, "Atlantic Possibilities," In Essays on Canadian Writing vol. 20 (Winter 1980-1). p. 48.

Young, Ewart. "Guardian Angles." Atlantic Guardian, Sept. 1946, p.1.

Andrews, Ralph L. (St. John's). Letter to author. 17 Aug. 1983.

Arkeley, Robert (Grand Falls, Nfld.). Personal interview. 27 May 1983.

Barbado, Rosemary (Brooklyn, N.Y.). Personal interview. 17 Mar. 1983.

Barnes, Ethel Mae (Centerreach, Long Island, N.Y.). Telephone interview. 19 Mar. 1983.

Bennett, Ray (Green's Harbour, Trinity Bay). Personal interview. 31 May 1983.

Blackmore, John (Grand Falls, Nfld.). Personal interview. 28 May 1983.

Blackmore, Michael (Grand Falls, Nfld.). Telephone interview. 25 May 1983.

Blackmore, Walter (Grand Falls, Nfld.). Telephone interview. 19 May 1983.

-----. Personal interview. 24 May 1983.

Boone, E.C. (St. John's). Personal interview. 5 July 1983.

Brake, Rev. Alan (New Harbour, Trinity Bay). Personal interview. 10 Sept. 1982.

Brake, Joyceann (New Harbour, Trinity Bay). Letter to author. 9 June 1983.

Brown, Brian and Shirley (New Harbour, Trinity Bay). Personal interview. 9 Sept. 1982.

Brown, C.George (New Harbour, Trinity Bay). Personal interview. 10 Sept. 1982.

Brown, Hannah (New Harbour, Trinity Bay), Personal interview, 10 Sept. 1982.

Brown, Jack (Grand Falls, Nild.), Personal interview, 26 May 1983.

Bugden, Gertrude (English Harbour, Trinity Bay). Personal interview. 29 May 1983.

Bugden, Louise (Grand Falls, Nfld.). Personal interview. 25 May 1983.

Burke, Maurice (St. John's). Letter to author. 27 Apr. 1983.

Burt, Bertha (Green's Harbour, Trinity Bay). Letter to author, 27 June 1983.

Butler, Debbie (Queens, N.Y.), Telephone interview, 17 Mar. 1983.

Cahill, Brian (Ottawa, Ontario). Letter to author, 1 Aug. 1983.

Cater, R.J. (Grand Falls, Nfld.). Letter to author. 7 Feb. 1983.

Collins, Clide (Port Rexton, Trinity Bay). Personal interview. 28 May 1983.

Cooper, Victor (Green's Harbour, Trinity Bay). Personal interview. 31 May 1983.

Cranford, Clarence (St. John's). Personal interview. 29 June 1983.

Cranford, Garry J. (Torbay, Nfld.). Letter to author. 12 March 1982.

----. Telephone interview. 27 March 1982.

----. Telephone interview. 7 Sept. 1982.

Cranford, Max (New Harbour, Trinity Bay). Personal interview, 1 June 1983.

Edwards, Charles (Grand Falls, Nfld.). Personal interview. 25 May 1983.

Feaver, Belle. (Brooklyn, N.Y.). Telephone interview. 6 Mar. 1983.

Fogwill, Jesse (St. John's). Telephone interview, 19 June 1983.

Fowlow, Ray (Port Rexton, Trinity Bay). Personal interview. 28 May 1983. Fudge, Bruce (Grand Falls, Nfld.). Personal interview, 25 May 1983. George, Lloyd (Dildo, Trinity Bay). Personal interview. 10 Sept. 1982. Gilbert, Blanche (Halifax, Nova Scotia). Telephone interview. 23 June 1983. Gilbert, David (Grand Falls, Nfld.). Personal interview. 28 May 1983. Godden, Peg (St. John's). Personal interview, 12 Feb. 1983. Gosse, Marion (St. John's). Telephone interview. 2 Apr 1984. Green, E.S. (Grand Falls, Nfld.). Letter to author. 16 July 1983. Harrington, Michael F. (St. John's), Personal interview, 23 Feb. 1983. Higdon, Ralph (New Harbour, Trinity Bay). Personal interview, 1 June 1983. Hillier, Charles (Grand Falls, Nfld.). Personal interview, 24 May 1983. -----. Personal interview. 26 May 1983 Hillier, Harrison (Green's Harbour, Trinity Bay), Personal interview, 10 Sept. 1982.

Hillier, Raymond (Rosemead, California). Letter to author. 6 Oct. 1983.

Horwood, Harold (Annapolis Royal, Nova Scotia). Letter to author. 30 Apr. 1983.

Howell, W.T. (Grand Falls, Nfld.). Letter to author. 29 June 1983.

Janger, Gunnar (New York, N.Y.). Telephone interview, 17 Mar. 1983.

----- Personal interview. 19 Mar. 1983.

Jones, Harvey (English Harbour, Trinity Bay). Personal interview. 29 May 1983.

Johnson, Mildred (Etobicoke, Ontario). Letter to author. 17 July 1983.

Kopeck, Thomas W. (Colorado Springs, Colorado). Letter to author. 16 June 1983.

Long, Ted (St. John's). Personal interview. 27 June 1983.

Lundrigan, William J. (Corner Brook, Nfld.). Letter to author. 19 July 1983.

Penney, Jo Ann (English Harbour, Trinity Bay). Telephone interview. 29 May 1983.

Penney, Willis (English Harbour, Trinity Bay). Personal interview. 29 May 1983.

Pike, Caroline (Brooklyn, N.Y.). Personal interview, 3 Mar. 1983.

Pinsent, Florence (Dildo, Trinity Bay). Personal interview. 10 Sept. 1982.

Pollett, Ronald J. (Brooklyn, N.Y.). Letter to author. 2 Dec. 1982.

Personal interview. 20 Mar. 1983.	
Letter to author, 17 May 1983.	
Letter to author, 23 May 1983.	
Pollett, Ronald J. and Caroline Pike (Brooklyn, N.Y.). Personal interview. 6 Mar. 1983.	
Pollett, W. Bramwell (New Harbour, Trinity Bay). Personal interview. 30 May 1983.	
Pollett, W. Bramwell and Margaret (New Harbour, Trinity Bay). Personal interview. 31 Au 1982.	g
Personal interview. 9 Sept. 1982.	
Personal interview, 31 May 1983.	
Power, Alphonsus (Grand Falls, Nfld.). Personal interview. 25 May 1983.	
Rowe, Harold and Alfreda (Green's Harbour, Trinity Bay). Personal interview. 31 May 1983.	
Russell, Daniel E. (Glen Cove, Long Island, N.Y.). Letter to author. 29 Dec. 1983.	
Ryan, Don (St. John's). Telephone interview. 12 Apr. 1983.	
Sanger, Jean (Dildo, Trinity Bay). Personal interview. 10 Sept. 1982.	
Sava, Charles C. (New York, N.Y.). Letter to author. 5 Dec. 1983.	
Scammell, Arthur R. (St. John's). Letter to author. 15 Sept. 1982.	
Personal interview, 31 Jan. 1983.	
Schroth, Raymond A. (Worchester, Massachusetts). Letter to author. 3 Jan. 1984.	
Silk, Alfreda (Grand Falls, Nfld.). Personal interview. 26 May 1983.	

Silk, Hiram (Grand Falls, Nfld.). Personal interview. 26 May 1983.

Smallwood, Joseph R. (St. John's). Personal interview. 9 Feb. 1983.

Snow, Abe (Coram, Long Island, N.Y.). Telephone interview. 18 Mar. 1983.

Thorne, Charles and Lucy (New Harbour, Trinity Bay). Personal interview. 31 Aug. 1982.

------. Personal interview. 9 Sept. 1982.

-----. Personal interview. 10 Sept. 1982.

Thorne, Elsie (New Harbour, Trinity Bay). Personal interview. 30 May 1983.

Thorne, James (Bishop's Falls, Nfld.). Personal interview. 27 May 1983.

Thorne, Leah (New Harbour, Trinity Bay). Personal interview. 1 June 1983.

Thorne, Ted (New Harbour, Trinity Bay). Personal interview. 31 Aug. 1982.

Tucker, Walter (Grand Falls, Nfld.). Personal interview. 26 May 1983.

Wall, Joe (Grand Falls, Nfld.). Personal interview. 24 May 1983.

Weinberg, Michael (New York, N.Y.). Telephone interview. 14 Mar. 1983.

Williams, Herbert (New Harbour, Trinity Bay). Personal interview. 10 Sept. 1982.

Woodman, Ada. (New Harbour, Trinity Bay). Personal interview. 9 Sept. 1982.

Woodman, Aubrey John (New Harbour, Trinity Bay). Personal interview. 9 Sept. 1982.

Woodman, Clarrie (New Harbour, Trinity Bay). Personal interview. 9 Sept. 1982.

Woodman, George (Bishop's Falls, Nfld.). Telephone interview. 27 May 1983.

Woodman, Louise (New Harbour, Trinity Bay). Personal interview, 31 Aug. 1982.

-----. Personal interview. 1 June 1983.

Woodman, Neil (New Harbour, Trinity Bay). Personal interview. 31 Aug. 1982.

Woodman, Reg (New Harbour, Trinity Bay). Personal interview. 30 May 1983.

VII. UNPUBLISHED AND ARCHIVAL SOURCES

Anglican Church of Newfoundland. "Anglican Ministers' Registry." Queen's College, St. John's.

"Bay Ridge, Brooklyn." Ms. file. Long Island Historical Society, Brooklyn, N.Y.

Brown C. George. "Traditions which Exist in the Woodman Area of New Harbour, Trinity Bay in General and the Repertoire of Aubrey John Woodman in Particular." Ms. 72-94, MUNFLA.

Buchanan, Roberta. "Some Aspects of the Use of Folklore in Harold Horwood's "Tomorrow Will Be Sunday." Lecture, Folklore Studies Assoc. of Canada, Learned Societies Conference, Montreal, 1980. Ms., CNS.

Chafe, Edward Vincent. "A New Life on 'Uncle Sam's Farm:' Newfoundlanders in Massachusetts, 1846-1859." M.A. Thesis, MUN, 1982.

Church Registry, 1864-1916. Church of St. Augustine's, New Harbour, Trinity Bay, Nfld.

Church Registry, and Vestry Minutes, 1917-18. All Saint's Church, English Harbour, Trinity Bay,

Cranford, Garry J. "Potheads and Drumhoops." Ms., in Cranford's private collection.

Crewe, Nimshi. Collected Personal Papers. PA/13, PANL.

Earle, Rev. Canon G.H. Ms., 76-93, Tape C2438, MUNFLA.

- "Fort Hamilton" file, Long Island Historical Society, Brooklyn, N.Y.
- George, Anne. "The Grand Falls Pulp and Paper Mill; its origin, growth and impact on the surrounding area." Ms., ED-12, MHG.
- George, Lloyd. "Some Heritage Values of the Area [from New Harbour to Blaketown] Served by the Lions Club." Lecture, Lions Club 11th Annual Charter Night, New Harbour, Trinity Bay, 28 March 1981. Ms., Georgé private collection.
- "Guardian Ltd." and "Guardian Press Ltd." files. Registry of Deeds, Companies, and Securities,
 Government of Newfoundland, Department of Justice.
- Harbour Grace, Nfld. Magistrate's Office. Letters and Documents. Coll-3, File 19, CNS Archives.
- Hicks, George, "Newfoundland From Shadow to Sunlight," Ms., 1968, CNS.
- [Hicks, George?] "Anglo-Newfoundland Development Company, Ltd.: Establishment and Fifty Years of Progress," Ms., in Walter Tucker collection.
- Jennings, Joe. "The Atlantic Guardian, 1945-1956." Ms., Patrick O'Flaherty collection, CNS Archives.
- Ludlow, A.N. "Grand Falls, A General History," Ms., G-14, MHG.
- Melvin, Patrick, "Grand Falls: Past, Present, and Future," Ms., G-16, MHG.
- Neary, Peter, comp. "United States Bureau of the Census: Newfoundland-born Population," Ms., 1971, CNS.
- "Newfoundland" file. Brooklyn Eagle Morgue, Brooklyn Public Library, Brooklyn, N.Y.
- Newfoundland. Dept. of Education. [Records, including documents relating to Pollett's school grades, and teacher's certification.]
- Newfoundland, Nominal Census, 1901; 1921, Microfilm, PANL,
- O'Hanlon, Timothy. "Neighborhood Change in New York City: A Case Study of Park Slope, 1850 through 1980." M.A. Thesis, N.Y.U., 1982.

Penney, Jan. "Research Paper on English Harbour, Trinity Bay," Ms., E-6, MHG.

Pollett, Ronald, file, Immigration Section, U.S. Dept. of Justice, N.Y. N.Y.

---- file. International Typographical Union, No. 6, N.Y., N.Y.

Pollett, Ronald J. Personal collection, Brooklyn, N.Y.

"Prospect Hall," file. Brooklyn Eagle Morgue, Brooklyn Public Library, Brooklyn, N.Y.

Reid-Newfoundland Correspondence. "Heart's Content Branch Line," file. GN 2/9, Box 6, 48/4/5, PANL.

Rowe, Martin. "The History of Green's Harbour, Trinity Bay." Ms., G-23, MHG.

Schwoeffermann, Catherine Ann. "An Exploration of the November Fifth Bonfire Celebration in Brigus, Newfoundland." M.A. Thesis, MUN, 1981.

SPG. "C" Mss., Box 1A/20, A-192, PANL.

Southcott, John A. "Church and School History of Grand Falls (1905-1940's)." Ms., G-18, MHG.

VIII, JOURNALS AND NEWSPAPERS

Atlantic Advocate 1956-7.

Atlantic Guardian 1945-57.

The Bulletin of the Newfoundland Association Inc. of Brooklyn, New York Jan. 1945.

Canadian Author and Bookman 1952-7.

Come Home Year Centennial Newsletter (St. John's) 1965-6.

Daily News (St. John's) 1947-56.

Evening Telegram (St. John's) 1920-2; 1947-56.

Family Fireside (Gerald S.Doyle Ltd., St. John's) 1947-51.

Grand Falls Advertiser 1947-56; 1980.

Harbor Grace Standard 1880.

Here in Newfoundland (St. John's) 1956-7.

The Maritime Advocate and Busy East 1951-6.

N.T.A. [Newfoundland Teachers' Assoc.] Journal 1955-7.

Newfoundland Quarterly 1955-60; 1984.

Newfoundland Companion 1945-7.

Newfoundland Stories and Ballads 1954-67.

Newfoundland Story 1947-53.

Newfoundland Weekly (Boston) 1924-7.

Newfoundland Weekly (New York) 1940-1.

Newfoundland Writer 1946-7.

The Newfoundlander 1949-54.

Protocol 1945-8.

Sunday Herald and Weekly Star 1947-55.

Western Star (Corner Brook, Nfld.) 1955-6.

IX. SELECTED SECONDARY MATERIALS

Abbott, David William. Rural to Urban Migration Adjustment: A Study of Newfoundlanders in Toronto. Toronto: York Univ., 1975.

Bennett, Marion T. <u>American Immigration Policies</u>; <u>A History.</u> Washington, D.C.: Public Affairs Press, 1963.

Bernard, William S. American Immigration Policy, New York: Harper, 1950.

Berthoff, Rowland Tappan. <u>British Immigrants in Industrial America, 1790-1950</u>, New York: Russell and Russell, 1950.

Bird, Caroline. The Invisible Scar, New York: McKay, 1966.

Brassard, Guy R. "Rev. Arthur C. Waghorne (1851-1900)." Canadian Botanical Assoc. Bulletin, Supp. to vol. 13, no. 2 (Apr. 1980), pp.17-18.

Bratton, J.S. The Impact of Victorian Children's Fiction, London: Croom Helm, 1981.

Breton, Raymond."Institutional Completeness of Ethnic Communities and the Personal Relations of Immigrants." <u>American Journal of Sociology</u>, no.70(1964), pp.193-205.

Brown, Alan A., and Egon Neuberger, Eds. <u>Internal Migration: A Comparative Perspective</u>, New York: Academic Press, 1977.

Bruce, Charles. The Channel Shore, Toronto: Macmillan, 1954.

Buckler, Ernest. The Mountain and the Valley, 1952; rpt. Toronto: McClelland and Stewart,

Carew, S.J. The Nine Lives of Paton College, St. John's: MUN, 1974.

Church of England Boards, Report of the Public Schools of Newfoundland, St. John's: Herald Printers, 1906-19 [annual publication for 1905-19].

Council of Higher Education, Newfoundland. Report of the Examinations Conducted by the Council of Higher Education, St. John's: 1912-17 [annual publication].

- Syllabus and Regulations of the Examinations of the Council of Higher Education, St. John's: J.W. Withers, 1914-16 [annual publication].
- Coles, Robert Reed, and Peter Luyster van Santvoord. <u>A History of Glen Cove.</u> Glen Cove, N.Y.: privately printed, 1967.
- Copp, Terry. The Anatomy of Poverty: The Condition of the Working Class in Montreal, 1897-1929, Toronto: McClelland and Stewart, 1974.
- Cox, Gordon, S.A. Folk <u>Music in a Newfoundland Outport</u>, National Museum of Man, Mercury Series, Canadian Centre for Folk Culture Studies, Paper No. 32, Ottawa: National Museums of Canada, 1980
- Cranford, Garry J. "The Conception to Trinity Bay Road Survey 1835-6." Newfoundland Quarterly, Nos. 2-3 (Summer/Fall 1981), pp.7-9.
- ------ (with Raymond Hiller). Potheads and Drumhoods; Δ Folk-history of New Harbour Trinity Bay, St. John's: Harry Cuff, 1983.
- Dorson, Richard M. "The Identification of Folklore in American Literature." <u>Journal of American Folklore</u>, vol. 70 (1957), pp.1-8.
- Drake, Carlos, "Literary Criticism and Folklore." <u>Journal of Popular Culture</u>, vol. 5 (1971), pp.289-97.
- Dundes, Alan. "The Study of Folklore in Literature and Culture: Identification and Interpretation." Journal of American Folklore, vol. 78 (1965), pp. 136-42.
- Everett-Green, E. Squib and his Friends, London: Thomas Nelson, [n.d.].
- Faris, James C. Cat Harbour: A Newfoundland Fishing Settlement. Newfoundland Social and Economic Studies. No.3. St. John's: MUN. 1972.
- Frank, Charles P. Edmund Wilson, New York: Twayne, 1970.
- Fraser, Raymond, and Clyde Rose and Jim Stewart, Eds. East of Canada: An Atlantic Anthology, Portugal Cove, Nfld.: Breakwater, 1976.
- Government of Canada, Census of Canada, 1921, Ottawa: F.A. Acland, 1929.

- Fourteenth Annual Report of Labour Organization in Canada, 1924, Ottawa: F.A. Acland, 1925.
- Government of Canada. Department of Labour. <u>The Labour Gazette</u>, 1912; 1921-4. Ottawa: Government Printing Office, 1912, 1921-4 [annual publication].
- Wages and Hours of Labour in Canada 1920-25. Supplement to The Labour Gazette.
 Jan. 1926. Ottawa: F.A. Acland, 1926.
- Government of Newfoundland. Census and Return of the Population of Newfoundland and Labrador, St. John's: 1901,1911,1921.
- -----. Education Act, 1916. Uncatalogued newspaper file, ACHL.
- Great Britain Newfoundland Royal Commission. Report. London: His Majesty's Stationery Office, 1933.
- Gwyn, Richard. Smallwood: The Unlikely Revolutionary. Toronto: McClelland and Stewart, 1968.
- Halpert, Herbert, and G.M. Story, Eds. <u>Christmas Mumming in Newfoundland: Essavs in Anthropology</u>, <u>Folklore</u>, and <u>History</u>, Toronto: Univ. of Toronto Press, 1969.
- Hiller, James. "The Origins of the Pulp and Paper Industry in Newfoundland," <u>Acadiensis</u>, vol. 2 (Spring 1982), pp.42-68.
- Holman, C. Hugh. <u>The Loneliness at the Core: Studies in Thomas Wolfe.</u> Baton Rouge, La.: Louisiana State Univ. Press, 1975.
- Horwood, Harold. Tomorrow Will Be Sunday, 1966; rpt. Don Mills, Ontario: Paperjacks, 1975.
- Howell, W.T., Comp. 'Moby Joe.' Grand Falls, Nfld.: Price (Nfld.) Pulp and Paper Co., Ltd., 1988.
- Hunter, Dr. A.C. "The Old Royal Readers." in <u>Baffles of Wind and Tiele: An Anthology of Newfoundhand Poetry. Prose and Drama.</u> Ed. Clyde Rose. Portugal Cove, Nild.: Breakwater, [1973], pp.41-57.
- Ives, Edward D. The Tape-Recorded Interview, Knoxville: Univ. of Tennessee Press, 1980.

- Jones, Kenneth Westcott, Romantic Railways, London: Arlington, 1971.
- Jaye, Michael C., and Ann Chalmers Watts, Eds. <u>Literature and the Urban Experience</u>: <u>Essavs on the City and Literature</u>, New Brunswick, N.J.: Rutgers Univ. Press, 1981.
- Johnson, Samuel. The Yale Edition of the Works of Samuel Johnson, vol.3, Eds. W.J. Bate and Albrecht B. Strauss. New Haven: Yale Univ. Press, 1969.
- Lawson, Alexander. Printing Types: An Introduction. Boston: Beacon Press, 1971.
- Lush, Fraser. "The Press in Central Newfoundland." <u>Journal</u> [of Business and Commerce], Nov.1967, pp.29-32.
- Mackay, R.A. <u>Newfoundland</u>; <u>Economic</u>, <u>Diplomatic</u>, <u>and Strategic Studies</u>, Toronto: Univ. of Oxford Press, 1946.
- MacLennan, Hugh. Each Man's Son. Boston: Little, Brown, 1951.
- Marble, Alice. The Road to Wimbledon. New York: Charles Scribner and Sons, 1946.
- McCann, Phillip, Ed. <u>Blackboards and Briefcases</u>; <u>Personal Stories by Newfoundland Teachers</u>, <u>Educators and Administrators</u>, St. John's: Jesperson, 1982.
- McGrath, R.T. "Why Recruiting is Low Outside St. John's." <u>The Gadet</u> (Newfoundland), March 1916, pp.28-9.
- Miller, Elizabeth Russell. The Life and Times of Ted Russell, St. John's: Jesperson, 1981.
- Mitchell, David J. "Living Documents': Oral History and Biography," <u>Biography</u>, vol. 3 (Fall 1980), pp.283-96.
- Mitzner, Julius. <u>Neighborhood Study of the Park Slope Community in Brooklyn.</u> New York: The New York School of Social Work, 1940.
- Morgan, Charlotte E. The Origin and History of the New York Employing Printers' Association, New York: AMS Press, 1968.

- Neary, Peter. "Canadian Immigration Policy and the Newfoundlanders, 1912-1939." Acadiensis, vol. 2 (Spring 1982), pp. 69-83.
- Nicolson, Colonel G.W.L. The <u>Fighting Newfoundlander</u>: A <u>History of the Royal Newfoundland</u>
 Regiment, St. John's: Government of Newfoundland, 1964.
- Noel, S.J. Politics in Newfoundland, Toronto: Univ. of Toronto Press, 1971.
- Nowlan, Michael O., Ed. Themes in Canadian Writing: The Maritime Experience, Toronto: Macmillan, 1975.
- Olney, James, Ed. <u>Autobiography: Essays Theoretical and Critical</u>. Princeton, N.J.: Princeton Univ. Press, 1980.
- Orton, Larry. "Newfoundlanders in Toronto." In <u>Perspectives on Newfoundland Society and Culture: Book of Readings.</u> Ed. Maurice A. Sterns. St. John's: MUN, 1974, pp. 82-3.
- Pascal, Roy. Design and Truth in Autobiography, London: Routledge and Kegan Paul, 1960.
- Poole, Cyril F. In Search of the Newfoundland Soul, St. John's: Harry Cuff, 1982.
- Pottinger, David. Printers and Printing, Freeport, N.Y.: Books for Libraries Press, 1941.
- Randell, Violet Bennett. Thursday's Child, New York: Carlton Press, 1975.
- Roosevelt, Elliott, and James Bough. <u>A Rendervous with Destiny: The Roosevelts of the White House.</u> New York: G.P. Putnam, 1975.
- Rose, Clyde, Ed. The Blasty Bough, Portugal Cove, Nfid.: Breakwater, 1976.
- Rowe, Frederick W. The Development of Education in Newfoundland, Toronto: Ryerson, 1964.
- -----. A History of Newfoundland and Labrador. Toronto: McGraw-Hill Ryerson, 1980.
- Rowe, Melvin. I <u>Have Touched the Greatest Ship; A Saga of Heart's Content</u>, St. John's: Town Crier, 1976.

Royal Readers, 7 vois. London: Thomas Nelson and Sons, [n.d.].

- Russell, Ted. The Chronicles of Uncle Mose, Ed. Elizabeth (Russell) Miller. Portugal Cove, Newfoundland: Breakwater, 1975.
- -----. The Holdin' Ground: A Radio Play, 1954; rpt. Toronto: McClelland and Stewart, 1981.
- Scammell, A.R. My Newfoundland, Montreal: Harvest House, 1966.
- Schroth, Raymond A. <u>The Eagle and Brooklyn: A Community Newspaper, 1841-1955.</u> Westport, Conn.: Greenwood Press, 1974.
- Shapiro, Stephen A. "The Dark Continent of Literature: Autobiography." <u>Comparative Literature Studies</u>, vol. 5 (1968), pp.421-54.
- Smallwood, Joseph R. I Chose Canada, Toronto: Macmillan, 1973.
- ———,Ed. <u>The Book of Newfoundland</u>, vols.1 and 2. St. John's: Newfoundland Book Publishers, 1937.
- -----,Ed. <u>Encyclopedia of Newfoundland and Labrador</u>, vol.1. St. John's: Newfoundland Book Publishers 1967, Limited, 1981.
- Stegner, Wallace. "The Provincial Consciousness." <u>Univ. of Toronto Quarterly</u>, vol. 43 (Summer 1974), pp.299-310.
- Stephenson, George M. A <u>History of American Immigration</u> 1820-1924. New York: Russell and Russell, 1964.
- Strauss, Victor. The <u>Printing Industry</u>, Washington, D.C.: Printing Industries of America, Inc., 1967.
- Tait, R.H. Newfoundland; A Summary of the History and Development of Britain's Oldest Colony from 1497 to 1939. [New York?]: Harrington Press, 1939.
- Thernstrom, Stephan, Ed. <u>Harvard Encyclopedia of American Ethnic Groups</u>, Cambridge: Harvard Univ. Press, 1980.
- Tizzard, Aubrey M. On Sloping Ground; Reminiscences of Outport Life in Notre Dame Bay, Newfoundland, St. John's: Breakwater, 1984.
- Tocque, Rev. Philip. Kaleidoscope Echoes, Ed. Annie Tocque. Toronto: Hunter Rose, 1895.

	A Peep at Uncle Sam's Farm, Boston: Charles H. Pierce and Co., 1851.
Waghorn.	e, A.C. "Christmas Customs in Newfoundland." <u>Journal of American Folklore</u> , vol. 6 (1893), p.63.
	"Death Signs and Weather Signs from Newfoundland and Labrador." <u>Journal of American Folklore</u> , vol. 13 (1900), pp.297-9.
Williams	Jack. The Story of Unions in Canada. [Toronto?]: J.M. Dent and Sons, 1975.

Wilson, Edmund. Memoirs of Hecate County, New York: Doubleday, 1947.

Wolfe, Thomas. From Death Until Morning, New York: Grosset and Dunlap, 1932.

Of Time and the River; A Legend of Man's Hunger in his Youth, 1935; rpt. Garden City, N.Y.: Garden City Books, [n.d.].

-----. You Can't Go Home Again, Garden City, N.Y.: Garden City Books, 1940.

Young, Ewart, Ed. This is Newfoundland, Toronto: Ryerson, 1949.





